



Assessment of the relative risk of degraded water quality to ecosystems of the Great Barrier Reef

Lead Authors: Jon Brodie, Jane Waterhouse, Jeffrey Maynard

Contributors: John Bennett, Miles Furnas, Michelle Devlin, Stephen Lewis, Catherine Collier, Britta Schaffelke, Katharina Fabricius, Caroline Petus, Eduardo da Silva, Daniel Zeh, Lucy Randall, Vittorio Brando, Len McKenzie, Dominique O'Brien, Rachael Smith, Michael Warne, Richard Brinkman, Hemerson Tonin, Zoe Bainbridge, Rebecca Bartley, Andrew Negri, Ryan Turner, Aaron Davis, Christie Bentley, Jochen Mueller, Jorge Alvarez-Romero, Nyssa Henry, David Waters, Hugh Yorkston, Dieter Tracey

Report No. 13/28

November 2013

Assessment of the relative risk of degraded water quality to ecosystems of the Great Barrier Reef

A Report for the Queensland Department of the Environment and Heritage Protection

Report No. 13/28

November 2013

Lead by Jon Brodie, Jane Waterhouse and Jeffrey Maynard with many contributing authors

Centre for Tropical Water & Aquatic Ecosystem Research

(TropWATER)

James Cook University

Townsville

Phone : (07) 4781 4262

Email: TropWATER@jcu.edu.au

Web: www.jcu.edu.au/tropwater/

Lead Authors:

Jon Brodie	TropWATER James Cook University
Jane Waterhouse	TropWATER James Cook University and C_2O Consulting
Jeffrey Maynard	Laboratoire d'Excellence 'CORAIL' USR 3278 CNRS – EPHE, CRIOBE, Papetoai,
	Moorea, Polynesie Francaise and C ₂ O Consulting

Contributors:

John Bennett	Department of Environment and Heritage Protection
Miles Furnas	Australian Institute of Marine Science
Michelle Devlin	TropWATER James Cook University and C2O Consulting
Stephen Lewis	TropWATER James Cook University
Catherine Collier	James Cook University
Britta Schaffelke	Australian Institute of Marine Science
Katharina Fabricius	Australian Institute of Marine Science
Caroline Petus	TropWATER James Cook University
Eduardo Da Silva	TropWATER James Cook University
Daniel Zeh	James Cook University
Lucy Randall	Australian Bureau of Agricultural and Resource Economics and Sciences
Vittorio Brando	Commonwealth Science and Industrial Research Organisation
Len McKenzie	TropWATER James Cook University
Dominique O'Brien	TropWATER James Cook University
Rachael Smith	Department of Science, Information Technology, Innovation and the Arts
Michael Warne	Department of Science, Information Technology, Innovation and the Arts
Richard Brinkman	Australian Institute of Marine Science
Hemerson Tonin	Australian Institute of Marine Science
Zoe Bainbridge	TropWATER James Cook University
Rebecca Bartley	Australian Institute of Marine Science
Andrew Negri	Australian Institute of Marine Science
Ryan Turner	Department of Science, Information Technology, Innovation and the Arts
Aaron Davis	TropWATER James Cook University
Christie Bentley	EnTOX, University of Queensland
Jochen Mueller	EnTOX, University of Queensland
Jorge G. Alvarez-Romero	TropWATER James Cook University and Australian Research Council Centre of Excellence for Coral Reef Studies, James Cook University
Nyssa Henry	Department of Science, Information Technology, Innovation and the Arts
David Waters	Department of Natural Mines and Resources
Hugh Yorkston	Great Barrier Reef Marine Park Authority
Dieter Tracey	Green Ant Photo Design

Citation

This report should be cited as:

Brodie, J., Waterhouse, J., Maynard, J., Bennett, J., Furnas, M., Devlin, M., Lewis, S., Collier, C., Schaffelke, B., Fabricius, K., Petus, C., da Silva, E., Zeh, D., Randall, L., Brando, V., McKenzie, L., O'Brien, D., Smith, R., Warne, M.St.J., Brinkman, R., Tonin, H., Bainbridge, Z., Bartley, R., Negri, A., Turner, R.D.R., Davis, A., Bentley, C., Mueller, J., Alvarez-Romero, J.G., Henry, N., Waters, D., Yorkston, H., Tracey, D., 2013. *Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef*. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/28, Townsville, Australia.

For further information contact:

Jon Brodie Centre for Tropical Water & Aquatic Ecosystem Research (TropWATER) James Cook University Email: Jon.Brodie@jcu.edu.au

This publication has been compiled by the Centre for Tropical Water & Aquatic Ecosystem Research (TropWATER), James Cook University.

© James Cook University, 2013.

Except as permitted by the *Copyright Act 1968*, no part of the work may in any form or by any electronic, mechanical, photocopying, recording, or any other means be reproduced, stored in a retrieval system or be broadcast or transmitted without the prior written permission of TropWATER. The information contained herein is subject to change without notice. The copyright owner shall not be liable for technical or other errors or omissions contained herein. The reader/user accepts all risks and responsibility for losses, damages, costs and other consequences resulting directly or indirectly from using this information.

Enquiries about reproduction, including downloading or printing the web version, should be directed to jane.waterhouse@jcu.edu.au.

Context to the report

The Queensland Government's Reef Water Quality Program provided funding for this up-to-date assessment of the relative risks of degraded water quality to ecosystems of the Great Barrier Reef to inform the development of Reef Plan 2013. The assessment, and its supplementary studies, informed the development of the 2013 Scientific Consensus Statement and, in particular, formed the basis of Chapter 3 of that statement. The Scientific Consensus Statement underpinned the management strategies and their priorities in Reef Plan 2013, including setting investment priorities for the Australian Government's Reef Rescue 2 program. This report, its supporting studies and the information collated for this assessment, will be a "living resource" that will continue to inform discussion of Reef Plan and other related management, investment and monitoring priorities at regional level in the reef catchments as their Water Quality Improvement Plans are updated.

The Queensland Government is committed to ongoing investment in Reef Plan science and, as new information becomes available, government policy will be adapted to take account of it. The report is provided in good faith, on the understanding that the information is not used out of the context explained above.

Acknowledgements

The authors of this report would like to thank the Queensland Department of Environment and Heritage Protection for the project funding.

A number of individuals have also assisted with data acquisition and input including Alana Grech and Rob Coles for advice on seagrass data, Dieter Tracey for assistance with graphics, John Bennett for project management and guidance, and Lucy Randall for training with MCAS-S.

Special thanks also go to Professor Barry Hart and Dr Jeff Dambacher for their comprehensive review of this report; their input and suggestions have contributed to a much improved final version of the report. The authors would also like to acknowledge Dr Roger Shaw and Mike Grundy for their valuable input and review of the associated Reef Plan Science Consensus Statement chapter on the relative risks of degraded water quality to the Great Barrier Reef which also informed the revision of this report.

Disclaimers

TropWATER advises that the information contained in this publication comprises general statements based on scientific research. The reader is advised and needs to be aware that such information may be incomplete or unable to be used in any specific situation. To the extent permitted by law, TropWATER (including its employees and consultants) excludes all liability to any person for any consequences, including but not limited to all losses, damages, costs, expenses and any other compensation, arising directly or indirectly from using this publication (in part or in whole) and any information or material contained in it.

Table of Contents

Citation		i
Acknowled	gements	iii
Disclaimers		iii
Executive S	ummary	1
1 Introdu	uction	11
Part A: Asse	essment of the relative risk of degraded water quality to ecosystems of the GBR	18
2 Assess	ment of the risk of pollutants to ecosystems of the Great Barrier Reef including differential risk	
between se	diments, nutrients and pesticides, and among NRM regions	18
2.1 Su	mmary of findings	18
2.2 Int	roduction	20
2.2.1	Risk assessment framework	20
2.3 M	ethods	25
2.3.1	Selecting and classifying variables	25
a)	Exceedance of suspended solids concentration thresholds	32
b)	Chlorophyll concentration exceedance	35
c)	Pollutant loading in river plumes	37
d)	Pesticide concentrations	40
e)	Recognising and assessing uncertainties in the selection of variables	42
2.3.2	Estimating habitat area	43
2.3.3	Assessment Method - Part 1: Differential risk between pollutants on GBR ecosystems	43
2.3.4	Assessment Method - Part 2: Combined risk of degraded water quality to GBR ecosystems	44
2.3.5	Assessment Method - Part 3: Relative risk of degraded water quality to GBR ecosystems	48
2.4 Re	sults	49
2.4.1	Habitat and regional NRM areas	49
2.4.2	Part 1: Relative importance of different pollutants to GBR ecosystems	49
a)	Sealments	51
b)	NUTRIENTS	60
C)		68
d)	Comparisons among variables and Regions	13
2.4.3	Part 2: Complete risk of degraded water quality to GBR ecosystems	/6
2.4.4	Part 3: Relative risk of degraded water quality to GBR ecosystems	81
a)	Assessment of end-of-catchment pollutant 10ads	81
(0 2.4.5		84
2.4.5	CONCIUSIONS	

Pa us	art B: C es, inc	Overall conclusions of the relative risk between sediments, nutrients and pesticides and between land dustries and catchments in the GBR	90			
3	Rela	ative risk from degraded water quality to GBR ecosystems	90			
	3.1	Additional information to support the assessment in Part A	90			
	3.2	Overall assessment of the relative risk of degraded water quality to GBR ecosystems	94			
	3.3	Overall conclusions	98			
4	Lim	itations to the risk assessment and future improvements1	.00			
5	5 References					
A	opendi	ix 1. Further information regarding remote sensing assessments	.21			
A	opendi	ix 2. Sensitivity analysis of options for combining water quality variables for determining relative risk 1	.24			

Executive Summary

A risk assessment method was developed and applied to the Great Barrier Reef (GBR) to provide robust and scientifically defensible information for policy makers and catchment managers on the key land-based pollutants of greatest risk to the health of the two main GBR ecosystems (coral reefs and seagrass beds). This information was used to inform management prioritisation for Reef Rescue 2 and Reef Plan 3. The risk assessment method needed to take account of the fact that catchment-associated risk will vary with distance from the river mouth, with coastal habitats nearest to river mouths most impacted by poor marine water quality.

The main water quality pollutants of concern for the GBR are enhanced levels of suspended sediments, excess nutrients and pesticides added to the GBR lagoon from the adjacent catchments. Until recently, there has been insufficient knowledge about the relative exposure to and effects of these pollutants to guide effective prioritisation of the management of their sources.

This report is presented in 2 main parts: Part A describes the risk assessment method and reports the main findings; Part B provides the overall conclusions which are also informed by a number of supporting studies which are provided as a separate report (Waterhouse, 2013). The key findings of all parts of the report and supporting studies are summarised below.

Part A: Risk assessment

The risk assessment method is described in Part A of this report and used a combination of qualitative and semiquantitative information about the influence of individual catchments in the 6 natural resource management (NRM) regions on coral reefs and seagrass ecosystems. The marine boundaries used for each NRM region are those accepted officially by the Great Barrier Reef Marine Park Authority. The area of GBR lagoon waters within each marine NRM region were also reported in recognition of other important habitats and populations that exist in these areas.

The relative risk of degraded water quality among NRM regions was determined by combining information on end-of-catchment pollutant loads and the estimated ecological risk of water quality to coral reefs and seagrass meadows in the GBR. Modelled end-of-catchment pollutant loads (generated from the Source Catchments model framework for the Paddock to Reef Program) were obtained for each NRM region for key pollutants, and only the anthropogenic portions of total pollutant loads were considered in relating the relative risk to NRM regions. The anthropogenic load is calculated as the difference between the long term average annual load, and the estimated pre-European annual load. This information was used to define a 'Loads Index'.

Ecological risk is generally defined as the product of the *likelihood* of an effect occurring and the *consequences* if that effect was to occur. However, in this assessment there is some inconsistency in our capacity across the variables to produce a true likelihood or true consequence estimate as mostly we have no or limited ability to produce these estimates right now. Therefore, ecological risk in the GBR is expressed simply as the area of coral reefs and seagrass meadows within a range of assessment classes (very low to very high relative risk) for several water quality variables in each NRM region in the GBR lagoon. Our method for calculating risk essentially assesses the likelihood of exceedance of a selected threshold. This likelihood was set as 1 for a parameter and location if observations or modelled data indicate that the threshold was not exceeded. Conversely, the likelihood was set as 0 if observations or modelled data indicate that the threshold was not exceeded. As consequences are mostly unknown at a regional or species level, potential impact was calculated as the area of coral reef, seagrass meadows and area of GBR lagoon waters (in km²) within the highest assessment classes of the water quality variables (reflecting the highest severity of influence). The effects of multiplying the habitat area by 1 or 0 for the likelihood mean that the final assessment of risk in this assessment is only an indication of potential

impact - the area of coral reef and seagrass meadows in which exceedance of an agreed threshold was modeled or observed. This becomes an assessment of 'relative risk' by comparing the areas of each habitat affected by the highest assessment classes of the variables among NRM regions, and was used to generate a 'Marine Risk Index' for coral reefs and seagrass meadows.

A suite of water quality variables were chosen that represent the pollutants of greatest concern with regards to agricultural runoff and potential impacts on GBR ecosystems. The selection of variables was informed by the supporting studies described in Part B of this report (Waterhouse, 2013), and include exceedance of ecologically-relevant thresholds for concentrations of total suspended solids (TSS) and chlorophyll *a* obtained from daily remote sensing observations, and the distribution of key pollutants including TSS, dissolved inorganic nitrogen (DIN) and photosystem II-inhibiting herbicides (PSII herbicides) in the marine environment during flood conditions (based on end-of-catchment loads and plume loading estimates). A spatial variable was included that represents the area of the GBR lagoon where primary crown-of-thorns starfish (COTS) outbreaks have been observed. COTS outbreaks are an important cause of coral loss on the GBR and appear to be a response to excess nutrient runoff from certain catchments that impact this 'COTS initiation zone'. In recognition of the importance of the influence of catchment discharges in driving COTS outbreaks, an index of regional contributions of river discharges to the COTS initiation zone is also included for coral reefs (COTS Influence Index).

The three indexes were then combined to generate a Relative Risk Index, which ultimately ranks the relative risk of degraded water quality to coral reefs and seagrass meadows in the GBR among NRM regions.

Supporting Studies (see chapters within Waterhouse, 2013)

The supporting studies of the project informed the selection of variables and methods of analysis used in the risk assessment. The studies also strengthen our understanding of the consequences of pollutant impacts on coral reefs and seagrasses. In particular: Chapter 1 emphasises the importance of nutrients in the initiation of COTS primary outbreaks and therefore loss of coral cover; Chapter 2 confirms the relative importance between nitrogen and phosphorus in driving productivity in the GBR lagoon; Chapter 3 reviews the effects of sediments and sedimentation on coral reef communities; Chapter 4 provides new information on the relative risk of PSII herbicides to coastal and marine ecosystems, identifying the highest risk areas to be in freshwater and coastal wetlands, estuarine areas and coastal seagrass and inshore reef communities; Chapter 5 develops relationships between flood plume frequency and end of catchment pollutant loadings to define plume water types related to water quality characteristics; Chapter 6 provides a case study that shows a correlation between plume water types and seagrass cover with a supporting review on the impacts of water quality on seagrass in an appendix (Chapter 7); and Chapter 8 summarises the results of water quality and phytoplankton samples collected in variable flood plume conditions between 2010 and 2012 to improve our understanding of phytoplankton population dynamics.

Key findings of the Supporting Studies include:

- A variety of experimental, modelling and observational evidence circumstantially, but strongly, supports the hypothesis that COTS primary outbreaks are initiated by an episode of greatly enhanced larval survival during conditions producing increased food availability for the filter-feeding pelagic larval stages (Chapter 1). The four primary COTS outbreaks originating in the Lizard Island Cairns region (14.5-17°S) since 1960 follow 2-5 years after wet seasons when early season (November-February) aggregate discharges from the Burdekin to Daintree Rivers exceeded 10 Km³.
- Hydrodynamic modelling and estimates of DIN loads were used to rank the individual contribution of significant rivers between the Daintree (16°S) and Burdekin (19°S) to regional runoff influences on the Lizard

Island – Cairns initiation region (Chapter 1). The initiation region is divided into two areas – north and south of Undine Reef (16°S). On a runoff volumetric basis, the Daintree has the largest influence followed by the Russell-Mulgrave and Tully. With DIN loads included, the Russell-Mulgrave and Tully Rivers are the most important in the northern part of the initiation region (14.5-16°S). For the southern sub-region (16-17°S) of the initiation region the Daintree, Barron and Burdekin Rivers ranked highest, indicating the additional significant influence of northward transport of the Burdekin River plume. When DIN loads are included, the influence of the Burdekin increases greatly, particularly in the southern part of the initiation region (16-17°S).

- Dissolved inorganic and particulate forms of nutrients discharged into the GBR are both important in driving ecological effects but increased nitrogen inputs are more important than phosphorus inputs (Chapter 2). Dissolved inorganic forms of nitrogen and phosphorus are considered to be of greatest concern compared to dissolved organic and particulate forms of nutrients, as they are immediately and completely bioavailable for algal growth. Particulate forms mostly become bioavailable over longer time frames, and dissolved organic forms typically have limited and delayed bioavailability.
- Across the 35 basins in the GBR catchment area, the highest annual average loads of total suspended solids are derived from the Burdekin River (3,306 kt/year), Fitzroy River (1,805 kt/year), Herbert River (482 kt/year), Mary River (362 kt/year) and Don River (330 kt/year) (Chapter 3). Recent research has highlighted gully and streambank erosion as the dominant sediment erosion processes occurring in the larger dry tropical river catchments of the GBR and Gulf of Carpentaria. Prior to these field studies, catchment modelling had identified hillslope erosion as the dominant source of sediment in these catchments. The very fine sediment fraction (<10 µm) has a shorter residence time than coarser sediment, and is less likely to be captured in storage pathways, such as reservoirs.
- Floodplume studies have found most sediment exported from GBR rivers (e.g. Burdekin, Fitzroy, Tully and Burnett Rivers and the Mackay Whitsunday rivers) is deposited within close proximity of the river mouth/estuary, near-shore zone and inner shelf of the GBR, with potential for remobilisation during subsequent wind and tide driven resuspension events. Such resuspension events can result in higher turbidity levels than measured in initial flood plumes. The delivery of this new sediment to the inner shelf sediment wedge plays a critical role on inshore turbidity regimes, with a recent study highlighting the influence of increased river flow/sediment loads on temporal variation in inshore turbidity.
- Turbidity reduces light for benthic organisms such as seagrass and corals (Chapter 3). Coastal coral reefs do grow in turbid water conditions at shallow depths, however biodiversity declines as a function of increased turbidity throughout the GBR. Variability in light conditions may be a greater stress than chronically reduced light, as energy demands to adjust to varying light conditions result in suboptimal energy gains.
- Sediment properties strongly influence the effects of sedimentation on corals. Nutrient enriched fine terrestrial silts found along the inshore GBR are most detrimental. Early life stages (e.g. fertilisation, settlement) of corals are particularly susceptible to poor water quality and sedimentation.
- Coral reefs most vulnerable to damage from turbidity and sedimentation are those found in locations with weak currents such as embayments (e.g. Keppel Bay, Cleveland Bay, Missionary Bay) and sheltered zones on deeper reef slopes, in places where fish abundances are low, and in regions that are frequently affected by other forms of disturbance such as cyclones, bleaching or crown-of-thorns starfish predation.
- Seagrass meadows are widespread and ecologically critical parts of the Great Barrier Reef Marine Park (Chapter 7). Recent, widespread loss of seagrass has conclusively demonstrated their sensitivity to water

quality, and the ecological ramifications of seagrass loss including dugong and turtle mortality. Seagrass meadow distribution, abundance, productivity, composition and resilience are structured by chronic and acute water quality impacts.

- Both suspended sediments and nutrient-stimulated blooms of epiphytes and plankton attenuate light and reduce light penetration to seagrass canopies. Light thresholds associated with event-based loss of seagrass have been derived during recent runoff events. PSII herbicides, which are found above guideline levels in the GBR, reduce photosynthetic efficiency in seagrass.
- Little is known about the types and concentrations of contaminants bound to sediment discharged by rivers into the GBR and the risk that these pose to GBR ecosystems.
- An assessment of pesticide risk to the GBR (Chapter 4) considered pesticides within two groupings: (1) the additive effects of PSII herbicides (diuron, atrazine, hexazinone, ametryn, tebuthiuron and simazine) normalised via two methods (toxicity equivalent: TEQ and multiple substances potentially affected fraction: ms-PAF) and compared to consequence values based on expert opinion and (2) the individual risk of other pesticides compared to their respective trigger value (in a hierarchical framework from the Great Barrier Reef Marine Park Water Quality Guideline trigger values, the Australian and New Zealand Water Quality Guidelines and other internationally-derived values). It was concluded that the Mackay Whitsunday and Burdekin region are considered to be at highest risk from PSII herbicides, followed by the Wet Tropics, Fitzroy and Burnett Mary regions. However, the risk of only a fraction of pesticides has been assessed, with only 6 out of the 34 pesticides currently detected included in the assessment, and therefore the effect of pesticides is most likely to have been underestimated.
- Concentrations of a range of pesticides exceed water quality guidelines in many fresh and estuarine waterbodies downstream of cropping lands. Of the five ecosystems that were considered within the Great Barrier Reef, it is suggested that the freshwater reaches of rivers and freshwater/coastal wetlands have the highest risk from pesticides (PSII herbicides and some non-PSII pesticides) followed by the estuarine reaches of the rivers, the coastal nearshore zone which includes intertidal and subtidal seagrass meadows, the inner shelf and the mid and outer shelf.
- Plume loading maps have been produced for dissolved inorganic nitrogen and total suspended solids for the period 2007 and 2011 (Chapter 5). These maps provide an indication of the relative influence of these pollutants in the GBR during flood plume conditions, classified from low to high. The area of coral reef and seagrass in each class varies considerably between NRM regions, reflecting the differences in the dispersal of pollutants and the locations of the ecosystems.
- Innovative remote sensing methods to map water clarity can also be used as an interpretative tool for understanding changes in seagrass meadow area and abundance at large spatial scales. This method was tested in Cleveland Bay, near Townsville in the northern GBR area (Chapter 6). Methods have been developed to map plume water types in the GBR by using MODIS true colour images reclassified in function of their dominant colour. Water types entering into the GBR lagoon though river discharges have been described as primary, secondary and tertiary, each characterised by different concentrations of Coloured Dissolved Organic Matter (CDOM), total suspended solids (TSS) and chlorophyll. Substantial seagrass loss has occurred in Cleveland Bay over the period 2008 to 2011 and changes in seagrass area and biomass were monitored annually for selected meadows; there was a strong correlation between total bay-wide meadow area and biomass and exposure to primary water (high TSS and CDOM, low light) at an annual time-scale and also at a 5 year time scale.

• Analysing the species composition and size class of phytoplankton associated with various stages of plume development may help provide the missing link between increased nutrient loads, higher nutrient concentrations, changed water quality conditions and possible changes to food web/primary production in GBR waters including COTS outbreaks. Thus it is essential that we have greater knowledge on the drivers and consequences of changes in water quality, and the associated phytoplankton response. Chapter 8 presents current information from both local and international sources on the potential relationship between phytoplankton communities and COTS populations, including results of water quality and phytoplankton community analysis from samples collected in Wet Tropics flood plumes between 2010 and 2012.

Part B: Relative risk of degraded water quality to GBR ecosystems

The results of the Relative Risk Index were combined with the outcomes of the Supporting Studies and published literature to assess the overall relative risk of degraded water quality to GBR ecosystems (Part B of this report). A tabulated summary of the overall outcomes is presented in Table i and illustrated in Figure i. In this table we used the individual assessments of each variable used in the ecological risk assessment to highlight where the variables dominate among the regions. The primary rank for each variable is listed, with an indication of whether it dominates in terms of coral reef or seagrass area, or both. The Marine Risk Index for coral reefs and seagrass is then shown for the results of the combined analysis of water quality variables. The regional anthropogenic loads as a proportion of the total GBR load for TSS, DIN and PSII herbicides are shown to identify the primary sources of anthropogenic loads delivered to the GBR, in addition to the Loads Index which combines this information. Additional information includes facts related to the influence of river discharges to the COTS Initiation Zone and additional pesticide information. The Relative Risk Index which represents the overall results of the risk assessment (Part A) is shown as the critical underpinning for the overall conclusions. This is then informed by the management issues and associated land uses which were derived from published evidence and expert judgement of the assessment team (informed by the preceding columns). The overall ranking of relative risk was developed by the assessment on the basis of the overall content of the table.

The information has been coupled with the summary of pollutant sources from land uses in each region to generate the management priorities identified Table ii. However, cost effective solutions for all of these management issues are not necessarily currently available for all of these priorities. These issues are discussed further in the Reef Plan Scientific Consensus Statement 2013 (see Thorburn et al. 2013).

Even though the nutrient related variables of Chlorophyll threshold exceedance and DIN plume loading were ranked highest in the Fitzroy region, there is insufficient knowledge of the sources of DIN in the Fitzroy region to make recommendations about management priorities to address nutrients in the region. Further knowledge of the role of particulate nitrogen, which is largely derived from both cropping and grazing lands, and the processing of this into dissolved inorganic nitrogen is important for making future management recommendations in the large grazing catchments of the Fitzroy region.

Overall conclusions:

The main finding of the study was that increased loads of suspended sediments, nutrients (nitrogen and phosphorus) and pesticides all pose a high risk to some parts of the GBR. However, the risk differs between the individual pollutants, between the source catchments, and with distance from the coast as per the conclusions below. These findings are represented in the map shown in Figure ii.

• Overall, increased concentrations of nitrogen from catchments between the Daintree and Burdekin Rivers pose the greatest risk to coral reefs. Runoff from these rivers during extreme and early wet seasons is

associated with outbreak cycles of the coral-eating COTS on the northern GBR shelf (15 to 17°S) that subsequently generate secondary outbreaks throughout the central and southern GBR. GBR-wide loss of coral cover due to COTS is estimated to be 1.4% per year over the last 25 years, and a new outbreak is underway. It is estimated that COTS have affected >1000 of the ~3000 reefs within the GBR over the last 60 years.

- Of equal importance is the risk to seagrass meadows from suspended sediments (TSS) discharged from
 rivers in excess of natural erosion rates, especially the fine fractions (clays). Whether carried in flood
 plumes, or re-suspended by wave action, suspended particulate matter creates a turbid water column that
 reduces the light required by seagrass and corals. High turbidity has been estimated to affect ~200 inshore
 reefs and most seagrass areas. The Burdekin and Fitzroy regions present the greatest risk to the GBR from
 increased suspended sediment loads.
- Loss of seagrass habitat as a result of cyclones, floods and degraded water quality appears to be associated with higher mortality of dugong and turtles.
- The risk to coastal seagrass beds (and freshwater and estuarine wetlands) from the six commonly used PSII herbicides (pesticides) was assessed as highest in the Mackay Whitsunday and Burdekin regions, followed by the Wet Tropics, Fitzroy and Burnett Mary regions to. Concentrations of a range of pesticides exceed water quality guidelines (thresholds) in many fresh and estuarine waterbodies downstream of cropping lands. However, the risk of only a fraction of pesticides has been assessed. Only 6 out of the 34 pesticides currently detected included in the assessment, and therefore the effect of pesticides is most likely underestimated.
- The ranking of the relative risk of degraded water quality between the NRM regions in the GBR is: Wet Tropics > Fitzroy > Burdekin > Mackay Whitsunday > Burnett Mary > Cape York. Priority areas for management of degraded water quality in the GBR are: Wet Tropics for nitrogen management; Mackay Whitsunday and lower Burdekin for PSII herbicide management; and Burdekin and Fitzroy for suspended sediment management.
- The regional ranking of risk to coral reefs from degraded water quality is: Wet Tropics > Fitzroy > Mackay Whitsunday > Burdekin> Cape York > Burnett Mary, while that for seagrass beds is: Burdekin > Wet Tropics > Fitzroy > Mackay Whitsunday > Burnett Mary > Cape York. The combined assessment of water quality variables in the GBR used the total area of habitat affected in the highest relative risk areas and end-of-catchment anthropogenic loads of nutrients, sediments and pesticides added to the GBR lagoon. Importantly in the Mackay Whitsunday Region, 40% of the seagrass area is in the highest relative risk class compared to less than 10% for all other regions. However the highly valuable seagrass meadows in Hervey Bay, and the importance to associated dugong and turtle populations in the Burnett Mary Region, were not included in the ranking analysis.

The risk assessment presented in this report has generated the most comprehensive assessment of the relative risk of degraded water quality to GBR ecosystems undertaken to date. However, it is important to reiterate that the rankings between NRM regions are relative, and do not represent absolute differences in the risk to GBR ecosystems. In this regard, even the lowest ranked regions of the Burnett Mary and Cape York region may pose a risk to GBR ecosystems, but relative to the other NRM regions they are considered to be of lower risk. This information can be useful in guiding investment priorities but should not be used in isolation from other knowledge related to regionally specific priorities supported by additional evidence, socio economic influences and limitations to the assessment that may have led to uncertainties in the results. The limitations to the assessment are described in Section 4 of this report, together with suggestions to improve upon the work presented herein.



Figure i. Map summarising the results of the Loads Index, Marine Risk Indexes (coral reefs and seagrass) and the Relative Risk Index. The assessment for coral reefs also includes the COTS Influence Index associated with the extent of the influence of regional river discharge to the COTS Initiation Zone; the scores are 100 for the Wet Tropics, and 16 for the Burdekin region. The method for deriving and combining the indexes is provided in Section 2.4.4.

Table i. Summary of the outcomes of the overall assessment of the relative risk of water quality in the GBR. Note that the Burnett Mary Region is shaded in grey to represent the fact that most reefs and seagrass meadows in this region were not included formally in the analysis and thus the validity of the result has high uncertainty.

Region	Dominant variables in marine assessment Variables where max area is in Region	Marine	Risk Index	Anthr as a the T	Regiona opogen proport otal GB	al iic Load tion of R Load	Loads Index	Additional Factors	Relative Risk Index	Management Issues	Associated land uses	Overall Ranking of Relative Risk
	CR = Coral Reef SG = Segarass	Coral Reef	Seagrass	TSS	DIN	PSII Herb						
Cape York	COTS Initiation Zone (CR)	12	4	3	<1	<1	0	Influence from catchment runoff is predominantly from Wet Tropics Rivers	9	The data in this uncertain due t of marine datas	Region are highly o limited validation sets.	LOW
Wet Tropics		100	83	9	20	61	100	86% volumetric contribution to COTS Initiation Zone	100	Nutrients Pesticides	Sugarcane Bananas	VERY HIGH
Burdekin	TSS 2mg/L (SG, CR) TSS 7mg/L (SG) TSS Plume loading (CR, SG) Chl 0.45μg/L (SG) DIN Plume loading (SG)	40	100	32	11	13	62	14% volumetric contribution to COTS Initiation Zone High risk from PSII herbicides to Ramsar listed freshwater wetlands in the lower Burdekin catchments	76	Sediments Pesticides Nutrients	Grazing Sugarcane (coastal)	HIGH
Mackay Whitsunday	Pesticide exposure (CR, SG)	54	37	4	6	12	25	High risk from PSII herbicides in Sandy Creek	50	Pesticides Nutrients	Sugarcane	MODERATE
Fitzroy	TSS 7mg/L (CR) Chl 0.45μg/L (CR) DIN Plume loading (CR)	86	59	17	5	4	28	Monitored loads of PSII herbicides were high in 2011 (not reflected in modelled baseline)	80	Sediments Pesticides Nutrients ¹	Grazing Cropping	HIGH
Burnett Mary	All variables rank relatively low	11	23	4	4	9	20	The Mary River has the fourth highest total and anthropogenic TSS load of all GBR catchments	19	Sediments	Grazing	UNCERTAIN
										All variables rar however, there in this result giv on the full exter corals and seag outside the GBF this assessment	nk relatively low, is high uncertainty ven the lack of data nt and condition of rass (which are RWHA) available for	

¹There is insufficient knowledge of the sources of DIN in the Fitzroy region to make recommendations about management priorities for these. Further knowledge of the role of particulate nitrogen, which is largely derived from grazing lands, and the processing of this into DIN is important for making future management recommendations in the large grazing catchments of the Fitzroy region.

Relative Priority	Management Priorities								
Thomas	Region	Pollutant Management	Key land uses	Comments					
1	Wet Tropics	Fertiliser nitrogen reduction	Sugarcane, Bananas						
	Burdekin	Erosion management in Burdekin	Grazing						
	Fitzroy	Erosion management in Fitzroy	Grazing, Cropping						
2	Burdekin	Pesticide reduction in (lower) Burdekin and Haughton	Sugarcane	Note that these actions should not be					
	Mackay Whitsunday	Pesticide reduction in all catchments	prioritised at the exclusion of other practices that are already in place to manage losses of other pollutants in the						
	Burdekin	Fertiliser nitrogen reduction in (lower) Burdekin and Haughton							
3	Mackay Whitsunday	Fertiliser nitrogen reduction	Sugarcane	Regions					
	Burnett Mary	Erosion management in all catchments	Grazing						
	Wet Tropics	Pesticide reduction in all catchments	Sugarcane						
	Fitzroy	Pesticide reduction in all catchments	Grazing, Cropping						
4	Burnett Mary	Further information is required to in assessment, including data on the ful condition of corals and seagrass (which a GBRWHA) in the region.	Habitat mapping, ecological value assessment and monitoring of ecosystem condition is required						
5	Cape York	Further information is required to und influences.	As a relatively low impacted area, management efforts should aim to maintain the current values of the region						

Table ii. Summary of management priorities for reducing the relative risk of pollutants to the GBR.



Figure ii. Illustration of the overall outcomes of the assessment of the relative risk of degraded water quality to Great Barrier Reef coral reefs and seagrass. The map shows the dominant land uses and priority pollutants and results of the overall relative risk ranking in each NRM region.

1 Introduction

Exposure to land-sourced pollution has been identified as an important factor in the world-wide decline in coral reef condition (Pandolfi et al. 2003; Burke et al. 2011). Different parts of the Great Barrier Reef World Heritage Area (GBRWHA) are exposed to different degrees of influence from land-sourced pollutants. The degree of exposure is a function of factors such as distance from the coast, the magnitude of river discharges, the distance from river mouths, wind and current directions and, the mobility of different pollutant types, and of course the different land-uses in the 6 natural resource management (NRM) regions of the Great Barrier Reef (GBR) catchment. This differential exposure to land-sourced pollutants has important consequences for the likely degree of degradation that habitats such as coral reefs and seagrass meadows may suffer as a result of land-sourced pollution and an assessment of exposure is required to prioritise management of such pollution on a regional basis.

To date the prioritisation of potential management responses between different pollutants, different land uses and industries, and different NRM regions has used methods such as Multiple Criteria Analysis (MCA) (Brodie and Waterhouse, 2009; Brodie et al. 2009; Cotsell et al. 2009; Greiner et al. 2005; Waterhouse et al. 2012). The methodologies used to date, though the best available at the time, were limited for reasons summarised in Table 1.1. While these analyses have proved useful for ongoing prioritisation of investment under Reef Plan, more specifically Reef Rescue and the selection of priority management areas under the Reef Protection Package, more sophisticated analyses are now needed to more confidently prioritise between pollutants, across the individual catchments. In addition it is worth noting that the availability of data related to seagrass meadows for these assessments remains relatively sparse.

The assessment approach described in this report is listed in the final column of Table 1.1 below. In 2011, the Australian Government funded a scoping project through the National Environment Research Program (NERP *Project 4.3: Ecological risk assessment of pesticides, nutrients and sediments on water quality and ecosystem* health) to review the methodology used in previous ecological risk assessment approaches for water quality in the GBR and make recommendations for a revised and improved assessment approach. The project team outlined a tiered risk assessment methodology that was recommended to provide a systematic, objective and transparent approach to quantify the relative risk of pesticides, nutrients and sediment to the GBR (Hayes et al. 2012). A meta-database was compiled to determine the availability of data for the risk assessment. However, due to timing and data and resource limitations, the proposed method was not able to be adopted by the Queensland Department of Environment and Heritage Protection (DEHP). Subsequently in 2012, the DEHP agreed to fund a second phase to the project through the Reef Protection Package Science Program (http://www.reefwisefarming.qld.gov.au/information/science.html) that required delivery of an assessment of the relative risk of pollutants to the GBR by March 2013. The approach that was selected for this project builds on previous assessments (Table 1.1). It still uses an MCA approach; however is improved with new input data, revised criteria and the application of a spatial multi criteria analysis tool, Multi-Criteria Analysis Shell for Spatial Decision Support (MCAS-S) developed by ABARES (refer to http://www.daff.gov.au/abares/data/mcass).

Table 1.1. Summary of the elements included in past and current assessments of the risk of water quality to the GBRused to inform Reef Plan management prioritisation.

Element	Reef Plan	Reef Rescue	Reef Protection Package	Reef Plan 3 / Reef Rescue 2 This report	
	Greiner et al. 2005	Cotsell et al. 2009	Waterhouse et al. 2012		
Method	МСА	MCA	MCA	MCA plus interpretive studies	
Data availability	Very limited	Limited	Moderate	Good	
Analysis end point	Coral reefs and seagrass	Coral reefs	Coral reefs, seagrass, water column	Coral reefs, seagrass, plus freshwater to marine ecosystems for pesticides	
Relative importance of pollutants	No	No	No	Yes	
Pesticide data	Very limited	Limited	Limited	Yes, with limitations	
Marine exposure estimate	Limited – from Devlin et al. 2003	Moderate – from Maughan and Brodie, 2009	Moderate – from Maughan and Brodie, 2009	Good – from recent work by Devlin et al. 2013a; Alvarez Romero et al. 2013	
Socio and economic values included	Yes	Yes	No	No	
Spatial coverage	All GBR, however Burnett Mary marine area outside of GBRMP excluded	All GBR, however Burnett Mary marine area outside of GBRMP excluded	Cape York and Burnett Mary excluded	All GBR, however Burnett Mary marine area outside of GBRMP excluded	

The objective of the project was to estimate the relative risk of pollutants in the GBR catchments to GBR ecosystem health. The assessment used a combination of qualitative and semi-quantitative information about the influence of individual catchments in the 6 NRM regions (Figure 1.1) on key GBR ecosystems (coral reefs and seagrass meadows). The marine boundaries used for each NRM region are those accepted officially by the Great Barrier Reef Marine Park Authority. The area of GBR lagoon waters within each marine NRM region were also reported in recognition of other important habitats and populations that exist in these areas.. Qualitative conclusions were drawn about coastal and estuarine wetlands where information was available. Unlike the risk assessment undertaken for Reef Plan in 2004 (Greiner et al. 2005), this assessment does not take into account the social and economic value of the assets such as tourism and fishing values.

This report is presented in 2 main parts:

Part A: Assessment of the relative risk of degraded water quality to ecosystems of the GBR.

Part B: Overall conclusions of the differential risk between sediments, nutrients and pesticides and between land uses, industries and catchments/regions in the GBR.

Supporting Studies (Waterhouse, 2013) were also a key component of this risk assessment and provided supporting evidence to determine differential risk between sediments, nutrients and pesticides and between land uses, industries and catchments.

The actual assessment of the relative risk of pollutants to the GBR ecosystems (Part A) is supported by a number of these specific studies that have either developed or confirmed our understanding of the characteristics of key pollutants that influence the risk to the GBR ecosystems (Supporting Studies). The results of these studies are presented as individual chapters in the Supporting Studies report and include:

- 1. Linkages between river runoff, phytoplankton blooms and primary outbreaks of crown-of-thorns starfish in the Northern Great Barrier Reef.
- 2. The Redfield Ratio and potential nutrient limitation of phytoplankton in the Great Barrier Reef.
- 3. Review of increased suspended sediment delivery to the Great Barrier Reef and the effects of subsequent sedimentation and light reduction on coral reefs.
- 4. Assessing the risk of additive pesticide exposure in Great Barrier Reef ecosystems.
- 5. Mapping of exposure to flood plumes, water types and exposure to pollutants (DIN, TSS) in the Great Barrier Reef: toward the production of operational risk maps for the World's most iconic marine ecosystem.
- 6. Assessment of seagrass and water quality influences in the Great Barrier Reef: A case study linking annual measurements of seagrass change to satellite water clarity data (Cleveland Bay, Queensland).
- 7. Review of the risks to seagrasses of the Great Barrier Reef caused by declining water quality.
- 8. Analysis of phytoplankton community structure in flood events in the Great Barrier Reef.

Together, the results of these assessments allow conclusions to be drawn about the relative risk of pollutants to the GBR and relate these back to land uses and catchments (Part B, Section 3).

All of these Chapters and this overall report have been reviewed by independent experts, with the exception of the report on phytoplankton in Chapter 8 of the Supporting Studies. Only a small amount funding was provided under this project to undertake the actual analysis of water quality and phytoplankton populations in flood conditions collected between 2010 and 2012. As limited funding was provided to report on the outcomes of these analyses, the report of this work provided in Chapter 8 is preliminary and has not been peer reviewed. It is intended that these results will be considered in the context of a broader review funded through NERP which is also related to knowledge of GBR phytoplankton populations and the links to nutrient dynamics and COTS outbreaks. These results will be published at a later date.

The information generated from this assessment will be invaluable to the review of the Department of Environment and Heritage Protection Reef Water Quality Program (formerly Reef Protection Package) and the development of Reef Plan 2013 (Reef Plan 3). In addition, it provides an integral component of the overall investment prioritisation being undertaken for Reef Plan 3 (incorporating Reef Rescue 2) illustrated in Figure 1.2. The investment prioritisation will take the assessment from this project in association with more detailed catchment assessments of the analysis of sub catchment pollutant loads to identify catchment hot spots for pollutant sources, and an assessment of the 'solvability' of management issues in different NRM regions and

catchment areas. The solvability assessment will be based on recent management practice adoption data for the sugar cane and grazing industries in the GBR catchments, and the cost effectiveness of improvements in water quality of the various practices. Together these assessments will provide robust, targeted scientific evidence of sources of high risk reef pollutants in the GBR catchment, with information to guide the best return on investment. This will enable better prioritisation of Reef Plan, Reef Water Quality Program and Reef Rescue 2 actions.

The outcomes of this project also inform one chapter of the Reef Plan Scientific Consensus Statement (SCS) 2013; *Chapter 3 Relative risks to the GBR from degraded water quality* (Brodie et al., 2013b) The SCS chapters and scope are illustrated in Figure 1.3. The evidence in these chapters also informs the methods and interpretation of results for this assessment and is referred to throughout this report.



Figure 1.1. Map showing the assessment boundaries considered in this risk assessment.



Figure 1.2. Assessment framework for the overall Reef Plan 3 and Reef Rescue 2 management prioritisation process. The *Management practice 'solvability'* assessment will be delivered by the Australian Government for Reef Rescue 2 and includes analysis of the management practice adoption and cost effectiveness data developed through the Reef Plan Paddock to Reef Program. The *Sources of pollutants* assessment is being conducted as a collaborative effort between the Australian Government and the Queensland Government catchment modelling team. This project delivers the *Relative risk of degraded water quality to the GBR* assessment described in this report.

Reef Plan Scientific Consensus Statement 2013

Supporting synthesis of evidence

Marine and coastal ecosystem impacts	Resilience of Great Barrier Reef marine ecosystems and drivers of change	Relative risks to the Great Barrier Reef from degraded water quality	Sources of sediment, nutrients, pesticides and other pollutants in the Great Barrier Reef catchment	The water quality and economic benefits of agricultural management practices
 The impacts of poor water quality on the reef, including habitat (seagrass, coral etc) and biodiversity (dugong etc) Improved evidence of linkages between different aspects of water quality and particular ecosystem impacts (e.g. COTS / DIN linkage etc) Delivery of pollutants during flood events (recent exposure mapping / vulnerability assessment etc) Reduction in coral cover and seagrass over time Time lags in the marine ecosystem (particularly to see improvements) 	 How have recent extreme weather events impacted the reef? What is the impact of other manmade disasters (oil spills, shipping collisions etc)? What can we expect in future? How will this impact management? 	 What is worse - N, P, sediment or pesticides? What are the status of and effects on different components of the marine system? When and where are the risks highest or the benefits of improved management greatest? Note that this Chapter is informed by the risk assessment presented in this report. 	 What is the relative importance of pollutant sources in terms of water quality impact? What proportion of pollutants are delivered through groundwater vs surface water vs overbank flow? What proportion of sediment is from bank erosion, gullies or hillslope erosion? Are fine sediments or coarse sediments being transported? What land types and areas are generating the most pollutants and when? 	 What locations, sub- catchments and soil types contribute the greatest risk? What is the water quality, economic and social benefit from improving management systems and practices? When and where are these best implemented spatially to get maximum water quality benefit at minimum cost? How effective are wetlands and riparian areas in reducing pollutants to the reef?

Figure 1.3. Scope of the Reef Plan Scientific Consensus Statement 2013 in relation to this risk assessment project.

Part A: Assessment of the relative risk of degraded water quality to ecosystems of the GBR

2 Assessment of the risk of pollutants to ecosystems of the Great Barrier Reef including differential risk between sediments, nutrients and pesticides, and among NRM regions

Jane Waterhouse, Jeffrey Maynard, Jon Brodie, Daniel Zeh, Lucy Randall, Stephen Lewis, Caroline Petus, Michelle Devlin, Eduardo da Silva, Miles Furnas, Britta Schaffelke, Katharina Fabricius, Vittorio Brando, Len McKenzie, Catherine Collier, Michael Warne, Rachael Smith, Nyssa Henry, Hugh Yorkston, Dieter Tracey

2.1 Summary of findings

Using the best available knowledge at the time and risk assessment methods that are appropriate within the resources of this study, we have investigated three key aspects of the relative risk of water quality to the GBR: (a) the relative importance of different pollutants to GBR ecosystems (coral reefs and seagrass meadows), (b) the combined risk of degraded water quality on GBR ecosystems, and (c) the relative risk of degraded water quality in the GBR through consideration of the combined risk and influence of end-of-catchment anthropogenic pollutant loads (ie. the current load less the estimated pre-European load) on GBR ecosystems.

The risk assessment method used a combination of qualitative and semi-quantitative information about the influence of individual catchments in the 6 natural resource management (NRM) regions on coral reefs and seagrass ecosystems. The area of GBR lagoon waters within each marine NRM region were also reported in recognition of other important habitats and populations that exist in these areas.

A suite of water quality variables were chosen that represent the pollutants of greatest concern with regards to agricultural runoff and potential impacts on GBR ecosystems. These include ecologically - relevant thresholds for concentrations of total suspended solids (TSS) and chlorophyll *a* (Chl) from daily remote sensing observations, and the distribution of key pollutants including TSS, dissolved inorganic nitrogen (DIN) and photosystem II-inhibiting herbicides (PSII herbicides) in the marine environment during flood conditions (based on end-of-catchment loads and surface water exposure estimates). A spatial variable was included that represents the area of the GBR lagoon where primary crown-of-thorns starfish (COTS) outbreaks have been observed. COTS outbreaks are an important cause of coral loss on the GBR and appear to be a response to excess nutrient runoff from certain catchments that impact this 'COTS initiation zone'. Anthropogenic end of catchment loads were included to link the outcomes of the marine assessment to catchment management. In recognition of the importance of the influence of catchment discharges in driving COTS outbreaks, an index of regional contributions of river discharges to the COTS initiation zone is also included for coral reefs.

For each variable, thresholds above which potential impacts have been observed were defined and classified into three to five classes (from lowest to highest), largely on the basis of the time (or probability) the ecosystem is likely to be exposed to concentrations above the threshold; these are defined as 'assessment classes'.

Key findings:

Relative importance of different pollutants on GBR ecosystems

 The area of coral reefs at highest risk from all of the sediment and nutrient variables (except for the COTS initiation zone) was greatest in the Burdekin and Fitzroy regions. The other regions (Cape York, Wet Tropics, Mackay Whitsunday and Burnett Mary) each had ~20% or less of the coral reef area affected for each variable.

- The area of seagrass at highest risk from the sediment and nutrient-related variables was greatest in the Burdekin region. The area of seagrass within the Wet Tropics region is second greatest for all sediment-related variables but the areas are less than one quarter of the areas affected in the Burdekin region in all cases. For the nutrient variables, the second greatest areas of seagrass within the highest assessment classes are in the Fitzroy.
- The COTS Initiation Zone straddles the boundary between the Cape York and Wet Tropics regions, with approximately 60% of reefs within the Zone located in the Cape York region.
- Of the NRM regions examined in the assessment, the Mackay Whitsunday region presents the highest ecological risk from pesticides with the Photosystem II (PSII) herbicide risk of 'High' and 'Medium' extending off the mouths of the Pioneer and O'Connell Rivers and Sandy Creek. This is followed by the Burdekin (due to the Barratta Creek and Haughton Rivers but not the Burdekin River itself), Wet Tropics, Fitzroy and Burnett Mary NRM regions. It should be noted that the risk to 'pesticides' here is represented by PSII herbicides as these are the dominant pesticides detected in catchments, however a total of 34 pesticides (herbicides, insecticides and fungicides) have been detected. In addition the high risks of PSII herbicides to wetland, estuarine and coastal habitats (which provide important ecosystem services to the GBR including fish nursery habitats), were not included in this stage of the assessment, but they are recognised as important and addressed in Supporting Studies Chapter 4 (Lewis et al. 2013a) and Part B of this report.

Combined risk of degraded water quality to GBR ecosystems

 When all water quality variables are combined, the risk is greatest for coral reefs in the Fitzroy and Mackay Whitsunday regions, and for seagrass in the Burdekin and Fitzroy regions. In most cases, the proportion of the habitat area in each region that is in the highest risk areas are less than 10%, except in the case of seagrass meadows in the Mackay Whitsunday region where 37% of the area of seagrass in the region is affected. This may have significant implications at a regional scale and warrants further consideration. These high risk areas often include highly valued tourism and recreation sites of the GBR. Examples include Fitzroy Island, Hinchinbrook Island, Magnetic Island, many of the islands in the Whitsunday Group and the Keppel Island group. Inshore seagrass meadows are also of critical importance to dugong and green turtle populations in the GBR.

Relative risk of degraded water quality to GBR ecosystems

- A combined assessment of anthropogenic end of catchment pollutant loads and the ecological risk of water quality variables in the marine environment allows us to draw conclusions about the overall risk of pollutants to the GBR. In summary, the greatest risk to each habitat in terms of the potential water quality impact from all of the assessment variables in the GBR and end -of -catchment anthropogenic loads of DIN, TSS and PSII herbicides is:
 - **Coral reefs:** Wet Tropics region, followed by the Fitzroy region. The rank of the remaining regions is the Mackay Whitsunday, Burdekin, Cape York and Burnett Mary region.
 - **Seagrass meadows**: Burdekin region, followed by the Wet Tropics. The rank of the remaining regions is the Fitzroy, Mackay Whitsunday, Burnett Mary and Cape York region.
 - Coral reefs and seagrass meadows combined: Wet Tropics, followed by the Fitzroy (80% of this influence) and Burdekin (76%). The relative risk to the Mackay Whitsunday region is half the greatest risk, followed by the Burnett Mary and Cape York regions. It must be reiterated that these

are relative assessments and therefore indicates that the Burnett Mary and Cape York are low relative to the other regions, but may be exposed to a range of risks that still warrant management to maintain the ecosystem values in these regions.

The risk assessment presented in this Section has generated the most comprehensive assessment of the relative risk of degraded water quality to GBR ecosystems undertaken to date. It is important to reiterate that the rankings between NRM regions are relative, and do not represent absolute differences. In this regard, even the lowest ranked regions of the Burnett Mary and Cape York region may pose a risk to GBR ecosystems, but relative to the other NRM regions they are considered to be of lower risk. This information can be useful in guiding investment priorities but should not be used in isolation from other knowledge related to regionally specific priorities, socio economic influences and limitations to the assessment that may have led to uncertainties in the results. The limitations to the assessment are described in Section 4 of this report, together with ways to improve upon the work presented herein.

2.2 Introduction

This section of the report presents the results of the most recent effort to assess the relative risk of the influence of sediments, nutrients and pesticides on key GBR ecosystems. As outlined in Section 1, the assessment considers the most relevant influences of water quality in the GBR - sediments, nutrients and pesticides.

This report refers to suspended (fine) sediments and nutrients (nitrogen, phosphorus) as 'pollutants'. Within this report we explicitly mean enhanced concentrations of or exposures to these pollutants, which are derived from (directly or indirectly) human activities in the GBR ecosystem or adjoining systems (e.g. river catchments). Suspended sediments and nutrients naturally occur in the environment; indeed, all living things in ecosystems of the GBR require nutrients, and many have evolved to live in or on sediment. The natural concentrations of these materials in GBR waters and inflowing rivers can vary, at least episodically, over considerable ranges. Pesticides do not naturally occur in the environment. Pollution occurs when human activities raise ambient levels of these materials (time averages, or event-related) to concentrations that cause environmental harm and changes to the physical structure, biological communities and biological functions of the ecosystem.

2.2.1 Risk assessment framework

Ecological Risk Assessment (ERA) is a term used for a variety of methods to determine the risk posed by a stressor, for example a pollutant, to the health of an ecosystem. "Risk" is usually defined as the probability that an adverse effect will occur as a result of ecosystem exposure to a certain concentration of the stressor. Risk is often quantified as the product of the *likelihood* of an event occurring (exposure) and the *consequences* (also measured as effects) of that event. Risk assessments are used as decision tools that rank risks to human values in order to prioritise management actions and investments (eg. Burgman, 2005; AS/NZS, 2004). A number of methodologies are available to carry out the analysis with Bayesian techniques now often favoured by decision makers (e.g. Hart et al. 2005; Hart and Pollino, 2008).

In an initial stage of the current project, an approach using a tiered risk assessment methodology that would provide a systematic, objective and transparent approach to quantify the relative risk of contaminants to the GBR ecosystems was developed (Hayes et al. 2012). This tiered ERA approach would have allowed a ranking of the rivers draining into in the GBR based on exceedance of water quality guidelines at marine sites whose water quality could be confidently attributed to individual rivers, and across the GBR lagoon as whole. However, this approach was not able to be used due to limitations in data availability and limitations with time and resources. Hence a simpler methodology suitable for the existing datasets, resources and timeframes has been developed based on a modification of the typical ERA framework.

In this assessment the relative risk of degraded water quality among NRM regions was determined by combining information on the estimated ecological risk of water quality to coral reefs and seagrass meadows in the GBR and end-of-catchment pollutant loads. This approach attempts to relate the water quality conditions in the GBR to catchment based influences, albeit in a relatively crude way.

Ecological risk is assessed using a relatively simple approach. The *likelihood of exposure* of a species or habitat to an impact is typically a function of the intensity of the impact (the concentration or load of a pollutant) and the length of time it is exposed to the impact. For example, a seagrass meadow may be exposed to a high intensity impact for a short period of time (acute), or to lower intensities for longer periods (chronic). When quantifying exposure, it is important to determine the threshold concentrations that lead to an effect on species or habitats, that is, the concentration that potentially leads to damage or mortality within hours or days, as well as understanding long-term average concentrations and the duration of exposure. This complicates the description of exposure thresholds given their values may change by one to two orders of magnitude between days, seasons and years. Hence, some key water quality variables such as suspended sediments are divided into different thresholds based on ecological responses and periods of exposure. To reflect this, each threshold is classified into several assessment classes to represent the potential differences between the duration and severity of the influence (from lowest to highest).

The *consequences* are the measured effects of the water quality exposure. Current knowledge of the effects of degraded water quality on the health of the GBR are summarised in the 2013 Scientific Consensus Statement. The GBR Water Quality Guidelines reflect our knowledge of ecological thresholds for water quality variables for coral reefs in the GBR (GBRMPA, 2009). However, only limited information is available to draw conclusions on the effects of the exposure of sediments, nutrients and pesticides on seagrass health. Evidence shows that one of the greatest drivers of seagrass health is the availability of light, which is reduced by increased suspended sediment and the secondary effects of increased nutrients such as increased growth of epiphytes and phytoplankton (Collier et al. 2012). However, in the absence of more regionally - and species-specific knowledge of pollutant impacts on seagrass, the same threshold concentrations have been used for coral reefs and seagrass meadows in this assessment. It is also recognised that the consequence of the exposure of species or habitats to a range of water quality conditions is complicated by the influence of multiple pressures, and many external influences including weather conditions, however it is difficult to factor these into the risk assessment in any quantitative way.

Given the above and recognising the inconsistencies in the spatial and temporal availability of the water quality data, our capacity to produce a true likelihood or true consequence estimate for this assessment is limited. It was therefore necessary to develop an effective, simple and standard methodology for the risk assessment that could be implemented with the available data, in a way that could be easily communicated and discussed with decision-makers and stakeholders. For this reason, ecological risk in the GBR is expressed simply as the area of coral reefs and seagrass meadows within a range of assessment classes (very low to very high relative risk) for several water quality variables in each NRM region in the GBR catchment. Our method for calculating risk essentially assesses the likelihood of exceedance of a selected threshold. This likelihood was set as 1 for a parameter and location if observations or modelled data indicate that the threshold was exceeded. Conversely, the likelihood was set as 0 if observations or modelled data indicate that the threshold was not exceeded. As consequences are mostly unknown at a regional or species level, potential impact was calculated as the area of coral reef, seagrass meadows and area of GBR lagoon waters (in km²) within the highest assessment classes of the water quality variables (reflecting the highest severity of influence). The effects of multiplying the habitat area by 1 or 0 for the likelihood mean that the final assessment of risk in this assessment is only an indication of potential impact - the area of coral reef and seagrass meadows in which exceedance of an agreed threshold was modelled or observed. This becomes an assessment of 'relative risk' by comparing the areas of each habitat

affected by the highest assessment classes of the variables among NRM regions, and was used to generate a 'Marine Risk Index'.

Modelled end-of-catchment pollutant loads (generated from the Source Catchments model framework for the Paddock to Reef Program) were obtained for each NRM region for key pollutants, and only the anthropogenic portions of total pollutant loads were considered. The anthropogenic load is calculated as the difference between the long term average annual load, and the estimated pre-European annual load. This information was used to define a 'Loads Index'.

The variables included ecologically relevant thresholds for concentrations of total suspended solids (TSS) and chlorophyll *a* from daily remote sensing observations, and the distribution of the loading of key pollutants including TSS, dissolved inorganic nitrogen and photosystem II-inhibiting herbicides (PSII herbicides) in the marine environment during flood conditions (based on an assessment of flood plume frequency and predicted distribution of end-of-catchment loads). A spatial variable was included that represents the area of the GBR lagoon where primary crown-of-thorns starfish (COTS) outbreaks have been observed. COTS outbreaks are an important cause of coral loss on the GBR and appear to be a response to excess nutrient runoff from certain catchments that impact this 'COTS initiation zone'. In recognition of the importance of the influence of catchment discharges in driving COTS outbreaks, an index of regional contributions of river discharges to the COTS initiation zone is also included for coral reefs (COTS Influence Index).

The three indexes were then combined to generate a Relative Risk Index for coral reefs and seagrass, which ultimately ranks the relative risk of degraded water quality to coral reefs and seagrass meadows in the GBR among NRM regions. The framework and how it relates to the typical ERA framework is shown in Figure 2.1. As described in the Introduction, the outputs of this assessment provide an integral component of the overall management and investment prioritisation being undertaken for Reef Plan 3 (incorporating Reef Rescue 2). A more detailed illustration of the overall process incorporating this assessment is shown in Figure 2.2.

The geographic boundaries of the assessment and the spatial distribution of the marine habitats in the assessment (coral reefs and seagrass - based on best available information) are shown in Figure 1.1. The area of GBR lagoon waters in each NRM region was also included as it contains other important habitats and biological populations such as fish and benthic organisms, however this was not included in the overall Relative Risk Index as assumptions regarding the importance of the potential impact on the wide range of ecosystems in the GBR are unknown. The marine boundaries used for each NRM region are those accepted officially by the Great Barrier Reef Marine Park Authority.



Figure 2.1. The risk assessment framework used in this project showing the components of the Marine Risk Index to represent marine water quality ecological risk to coral reefs and seagrass meadows, a Loads Index to represent catchment influences on GBR water quality using end of catchment anthropogenic pollutant loads and a COTS Influence Index to factor in the importance of river discharges on the COTS Initiation Zone for coral reefs.



Management Prioritisation

- Step 1: Assess ecological risk by NRM regions, pollutants and coral reefs & seagrass meadows based on presence of habitat & exceedance of thresholds / exposure – Marine Risk Index
- Step 2: Assess river influence by Regions, pollutants and habitats = Loads Index + COTS Influence Index (for coral reefs only)
- Step 3: Combined assessment of the Marine Risk Index + Loads Index + COTS Influence Index (coral reefs only) = Relative Risk Index by region and habitats
- Step 4: Combine related information on priority pollutant sources and land use with Relative Risk Index to generate management priorities and overall relative risk ranking among NRM regions

Figure 2.2. The overall framework being used for Reef Plan 3 (incorporating Reef Rescue 2) management and investment prioritisation. This project contributes Steps 1 to 3 and part of Step 4. An additional assessment of the 'solvability' of management issues in different NRM regions and catchment areas is being conducted by the Australian Government. The assessment is based on recent management practice adoption data for the sugar cane and grazing industries in the GBR catchments, and the cost effectiveness of improvements in water quality of the various practices.

2.3 Methods

A 3-part approach of estimating the relative risk of pollutants to the GBR at a regional level was applied (illustrated in Figure 2.3):

- Assessment of the relative importance of different pollutants on GBR ecosystems (coral reefs and seagrass). This identifies the areas where each water quality variable is considered to pose the greatest relative risk to coral reefs and seagrass between the NRM regions. The output can be used to guide priorities for management of individual pollutants between NRM regions. The methods are described in Section 2.3.3.
- 2. Combined risk of degraded water quality to GBR ecosystems. The combined assessment takes into account all assessment classes for each variable to generate a Marine Risk Index for coral reefs and seagrass. The areas within the Risk Index represent the areas of highest relative risk to degraded water quality in the GBR and identify the areas where coral reefs and seagrass are most likely to be under pressure from degraded water quality. The methods are described in Section 2.3.4.
- 3. The relative risk of degraded water quality to GBR ecosystems. This relates the results of Part 1 and Part 2 to land based influences using an assessment of end-of-catchment anthropogenic loads and river discharges (Loads Index and COTS Influence Index). These results inform the regional management priorities required to address the risks identified in Part 1 and Part 2 in terms of where to focus effort on which pollutants. The methods are described in Section 2.3.5.

Justification for the selection and classification of variables is provided in Section 2.3.1. Section 2.3.2 describes the habitat areas in each region.

2.3.1 Selecting and classifying variables

We have chosen a suite of water quality variables that represent the pollutants of greatest concern with regards to agricultural runoff and potential impacts on GBR ecosystems. Ecological impacts of terrestrial runoff on coral reefs and seagrasses beds can be experienced as either acute, short term changes associated with formation of high-nutrient, high-sediment, low salinity flood plumes or the more chronic impacts associated with changes in long-term water quality concentration (Devlin et al. 2012). The ecological impact of terrestrial pollutants varies not only with the type of pollutant, the magnitude and extent of the riverine influence but also with the ecosystems being affected and the frequency and duration of plume occurrence (e.g. Devlin et al. 2013a). Long time series of pollutant concentration data provides a way of assessing chronic stress, while river plume models can help to develop risk maps by defining areas which may experience acute or chronic high exposure to pollutants or stressors (Alvarez-Romero et al. 2013). Details of the pollutant movement and frequency of inundation can be key measurements in attributing water quality decline to ecosystem change. This assessment uses a combination of variables that represent chronic and acute stress.



Figure 2.3. Illustration of the steps in the assessment of the relative risk of water quality pollutants to GBR ecosystems.

The selected variables are summarised in Table 2.1. These include ecologically relevant thresholds for concentrations of total suspended solids (TSS) and chlorophyll *a* from daily remote sensing observations, and the distribution of key pollutants including TSS, dissolved inorganic nitrogen (DIN) and photosystem II-inhibiting herbicides (PSII herbicides) in the marine environment during flood conditions (based on end-of-catchment loads and surface water exposure estimates). A spatial variable is included that represents an area of the GBR lagoon where primary crown-of-thorns starfish (COTS) outbreaks have most frequently been observed (see Chapter 1 of the Supporting Studies, Furnas et al., 2013a). COTS outbreaks are an important cause of coral loss on the midshelf and outer reefs of the GBR (De'ath et al. 2012) and are, based on current understanding, a response to excess nutrient runoff from certain catchments that reaches this 'COTS initiation zone' (Fabricius et al. 2010). The relevance of each of these variables is described below. More detailed information on pollutant impacts GBR ecosystems is provided in the recently completed Scientific Consensus Statement *Chapter 1 Marine and coastal ecosystem impacts from degraded water quality* (Schaffelke et al. 2013).

For each variable, thresholds above which impacts have been observed or predicted were defined and classified into three to five classes (from lowest to highest). The classification of each variable is described in Section 2.3.1a, 2.3.1b and 2.3.1c below. The selected variables and thresholds represent long-term conditions (chronic exposure) and wet season pollutant loadings in flood plumes (acute exposure).

Additional variables were considered that have not been included here due to the current lack of data showing their temporal and spatial patterns and ecological impacts. These include: phosphorus exposure, chronic exposure to PSII herbicides and non-PSII herbicides, and time series of pesticide concentration data. The decision to select DIN as primary nutrient variable within the assessment is supported by the conclusions of Chapter 2 of the Supporting Studies (Furnas et al. 2013b) which considered the relative importance of nutrient forms and of nitrogen and phosphorus in the GBR. The analysis indicates that dissolved inorganic and particulate forms of nutrients discharged into the GBR are both important in driving ecological effects but increased nitrogen inputs are more important than phosphorus inputs. Dissolved inorganic forms of nitrogen and phosphorus of greatest concern compared to dissolved organic and particulate forms of nutrients, as they are immediately and completely bioavailable for algal growth (see Furnas et al. 2013b). Particulate forms mostly become bioavailable over longer time frames, and dissolved organic forms typically have limited and delayed bioavailability (see Furnas et al. 2013b).

For each of the variables shown in Table 2.1 a classified spatial data layer was prepared. ArcGIS was the primary tool used for spatial analysis, however, the Multi-Criteria Analysis Shell for Spatial Decision Support (MCAS-S¹) was also used for some aspects of the assessment. MCAS-S was initially the preferred tool as it was also used for the associated Reef Rescue 2 Investment Prioritisation process conducted by the Australian Government. However, MCAS-S is a raster-based tool that therefore loses a degree of resolution of some spatial layers depending on the selected grid size. Using a 1 km² grid, this was problematic for the assessment of the area of coral reefs because the area was considerably different after the raster conversion. A number of approaches were tested to overcome these limitations including a presence/absence approach for coral reefs within a grid cell, but this resulted in significant overestimation of the areas especially in the inshore areas where reef sizes are typically small or there are many fringing reefs. The differences between regions were not comparable and therefore, significantly distorted the results of the assessment. Therefore ArcGIS was used for all area estimates (in km²) for all variables as it is possible to calculate the area within the reef polygon; seagrass habitat is already on a 1 km² grid.

¹ MCAS-S: Multi-Criteria Analysis Shell for Spatial Decision Support developed by ABARES, refer to http://www.daff.gov.au/abares/data/mcass

MCAS-S is a useful tool for spatial analysis of datasets that has adopted much of the functionality of the ArcGIS interface but that allows independent data processing and is relatively easy to use. MCAS-S functionality promotes participatory processes that require the understanding of relationships between decision- making requirements and the available data. Of particular value in this regard are software features that enable interactive cognitive and 'live up-date' mapping of alternative views (Lesslie et al. 2008).

As described in Lesslie et al. (2008), MCAS-S requires the user to prepare raw spatial data for a project in a raster format of consistent geographical extent, pixel resolution and projection. Primary data can then be selected from the menu and dragged into a display workspace whereupon the spatial data layer can be classified using a variety of simple classification methods and tools for tailoring input data. With the creation of individual class or rule layers, the user can apply them in weighted combinations to construct composite layers contributing to themes of interest. Key functions for combining and comparing spatial layers are represented schematically in Figure 2.4.



Figure 2.4. Flow chart for MCAS-S showing stages and functionality. Source: Derived from Lesslie et al. (2008).

Variables		As	ssessment Cla	ISS		Data source/methodology	
	Very Low 1	Low 2	Medium 3	High 4	Very High 5		
Sediments							
Total Suspended Solids (TSS) concentration (mg/L)						Based on daily observations of TSS in the period 1 Nov 2002 to 30 April 2012. Data has been interpolated across reefs (which are masked during image processing) using Euclidean Allocation in ArcGIS. Classification of frequency of exceedance is	
Frequency of exceedance %						for extraction described in Brando et al. (2013).	
Threshold a: 2 mg/L	<1	1-10	10-20	20-50	50-100	Threshold correlates strongly with declines in ecosystem condition such as increased macroalgal growth and declining diversity. Average annual threshold for TSS in the Great Barrier Reef Water Quality Guidelines. Refer Section 2.3.1a(i)	
Threshold b: 7mg/L	0	<1	1-10	10-20	20-100	Threshold is equivalent to a turbidity of 5 nephelometric turbidity units (NTU). Shown to have various ecosystem effects including coral reef stress, declines in seagrass cover (Collier et al. 2012), fish habitat choice, home range movement and (above 7.5 nephelometric turbidity units) foraging and predator-prey relationships (Wenger et al. 2013). Refer Section 2.3.1a(ii).	
TSS Plume Loading (mean 2007-2011)	Catego	ry 1	Category 2	Category 3		The frequency and extent of the influence of flood plumes containing differing concentrations of total suspended solids is used to provide an estimation of the extent of surface exposure of coral reefs and seagrass during wet season conditions. Modelled using an assessment of plume frequency from satellite imagery and monitored end of catchment loads in each wet season (Nov to May) from 2007 to 2011 (Devlin et al, 2013a). The mean of the five annual maps was selected as a way of factoring in inter-annual variability in river discharge, although it is recognised that this period was characterised by several extreme rainfall events. Refer Section 2.3.1c(v).	

Table 2.1. Summary of water quality variables.	assessment classes and data sources	s included in the marine risk assessment.						
Variables	Assessment Class					Data source/methodology		
----------------------------------------------------------------------------	--------------------	----------	---------------	-----------	-------------------	---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	--	--
	Very Low 1	Low 2	Medium 3	High 4	Very High 5			
Nutrients								
Chlorophyll concentration (µg/L)						Assessment classes were based on daily observations of Chlorophyll concentrations over the period 1 Nov 2002 to 30 April 2012. Data was interpolated		
Frequency of exceedance %						Allocation in ArcGIS. Classification is based on the number of valid observations in the full observation period. Method for extraction described in Brando et al. (2013).		
0.45 μg/L	<1	1-10	10-20	20-50	50-100	Chlorophyll is an indicator of nutrient enrichment in marine waters. De'ath and Fabricius (2008) identified 0.45 μ g/L as an important ecological threshold for macroalgal cover, hard coral species richness, octocoral species richness. Annual average threshold for chlorophyll in the Great Barrier Reef Water Quality Guidelines. Significant benefits for the ecological status of reefs in the Region are likely if mean annual chlorophyll concentrations remain below this concentration. Refer Section 2.3.1b(iii).		
Dissolved Inorganic Nitrogen (DIN) Plume Loading (mean 2007-2011)	Catego	ry 1	Category 2	Cate	gory 3	Elevated DIN is an indicator of nutrient enrichment. High concentrations of DIN can reduce coral recruitment (Babcock and Davies 1991; Loya et al. 2004), enhance coral bleaching susceptibility (Wooldridge and Done, 2009) and change the relationship between coral and macroalgal abundance (De'ath and Fabricius, 2010). Elevated concentrations can also be deleterious to seagrass by lowering ambient light levels via the proliferation of local light absorbing algae thereby reducing the amount of photosynthesis in seagrass, particularly in deeper water (Collier, 2013).		
						Modelled using an assessment of plume frequency from satellite imagery and monitored end of catchment loads in each wet season (Nov to May) from 2007 to 2011 (Devlin et al, 2013a). The mean of the five annual maps was selected as a way of factoring in inter-annual variability in river discharge, although it is recognised that this period was characterised by several extreme rainfall events. Refer Section 2.3.1c(v).		
COTS Initiation Zone	Out of the zone				In the Zone	Shows an area defined to be highest risk in initiating COTS outbreaks, defined as the area between Latitude 14.5°S and 17°S and described in Furnas et al. (2013a). Data from this area shows prolonged periods of high Chl concentrations that exceed 0.8 μ g/L, which is important for COTS larval survival. Refer Section 2.3.1b(iv).		

Variables	Assessment Class			ass		Data source/methodology		
	Very Low 1	Low 2	Medium 3	High 4	Very High 5			
Pesticides								
PSII Herbicide modelled concentration (μg/L)	0.025-0.1	0.1-0.5	0.5-2.3	2.3-10	>10	Based on an estimate of the relationship between Colour Dissolved Organic Matter (CDOM) and salinity, and then a modelled salinity to PSII herbicide concentration relationship in a flood plume event in one river in each NRM region in 2009-2011. Data has been interpolated across reefs (which are masked during image processing) using Euclidean Allocation in ArcGIS. Risk posed was determined using a number of methods - some only assessed acute toxic effects, others both acute and chronic. Described in Lewis et al. (2013a). Refer Section 2.3.1b(iv).		
						 >0.025-0.1 μg/L: No observable effect; 0.1-0.5 μg/L: Photosynthesis is reduced by up to 10% in corals (Negri et al. 2011); seagrass (Haynes et al. 2000; Chesworth et al. 2004; Gao et al. 2011; Flores et al. in review) and microalgae (Magnusson et al. 2008, 2010). The effect on primary production is minor. 0.5-2.3 μg/L: Photosynthesis is reduced by between 10% and 50% in corals (Negri et al. 2011); seagrass (Haynes et al. 2000; Chesworth et al. 2004; Gao et al. 2011; Flores et al. in review) and microalgae (Magnusson et al. 2008, 2010). The community structure of tropical microalgae (Magnusson et al. 2008, 2010). The community structure of tropical microalgae can be affected by concentrations of diuron as low as 1.6 μg/L (Magnusson et al. 2012). The effect on primary production is moderate. 2.3-10 μg/L Photosynthesis is reduced by between 50% and 90% in corals (Jones and Kerswell, 2003; Negri et al. 2011); seagrass (Chesworth et al. 2004; Gao et al. 2011; Flores et al. in review) and microalgae (Magnusson et al. 2008, 2010). A 50% reduction of growth and biomass of tropical microalgae was also reported in this concentration range (Magnusson et al. 2008). The community structure of tropical microalgae is significantly affected and this causes significant changes in the tolerance of microbial communities to herbicides (Magnusson et al. 2012). The effect on primary production is major. > 10 μg/L: reduced growth and mortality in seagrass (Gao et al. 2011) and loss of symbionts (bleaching) in corals (Jones et al. 2003; Negri et al. 2015). 		

a) Exceedance of suspended solids concentration thresholds

The effects of elevated concentrations of suspended solids on GBR ecosystems including coral, seagrass and algal communities are described in Chapter 3 of the Supporting Studies (Brodie et al. 2013a). The greatest influence of increased turbidity caused by resuspension of sediment in waters of depths less than 12 metres is reduced light for benthic phototropic communities including coral reefs and seagrass (Larcombe et al. 1995; Anthony et al. 2004; Orpin et al. 2004; Alongi and McKinnon, 2005). This resuspension driven turbidity persists for many months of the year in GBR coastal waters. Suspended solids in flood plumes also reduce light for benthic communities but the effects are only present for short periods, typically days to weeks. Hence, a long term time series is most relevant in the assessment of chronic effects of elevated suspended solids and turbidity on habitats. However, typically the resuspended sediment is that which was delivered as a sediment loading during the previous wet season and potentially earlier wet seasons as well. Hence, there is a strong connection between turbidity and river loadings of sediment (Fabricius et al. 2013). The TSS plume loading modelling (Devlin et al. 2013a) allows us to assess loadings and in a sense predict the likely conditions of suspended sediment in various areas of the GBR lagoon. When fine sediment is delivered to shallow areas less than 12 metres, it is a good indicator for likely resuspension later in the year. Therefore, both the concentration data and the loading data are relevant to this assessment. The actual exposure of benthic organisms (for example in Cleveland Bay) to flood plume turbidity is more relevant for the assessment of acute effects (see Devlin et al. 2013b).

Method:

Using remote sensing imagery in the period 2002 to 2012, we defined the areas where the TSS concentration exceeded of the different ecologically-relevant threshold values at different frequency intervals (see Table 2.2). The method for the retrieval and processing of the remote sensing data is described in Brando et al. (2013) and is summarised below.

Data collected by MODIS Aqua provide a time series from 1 November 2002 to 30 April 2012 of water quality estimates with spatial coverage at 1 km² resolution for the whole-of-GBR lagoon, nominally on a daily basis (except overcast days). The water quality estimates were retrieved from the MODIS Aqua time series using two coupled physics-based inversion algorithms developed to accurately retrieve water quality parameters for the optically complex waters of the GBR lagoon (Brando et al. 2008; Schroeder et al. 2008; Brando et al. 2010a,b; Brando et al. 2012; Schroeder et al. 2012). This was necessary because chlorophyll concentrations retrieved with the MODIS standard algorithms provided by NASA are up to two-fold inaccurate in GBR waters (Qin, Dekker et al. 2007), while CSIRO's regionally parameterised algorithms account for the significant variation in concentrations of Colour Dissolved Organic Matter (CDOM) and TSS and achieve more accurate retrievals (Brando e al. 2010a,b). For this work the whole MODIS Aqua time series was reprocessed with the most recent updates in NASA's software (SeaDAS version 6.4), incorporating the improved knowledge of instrument temporal calibration to improve temporal stability of the time series of the MODIS Aqua ageing sensor.

The comparison of MODIS Aqua retrievals of ChI, CDOM and turbidity data to *in situ* data showed that the a-LMI water quality algorithm coupled with the ANN atmospheric correction is more accurate than NASA's algorithms for GBR waters (Brando et al. 2013). The parameterisation and validation on the remote sensing retrievals was mainly based on observations performed in coastal and lagoonal waters during the dry season between Keppel Bay and the Wet Tropics region. The accuracy of the retrieval is likely to be lower in shallow and turbid waters systems such as Princess Charlotte Bay, Broadsound and Shoalwater Bay, as there is no data available for parameterisation and validation. Details on the algorithm's theoretical basis, parameterisation and validation are provided in Brando et al. (2013).

The frequency of exceedance of the TSS threshold was calculated by analysing 'daily' observations of TSS concentrations at a scale of 1 km² pixels. The assessment classes were defined using the total potential number of observations, and the maximum number of valid observations in the assessment period (Table 2.2). The low number of valid observations is a result of the strict quality control criteria applied to the imagery: pixels with cloud or cloud shadow, low view and illumination angles (solar zenith and observer zenith higher than 60 degrees) are flagged and dismissed as are pixels where the atmospheric correction failed. In the wet season a valid observation is obtained approximately 1 in every 5 days (23%), while for the dry season valid observations were obtained approximately 2 of every 5 days (41%), equating to less than 2 valid observations is 3,650. The assessment classeswere defined using expert opinion, informed by Jenks natural breaks on the basis of the frequency of exceedance of the thresholds (<1%, 10%, 20%, 50% and 100%) as a proportion of the total number of valid observations.

Table 2.2. Number of valid remote sensing observations (in the context of the total potential observations) throughout the assessment period, and the frequency of exceedances used to define assessment classes for the assessments. The selected variables are all Annual and highlighted in grey shading.

	Num	ber of observat	tions
Item	Wet		
	season	Dry Season	Annual
	10 yrs	10 yrs	10 yrs
No Pixels (maximum potential observations)			
1 pixel = 1 day	1820	1820	3650
Valid observations (pixels with data in assessment period)	427	755	1182
Actual as proportion of potential observations	23%	41%	32%
Frequency of exceedance (based on valid observations)			
<1% exceeding thresholds	<4	<7	<10
10% exceeding threshold	43	76	118
20% exceeding threshold	85	151	236
50% exceeding threshold	214	378	591
100% exceeding threshold	427	755	1182

Assessment classes:

Several ecologically-relevant values were explored for TSS using the wet season, dry season and annual datasets over the 10-year period. The threshold concentrations considered in the assessment are the GBR Water Quality Guideline value of 2 mg/L and turbidity 5 NTU (which equates to TSS 6.6 mg/L but is rounded to 7 mg/L for reporting only). The justification for these is described below and further detailed in Chapter 3 the Supporting Studies (Brodie et al. 2013a).

i Threshold a - TSS 2 mg/L

The assessment classes for TSS were obtained from the annual dataset from the frequency of exceedance (expressed as a percentage) of the *Threshold a* concentration of 2 mg/L (1.5 NTU) which is the TSS Water Quality Guideline value (GBRMPA, 2009). Fabricius (2011) reviewed the factors determining the vulnerability of specific coral reefs to damage from turbidity and sedimentation. The study found that well-flushed locations with strong currents, shallow reef crests surrounded by a deep water body, and reefs inhabited by healthy populations of

fishes are likely to have the highest levels of resistance and resilience to degradation from sedimentation and high or variable turbidity. In contrast, damage is most likely to occur in locations with weak currents such as embayments (e.g. Keppel Bay, Cleveland Bay, Missionary Bay), and within these sheltered zones on deeper reef slopes, in places where fish abundances are low, and in regions that are frequently affected by other forms of disturbance such as cyclones , bleaching or crown-of-thorns starfish predation. Despite these complications, water clarity with an annual mean of less than 10 m Secchi depth (or TSS concentration greater than 2 mg/L), has been found to relate to strongly altered ecosystem properties throughout the GBR. These values have therefore been adopted as Water Quality Guidelines for inshore waters by the Great Barrier Reef Marine Park Authority (GBRMPA, 2009).

As described above, the degree of exposure of an organism or ecosystem to a stressor is typically a function of the both amount (= concentration, level or load) of the stressor, and the length of time it is in contact with the stressor (Figure 2.5). A reef may be exposed to a high level of a stressor for a short period of time, or to lower levels for longer periods. When quantifying exposure levels, it is therefore important to determine peak concentrations (potentially leading to damage or mortality within hours to days), as well as to quantify long-term mean (or median) concentrations and the duration of exposure. This complicates the definition of exposure thresholds in turbidity and sedimentation given their values may change by one to two orders of magnitude between days, seasons and years. It is for these reasons that we have considered two different thresholds of TSS in this assessment.



Figure 2.5. Conceptual representation of the exposure of corals to turbidity and suspended solids concentration, and the severity of response, are a function of exposure time and the concentration of the stressor (from Fabricius 2011). The red dotted line shows the GBR WQ Guideline value for TSS of 2 mg/L.

The data extracted from the remote sensing assessment was classified into the assessment classes shown in Table 2.1 are based on the total number of daily observations that exceeded the threshold value over the 10-year assessment period as a proportion of the total number of valid observations (Table 2.2).

ii Threshold b - TSS 7 mg/L (turbidity 5 NTU)

The annual assessment data for TSS 6.6 mg/L (rounded to 7 mg/L for reporting here) was used in the analysis and was correlated with effects on several ecosystems:

- Coral reefs Cooper et al. (2008) suggest long-term turbidity concentrations >3 NTU lead to sub-lethal stress, whereas long-term turbidity concentrations >5 NTU correspond to severe stress effects on corals at shallow depths.
- Seagrass change in seagrass cover at Picnic Bay showed a strong correlation between a decline in seagrass cover and mean turbidity greater than 4 NTU over several days (Collier et al. 2012; Chapter 6 of Supporting Studies, Devlin et al. 2013b). However, this effect was observed using a small dataset and requires further investigation.

• Fish – Wenger et al. (2013) showed that turbidity of 5 NTU was the threshold for fish habitat choice and home range movement, and further Wenger et al. (2012) showed that 7.5 NTU was the threshold for foraging ability and changes in predator prey relationships.

The data extracted from the remote sensing assessment was classified into the 'frequency of exceedance' classes shown in Table 2.2. These classes then were used to establish the relative severity classes (very low to very high) shown in Table 2.1 and are based on the proportion of daily observations that exceeded the threshold value over the 10-year assessment period. Note that the classes are different to those for threshold 1 and reflect the increased severity of the impacts associated with this higher TSS concentration threshold.

b) Chlorophyll concentration exceedance

Chlorophyll (Chl) concentrations are relevant year round as an indication of nutrient enrichment in marine waters. Chl concentrations are particularly relevant when considering drivers of Crown of Thorn Starfish (COTS) outbreaks. The threshold concentrations for Chl related to adverse effects on coral reefs are well established, but only preliminary results are available to relate Chl concentrations to seagrass health. Further discussion of the impacts of nutrient enriched waters on GBR ecosystems is described in the *Reef Plan SCS Chapter 1 Marine and coastal ecosystem impacts* (Schaffelke et al. 2013).

Method:

Using remote sensing imagery in the period 2002 to 2012, we defined the areas where the Chl concentration exceeded of the different ecologically-relevant threshold values at different frequency intervals (see Table 2.2). The method for the retrieval and processing of the remote sensing data is described in Brando et al. (2013) and is the same as that applied for TSS exceedance described in Section 2.3.1a above.

Assessment classes:

Several ecologically-relevant values were explored for Chl using the wet season, dry season and annual datasets over the 10-year period. These ranged between 0.13 μ g/L which considered to be insignificant in terms of potential ecological impacts, various concentrations within the GBR Water Quality Guidelines (seasonal and cross shelf differences) including 0.45 μ g/L, and the highest threshold of 2 μ g/L which is known to be associated with severe ecosystem impacts. Based on expert opinion and data availability the final concentration included in the assessment was Chl 0.45 μ g/L, described below.

iii Chl 0.45 μg/L

Syntheses and interpretation of long-term Chl a and turbidity datasets were presented in De'ath and Fabricius (2008) to underpin the development of the GBR Water Quality Guidelines. For Chl, an analysis of the response of macroalgal cover, species richness of hard corals, and species richness of phototrophic and heterotrophic octocorals, coupled with an assessment of the spatial distribution of water quality, concluded that there would be significant benefits ecological status of reefs in the GBR if the mean annual Chl concentration was kept below 0.45 μ g/L in both coastal and inner shelf zones in all regions.

In addition, a variety of experimental, modelling and observational evidence indicates that initiation of COTS outbreaks is coupled to enhanced survival of the pelagic larval phase as a result of increased food availability, particularly if Chl concentrations are greater than $0.45 \mu g/L$.

For these reasons we selected the value of Chl 0.45 μ g/L as the most appropriate threshold representing long-term average conditions.

There is less information available on the relationship between seagrass health and chlorophyll concentrations however it is clear that elevated chlorophyll concentrations increase light attenuation and reduce light availability to seagrasses (Chapter 7 of Supporting Studies, Collier, 2013). However, preliminary assessments by Collier et al. (in review) show that decline in seagrass cover is correlated with increasing average daily Chl concentrations above $0.3 \ \mu g/L$, however, there is insufficient information to differentiate this response from those observed at the Guideline value of $0.45 \ \mu g/L$. This value has been adopted in the GBR Water Quality Guidelines as also relevant to seagrass, and in the absence of any further information has also been used in this risk assessment for seagrass.

The data extracted from the remote sensing assessment was classified into the 'frequency of exceedance' classes shown in Table 2.2. These classes were used to establish the relative severity classes (very low to very high) shown in Table 2.1 and are based on the total number of daily observations that exceeded the threshold value over the 10-year assessment period.

iv COTS specific assessments

Two of the potential threshold concentrations for Chl obtained from the remote sensing data were based on the relationship between survival of COTS larvae and Chl concentrations (see Fabricius et al. 2010 and Figure 2.6 below). These values were Chl concentration of 0.8 μ g/L where COTS larval survival starts to increase, and 2 μ g/L where 100% COTS larval survivorship occurs.





The case study analysis presented in Chapter 1 of the Supporting Studies (Furnas et al. 2013a) strongly supports the hypothesis that river discharge (and associated nutrient loads) is the primary driver of regional chlorophyll concentrations within the GBR lagoon. The critical factor is the amount of discharge during the early part of the wet season (November-February) when COTS larvae are present in the water column. There appears to be an important threshold on the order of 10,000,000 ML (10 km³) of discharge in this period to produce sufficient phytoplankton at the proper time to sustain high COTS larval survival and promote a subsequent COTS outbreak. Further development of a COTS outbreak, however, depends on there being sufficient coral cover to sustain local adult populations.

Based on this information and the importance of the timing of wet season discharges, it would have been beneficial to undertake further analysis of the long term Chl dataset for the period 1 November to 28 February each year over the 10 years of available data. However, given the limitation of time and resources associated with the project it was not able to be completed in time for this assessment.

In the absence of this more detailed assessment of Chl concentrations in specific periods, a high risk area for the initiation of COTS primary outbreaks has been defined. A variety of experimental, modelling and observational evidence indicates COTS primary outbreaks are initiated by an episode of greatly enhanced larval survival due to increased food availability (chlorophyll concentrations >0.45 µg/L) for the filter-feeding pelagic larval stage (Brodie et al. 2005). Observational evidence indicates that primary COTS outbreaks originate on reefs within a ~250 km sector between Cairns and Lizard Island (17.2°–14.5°S latitudes). Primary COTS outbreaks in the Cairns-Lizard Island region since 1960 (n=4) have occurred 2-5 years after wet seasons when aggregate early wet season (November-February) discharge from the Burdekin to Daintree Rivers exceeded 10 km³. The 2-5 year delay is consistent with the time it takes for COTS to grow up from small (<0.5 cm) recruiting juveniles with cryptic behaviour at settlement to conspicuously large (>50 cm) and non-cryptic adults of breeding size that produce large feeding scars. The available data and more recent satellite imagery shows that extensive flooding from central GBR and Wet Tropics rivers leads both to large nutrient inputs (Furnas, 2003) and widespread phytoplankton growth (blooms) over large areas of the GBR shelf, including the region between Cairns and Lizard Island. Regional chlorophyll concentrations in the nominal outbreak initiation region (1989-2012) are inversely related to regional salinity which is most strongly influenced by wet season rainfall and river runoff (more rainfall = more runoff = lower salinities). Regional chlorophyll concentrations during the critical pelagic larval period (Nov-Feb) are positively correlated with estimates of concurrent river runoff at both the local and regional scales (Furnas et al. 2013a). The 'high-risk' area for primary COTS outbreaks between 14.5°S and 17°S contains approximately 200 reefs with an aggregate outer reef slope circumference close to 2,000 km (GBRMPA GIS Group, personal communication).

In this assessment, we have used a binary assessment approach where the reefs are either within or outside of the COTS Initiation Zone. As shown in Table 2.1, the areas are either classified as Very Low or Very High depending on whether the reefs are located within the COTS Initiation Zone.

c) Pollutant loading in river plumes

Ecological impacts of terrestrial runoff on coral reefs and seagrass meadows can be experienced as either acute, short term changes associated with formation of high-nutrient, high-sediment, low salinity flood plumes or the more chronic impacts associated with long-term changes in water quality (Devlin et al. 2012). The ecological impact of terrestrial contaminants varies not only with the type of pollutant, the magnitude and extent of the riverine influence but also with the ecosystems being affected and the frequency and duration of plume occurrence (see for example Devlin et al. 2012). River plume models can help to develop risk maps by defining areas which may experience acute or chronic high exposure to pollutants or stressors (Alvarez-Romero et al. 2013). Details of the pollutant movement and frequency of inundation can be key measurements in attributing water quality decline to ecosystem change. These contribute to the 'likelihood' component of the risk equation.

Land sourced runoff containing elevated nutrient concentrations results in flood plumes in the GBR lagoon which may result in a range of impacts on coral communities (Fabricius et al. 2005; Fabricius, 2011; Brodie et al. 2011). Dissolved inorganic and particulate forms of nutrients discharged into the GBR are both important in driving ecological effects but increased nitrogen inputs are more important than phosphorus inputs (see Chapter 2 of the Supporting Studies, Furnas et al. 2013b). Dissolved inorganic forms of nitrogen and phosphorus are considered to be of greatest concern compared to dissolved organic and particulate forms of nutrients, as they are immediately and completely bioavailable for algal growth. Particulate forms mostly become bioavailable over longer time frames, and dissolved organic forms typically have limited and delayed bioavailability (Furnas et al. 2013b).

Most studies in GBR waters show that high levels of dissolved inorganic nitrogen and phosphorus can cause significant physiological changes in corals, but do not kill or greatly harm individual coral colonies (reviewed in

Fabricius, 2005). However, exposure to dissolved inorganic nitrogen can lead to declining calcification, higher concentrations of photo-pigments (affecting the energy and nutrient transfer between zooxanthellae and host; Marubini and Davies, 1996), and potentially higher rates of coral diseases (Bruno et al. 2003). Macroalgae and heterotrophic filter-feeders benefit more from dissolved inorganic and particulate organic nutrients than do corals. As a result, corals that can grow at extremely low food concentrations may be out-competed by macroalgae and/or more heterotrophic communities that grow best in high nutrient environments (Fabricius, 2011). Densities of benthic filter feeders – such as sponges, bryozoans, bivalves, barnacles and ascidians – increase in response to nutrient enrichment (Costa Jr et al. 2000). In high densities some filter feeders, such as internal macro-bioeroders, can substantially weaken the structure of coral reefs and increase their susceptibility to storm damage. Critically, more recent research shows that direct interactions between nutrients species such as nitrate and enhanced coral bleaching susceptibility will be important as a clear example of direct synergy between climate change stress and nutrient enrichment stress (Wooldridge 2009a; Wooldridge and Done 2009).

The impacts of nutrients on seagrass are less well known and there has been limited, detailed exploration of nutrient dynamics and nutrient limitation in the GBR, with notable exceptions (Udy et al. 1999; Mellors, 2003). Therefore, nutrients as an environmental driver has so far been difficult to elucidate because of other overriding factors such as light limitation, which tends to be a primary driver (Collier and Waycott 2009). Nutrient enrichment can stimulate seagrass growth (Udy and Dennison 1997; Udy et al. 1999) if other factors, such as light availability, are not limiting (Chapter 7 of the Supporting Studies, Collier, 2013). Although a theoretical nutrient toxicity level does exist, nutrient over-enrichment tends to impact at ecosystem scales and follow a path of eutrophication with excessive production of organic matter. In addition, nutrients favour the growth of plankton, macroalgae and epiphytic algae, all of which attenuate light to seagrass leaves (Collier, 2013). In the GBR some very high epiphyte loads occur on seagrass meadows of the GBR (Mckenzie et al. 2012) and are likely to reduce light reaching seagrass leaves. However, to date, these have largely been seasonal blooms, and epiphyte cover has not correlated well with seagrass abundance (Mckenzie et al. 2012). Although nutrient enrichment has been linked to high algal cover (Campbell et al. 2002), seagrass loss has rarely been attributed to nutrient over-enrichment. Further discussion of the impact of flood plumes and degraded water quality on seagrass ecosystems in the GBR is included in Chapter 6 of the Supporting Studies.

The impacts of elevated suspended solids to coral reefs and seagrass meadows were described above in Section 2.3.1a. It is important to note that particulate matter in plumes changes from 'clay' (or mineral)-based material in inshore regions, to organic matter (algal material) in offshore regions. These different types of particulate matter can have different effects on coral reefs and seagrass meadows as described in Chapter 3 of the Supporting Studies (Brodie et al. 2013a).

Further discussion of the impacts of TSS and DIN on GBR ecosystems is provided in SCS Chapter 1 (Schaffelke et al. 2013). Given the importance of flood plumes in delivering TSS and DIN to the GBR, the following variables related to plume loadings have been included in the assessment.

v DIN and TSS Plume Loading (mean 2007-2011)

Method:

Plume loading maps have been developed for TSS and DIN over the period 2007 to 2011. To date the method has not been developed for phosphorus due to limitations in our understanding of the relationship between end of catchment loads of phosphorus and marine concentrations due to complex dynamics in phosphorus processing in marine waters. However, in the assessment of the relative importance of nitrogen and phosphorus (Chapter 2 of the Supporting Studies, Furnas et al. 2013b) it was concluded that increased nitrogen inputs are more important than phosphorus inputs in driving ecological effects in the GBR.

The method for assessing plume loadings is described by Alvarez Romero et al. (2013), with further detail in Chapter 5 of the Supporting Studies (Devlin et al. 2013a). The method involves:

- i. Classification of GBR surface waters into colour classes corresponding to plume surface water, nonplume surface water, cloud and sun glint areas;
- ii. Creation of weekly plume colour class maps and mean annual color classes by overlaying maps created in *(i)*. Weekly composites maps minimize the amount of area without data per image due to masking of dense cloud cover, common during the wet season (Brodie et al. 2010), and intense sun glint;
- iii. Creation of maps of annual frequency of occurrence of plumes, by overlaying weekly composites created in *(ii)*. These maps help evaluating the annual frequency of occurrence of plumes, representing the number of weeks plumes are present (*i.e.* classified as plume surface water) in every pixel/areas of the GBR and during wet seasons;
- iv. Creation of spatially distributed TSS and DIN loading maps. This step involves: (a) the calculation of the percentage of the TSS and DIN load delivered by each of the 7 GBR priority rivers (Joo et al. 2012; Reef Plan 2003) in relation to the total TSS and DIN load from the catchments in Wet Tropics, Burdekin, Mackay Whitsundays and Fitzroy NRMs, and (b) the creation of grids representing the annual average distribution of TSS and DIN load delivered by the seven major rivers in the study region by multiplying their proportional contribution to the region-wide TSS and DIN loads with a cost-distance grid (see Alvarez-Romero et al. 2013) defining the maximum area of influence and the dispersal of pollutants in the sea. The individual spatially distributed grids (one per river) are then summed to represent the full TSS load per cell; the overlap of two or more grids defined cells influenced by multiple rivers;
- v. The production of annual risk maps for TSS and DIN loading within the GBR by multiplying the annual frequency of plume occurrence grid (*iii above*) by the grid representing the sum of spatially distributed TSS and DIN loads for all rivers (*iv above*). Exposure values are finally grouped in 5 categories of exposure (from very low to very high) to investigate spatial variation in exposure; and
- vi. Annual exposure maps were then reclassified into three categories of risk (high, medium, low see Table 2.1). For this assessment the mean of the 5 annual maps in the period 2007 to 2011 was selected as a way of factoring in some inter-annual variability, although it is recognised that this period was characterised by several extreme rainfall events.

The above assessment approach was not completed for Cape York because of outstanding issues with validation of true colour in this area. However, in recognition that there is some influence of river plumes in the Cape York region, a more crude estimate of TSS and DIN plume loading in this region has been applied which is summarised here and described in more detail in Chapter 5 of the Supporting Studies (Devlin et al. 2013a). This involved:

- a) Defining the primary and secondary plume types in the GBR for 2011 (see Alvarez-Romero et al. 2013; Devlin et al. 2013a).
- b) For TSS plume loading: Using the extent of the primary plume type in Cape York to define the extent of surface exposure. This is where coastal waters are characterised by elevated Colour Dissolved Organic Matter (CDOM) and TSS, with TSS concentrations dropping out rapidly as the heavier particulate material flocculates and settles to the sea floor. This area was then allocated as Low exposure as a conservative estimate given that there is limited information on the frequency of the occurrence of

these conditions, and low confidence in the remote sensing outputs in this Region. The remaining area in the Region was assessed as having no plume exposure.

c) For DIN plume loading: Using the extent of the secondary plume type in Cape York to define the extent of plume exposure. This is where intermediate waters are characterised by a region where CDOM is elevated, TSS concentrations are reduced due to sedimentation, and the increased availability of light and nutrient availability prompt phytoplankton growth measured by elevated Chl concentrations. This area was then allocated as Low exposure as a conservative estimate given that there is limited information on the frequency of the occurrence of these conditions, and low confidence in the remote sensing outputs in this Region. The remaining area in the Region was assessed as having no plume exposure.

Note that this assessment for Cape York is not included in the overall normalisation of loads for the model described in Step (iv) above, however, exposure scores for pollutants, particularly DIN, are typically low for the northern GBR and would not add significantly to the calculation of the normalised loads.

We also assessed the option of just using the plume loading maps from 2011 as a representation of a year of peak flood conditions, and for comparability to the pesticide exposure maps. However, it was concluded that the results were too biased to the influence from the large rivers of the Burdekin and Fitzroy which experienced flow conditions well above long term median records in 2011. Accordingly, the mean of the full set of available data (2007-2011) was used to represent variation in the characteristics of river discharges and hence flood plumes between the catchments between years.

Assessment classes:

The following three relative pollutant loading classes were allocated: Low, Medium and High. As an indication of the relative differences between these classes, preliminary satellite/in-situ match-up analyses have been performed to validate the 2007-2011 annual exposure maps and relate *in situ* data to the plume loading (exposure) classes and described in Chapter 5 of the Supporting Studies (Devlin et al. 2013a). There is a strong correlation between the in situ data and the assessment classes for DIN and TSS. For example, DIN concentrations increased linearly ($R^2 = 0.95$) with the highest concentrations corresponding to the highest DIN exposure category. This reflects the conservative mixing that has been described for DIN in the GBR lagoon (Devlin and Brodie, 2005; Devlin et al. 2012). TSS concentrations still increased along an increasing gradient of the level of exposure, with a rapid rise in concentrations in the highest exposure categories ($R^2 = 0.95$). This exponential increase reflects the sedimentation of the larger, heavier particles in the low salinity zones, and the transport of the finer sediment over much larger spatial scales. This sediment process is described for the Burdekin (Bainbridge et al. 2012), Tully (Devlin and Schaffelke, 2009) and reflects the higher risk associated with the availability of the finer sediment over longer time scales (Fabricius et al. 2012; Brodie et al. 2012b).

d) Pesticide concentrations

vi PSII Herbicide modelled concentration 2009-2011

Waters of the GBR lagoon are contaminated with a range of pesticides including herbicides, insecticides and fungicides. Pesticides, unlike nutrients, sediments and metals, have no natural sources and their concentrations have been positively correlated with low salinity associated with river runoff (Lewis et al. 2009; Kennedy et al. 2012a). Therefore, the occurrence of pesticides in the GBR can be attributed with great confidence to agriculture in the catchments that result in river discharge into the GBR lagoon. Of the 34 pesticides that have been detected in catchments draining to the GBR, several persistent and mobile PSII herbicides dominate the

pesticides identified in water samples and passive samplers in both near-shore and offshore sites on the GBR. Further information describing the relevance of PSII herbicides to this assessment is described in Chapter 4 of the Supporting Studies (Lewis et al. 2013a).

Multiple PSII herbicides are usually detected in water samples from the GBR (Lewis et al. 2012) and their combined effects on microalgae are additive (Shaw et al. 2009; Magnusson et al. 2010). This additive toxicity is not currently addressed in regulatory guidelines (King et al. in press; Lewis et al. 2012) and is considered to be important in this assessment. The reduced photosynthesis in algae due to herbicide exposure causes reductions in the growth of these algae (Magnusson et al. 2008) and changes in species composition (Magnusson et al. 2012) but the effects of chronic exposures in near-shore environments remain largely unknown. This assessment incorporates an assessment of the acute exposure of PSII herbicides in the 2009-11 wet seasons.

Method:

A full description of this method is provided in Chapter 4 of the Supporting Studies (Lewis et al. 2013a). A modelling approach based on the relationship between CDOM and sea surface salinity (Schroeder et al. 2012), was used with the results of in situ end of catchment and GBR lagoon pesticide concentration results for the 2009-2010 and 2010-2011 wet seasons.

Pesticide concentrations were assessed at the end-of-catchment monitoring sites in the 2009-2010 and 2010-2011 water years (Smith et al. 2012; Turner et al. 2012, 2013) to identify the periods where the higher concentrations coincided with elevated stream flows (based on the gauges of the Queensland Department of Natural Resources and Mines; QDNRM, 2012). Moderate Resolution Imaging Spectroradiometer (MODIS) Level-0 data with 1 km² resolution were acquired from the NASA Ocean Colour website (<u>http://oceancolor.gsfc.nasa.gov</u>). The most appropriate satellite image (i.e. the most free of cloud cover and sun glint) was selected for each NRM region within one week following the highest PS-II concentration. MODIS images were processed with the SeaWiFS Data Analysis System (SeaDAS). The semi-analytical model developed by Garvel-Siegel-Maritorena (GSM, Maritorena et al. 2002) implemented in SeaDAS was used to retrieve the absorption coefficient for dissolved and detrital material (CDOM+D). Bio-optical algorithms often fail to retrieve correct information over reef bottom type. Pixels values corresponding to reef locations were thus masked out from the CDOM regional maps. The Cape York region was excluded from this process due to the lack of monitoring data and the limited use of pesticides in this region.

CDOM was extracted from the satellite images and the relationship established by Schroeder et al. (2012) between measured salinity and CDOM was used to estimate sea surface salinity in the flood plumes. All of the regional pesticide maps were imported in ArcGIS for post-processing. Missing information (related to atmospheric perturbations, cloud cover or reefs that were masked out) was interpolated in ArcGIS. Pesticide levels were classified into different level of risk and the areas of reef and seagrass meadows at risk for each NRM region were quantified.

Two different but complimentary methods were used to determine the risk posed by mixtures of PSII herbicides. These were the Toxic Equivalence Quotient (TEQ) method (eg. Kennedy et al. 2012a; Smith et al. 2012) and the multiple substances potentially affected fraction (ms-PAF) method (Traas et al, 2002). Importantly both methods use the concentration addition model to determine the toxicity of mixtures of PSII herbicides. The maps shown in this assessment are from the TEQ method.

Assessment classes:

The key PSII herbicides of concern (diuron, hexazinone, atrazine, tebuthiuron, ametryn and simazine) were normalised to an herbicide-equivalent concentration which is based on the relative toxicity of diuron; the risk

posed by PSII herbicides collectively could then be examined using the concentration addition model for joint toxicity (see Kennedy et al. 2012a). The relative toxicities (EC50s and EC25s) of marine organisms including coral species (*Seriatopora hystrix* and *Acropora formosa*), diatoms (*Phaeodactylum tricornutum*) and green algae (*Chlorella vulgaris*) (Jones and Kerswell, 2003; Bengtson Nash et al. 2005; Muller et al. 2008) to each PSII herbicide compared to diuron was determined and then averaged to produce the relative toxicity factors (RTFs) (Kennedy et al. 2012a). The TEQ method was applied to the measured EC50s and EC25s of PSII herbicides that inhibit the effective quantum yield (YII) in plants. Inhibition in YII by PSII herbicides is proportional to inhibition of photosynthesis and growth in tropical microalgae (Magnusson et al. 2008) as well as reduced energy acquisition by the host coral from its photosynthetic symbionts (Cantin et al. 2009).

Based on the toxicity of diuron calculated in several studies on coral and seagrass species we devised a set of threshold values that were considered to match the following risk classifications:

- Very High: >10 µg/L causes reduced growth and mortality in seagrass (Gao et al. 2011) and loss of symbionts (bleaching) in corals (Jones et al. 2003; Negri et al. 2005). The effect on health and survival of foundation species of the GBR can be catastrophic.
- High: 2.3 10 μg/L Photosynthesis is reduced by between 50% and 90% in corals (Jones and Kerswell, 2003; Negri et al. 2011); seagrass (Chesworth et al. 2004; Gao et al. 2011; Flores et al. in review) and microalgae (Magnusson et al. 2008, 2010). A 50% reduction of growth and biomass of tropical microalgae was also reported in this concentration range (Magnusson et al. 2008). The community structure of tropical microalgae is significantly affected and this causes significant changes in the tolerance of microbial communities to herbicides (Magnusson et al. 2012). The effect on primary production is major.
- Medium: 0.5-2.3 μg/L Photosynthesis is reduced by between 10% and 50% in corals (Negri et al. 2011); seagrass (Haynes et al. 2000; Chesworth et al. 2004; Gao et al. 2011; Flores et al. in review) and microalgae (Magnusson et al. 2008, 2010). The community structure of tropical microalgae can be affected by concentrations of diuron as low as 1.6 μg/L (Magnusson et al. 2012). The effect on primary production is moderate.
- Low: 0.1-0.5 μg/L Photosynthesis is reduced by up to 10% in corals (Negri et al. 2011); seagrass (Haynes et al. 2000; Chesworth et al. 2004; Gao et al. 2011; Flores et al. in review) and microalgae (Magnusson et al. 2008, 2010). The effect on primary production is minor.
- Very Low: 0.025-0.1 μg/L No observed effect on photosynthesis in corals (Negri et al. 2011); seagrass (Haynes et al. 2000; Flores et al. in review) and microalgae (Magnusson et al. 2008, 2010).
- No Risk: < 0.025 μg/L

The highest risk classification determined for any point in a flood plume from a catchment was adopted as the risk posed by that catchment.

Further explanations of the methods are provided in Chapter 4 of the Supporting Studies (Lewis et al. 2013a).

e) Recognising and assessing uncertainties in the selection of variables

For all variables, any relative differences in uncertainty and hence our confidence in the data can only be assessed highly subjectively. If such qualitative assessments of uncertainty in our methodologies and data were undertaken, uncertainty would be assessed as varying as much within as among NRM regions. As we compare

results for NRM regions in the final combined relative risk assessment, the various methodologies used to generate the data are considered to have roughly the same uncertainty and with the limited time and resources, no specific estimates were considered.

Further discussion of the uncertainties and limitations of the assessment are presented in Part B of this report, Section 4.

2.3.2 Estimating habitat area

The habitats considered in the assessment were for coral reefs and seagrass meadows, based on the best available information. The area of GBR lagoon waters in each NRM region was also included in the assessments as it contains other important habitats and biological populations such as phytoplankton, fish and benthic organisms. The area estimates in each Region are based on the GBRMPA Spatial Data Centre's coral reefs spatial data file (December 2012) and seagrass spatial data files supplied by TropWATER James Cook University. The seagrass habitat map used is comprised of a composite of the survey data up to 2010 (observed habitat) and a statistical model of seagrass present in GBRWHA waters >15 metres depth. In this model spatial distribution is a statistically modeled probability of seagrass presence (using generalised additive models with binomial error and smoothed terms in relative distance across and along the GBR), based on ground truthed points (Coles et al. 2009). Locations with seagrass habitat probability >0.5 were included in the assessment.

2.3.3 Assessment Method - Part 1: Differential risk between pollutants on GBR ecosystems

The relative risk of different pollutants between regions was estimated by calculating the area of coral reefs, seagrass meadows and GBR lagoon waters associated with an NRM region within each of the assessment classes for each variable. As the first step, the assessment classes for each variable were allocated score in MCAS-S between 0 (lowest severity) and 1 (highest severity) at the 1 km² pixel scale. These are shown in Table 2.3 below. Pixels in the highest assessment class all received the maximum value of 1. For example, for the TSS threshold of 2 mg/L the scores for the frequency of exceedance classes would be Very Low (<1% exceedance) = 0; Low (1-10% exceedance) = 0.25; Medium (10-20% exceedance) = 0.5; High (20-50% exceedance) = 0.75; and Very High (50-100% exceedance) = 1.0. The areas of coral reefs, seagrass meadows and GBR lagoon waters were then reported for each assessment class in each region in ArcGIS.

To compare results between regions, only the areas affected in the highest assessment class for each variable was considered as these were determined through expert opinion to be the most ecologically relevant in determining risk, except for PSII herbicides where the highest and second highest assessment classes were included in recognition of the toxicity of both of these classes. The output is a map and a table showing the area (km²) of coral reef, seagrass and GBR lagoon waters within each assessment class for all variables in all NRM regions. The assessment classes used in this part of the analysis are highlighted in grey in Table 2.3. For example, the TSS threshold of 2 mg/L used in the assessment is the highest frequency of exceedance class of 50 to 100% (Very High).

The results were then anchored for comparison; the maximum area is set as an anchor point and given a value of 100, and all other area calculations are then expressed as a proportion of the maximum (values between 0 and 100). For example, for the TSS threshold of 2 mg/L the area of coral reef within the highest assessment class (Very High – 50-100% exceedance of the threshold) is 9 km² in the Burdekin, 6 km² in the Fitzroy, and all other regions have less than 2 km² of coral reef in the area of the Very High assessment class. In this case the maximum area of 9 km² is in the Burdekin, so the Burdekin is allocated a value of 100. The other areas are then reported as a proportion of the maximum value of 9 km²; the Fitzroy is therefore 61% and all other regions are

less than 15%. It can then be concluded that the area of coral reefs in the Fitzroy region within the Very High assessment class for TSS threshold of 2 mg/L is 61% of that in the Burdekin region.

Table 2.3. Summary of the classes used to assess the differential risk between pollutants on GBR ecosystems (Part 1), and the weightings given to each assessment class for the MCAS-S combined assessment (Part 2). The cells shaded in grey show the classes included in assessing the relative risk between variables (Part 1). The overall weighting for the water quality variables used in the combined assessment are also shown. The variables are described in Table 2.1.

Variables	Overall	Assessment Class					
	weighting	Very Low	Low	Medium	High	Very High	
		1	2	3	4	5	
TSS threshold exceedance 2mg/L							
Frequency of exceedance (%)		<1	1-10	10-20	20-50	50-100	
MCAS-S ¹ score	1/7	0	0.25	0.5	0.75	1.0	
TSS threshold exceedance 7 mg/L						20-50	
(5NTU)		0	<1	1-10	10-20	50-100	
Frequency of exceedance (%)							
MCAS-S score	1/7	0	0	0.33	0.66	1.0	
TSS Plume Loading		Categ	gory 1	Category 2	Category 3		
(mean 2007-2011)							
MCAS-S score	1/7	0.3	33 ²	0.66	1.0 ³		
Chl threshold exceedance							
(0.45µg/L)		<1	1-10	10-20	20-50	50-100	
Frequency of exceedance (%)							
MCAS-S score	1/7	0	0.25	0.5	0.75	1.0	
DIN Plume Loading		Category 1		Category 2	Category 3		
(mean 2007-2011)							
MCAS-S score	1/7	0.3	33 ²	0.66	1.0 ³		
COTS Initiation Zone		Outside				Within Zone	
		Zone					
MCAS-S score	1/7	0				1.0	
PSII Herbicide modelled							
concentration		0.025-0.1	0.1-0.5	0.5-2.3	2.3-10	>10	
(2009-2011) (μg/L)							
MCAS-S score	1/7	0.25	0.5	0.75	1.0	No occurrence	

¹ MCAS-S: Multi-Criteria Analysis Shell for Spatial Decision Support developed by ABARES, refer to <u>http://www.daff.gov.au/abares/data/mcass</u>; ² This class covers Very Low and Low; ³ This class covers High and Very High.

2.3.4 Assessment Method - Part 2: Combined risk of degraded water quality to GBR ecosystems

To consider the combined risk of the selected water quality variables to marine ecosystems, the spatial layers for the individual variables described above were combined and the scores allocated to each assessment class for each variable were summed at the 1 km² pixel scale (see Figure 2.7). Anchoring and normalising the data to the uni-directional scale ensures that the ranges in data values (very different among variables) are standard and can be summed to produce a single score for each pixel. Pixels with no data were not included in the final averaging. It is recognised that this approach to standardising the assessment classes has limitations when considering the equivalency of the assessment classes for each variable.



Figure 2.7. Example of the results in one pixel from a Composite in MCAS-S. The result for the pixel from each layer is summed to give a combined score. These scores are then classified into five assessment classes (Very Low to Very High). In this example the combined score gives the pixel a score within the High assessment class in terms of relative risk of degraded water quality.

The classifications, scores and overall weightings for this assessment were based on expert opinion and are shown in Table 2.3. Ideally the classes for each variable would be scaled so that they are equivalent in terms of potential ecological impacts to provide comparable weightings between variables. However, it is recognised that this may not be the case for all variables given the inconsistencies in the temporal and spatial characteristics of the datasets. As temporal and spatial resolution of data increases and the knowledge of the impacts of sediments, nutrients and PSII herbicides on GBR ecosystems is advanced, this capability can be improved in future assessments.

The data layers were then combined using the *Composite tool* in MCAS-S which essentially sums the results of the scores for each 1 km² pixel in each spatial layer to create a combined spatial layer. Figure 2.8 illustrates the use of the MCAS-S tool. The result of the summed pixels can then be normalised in MCAS-S to generate a score between 0 and 1 for each pixel (see example in Figure 2.9). The results for each pixel were further classified into five even break classes ranging from Very Low to Very High to provide a classification of relative risk of degraded water quality. A more detailed classification of 10 classes was also tested but the spread of the data resulted in

limited benefit from using more than five classes. An example of the process applied in MCAS-S is shown in Figure 2.7.

The area of coral reefs, seagrass meadows and GBR lagoon waters within each of the five assessment classes was calculated in ArcGIS and tabulated for comparison between regions. A **Marine Risk Index** was defined by summing the areas of coral reefs, seagrass and GBR lagoon waters in the Very High and High assessment classes of the combined layer, and anchoring those results to the maximum area among regions. This enabled an assessment of the relative differences between regions in terms of combined water quality risk for coral reefs, seagrass and GBR lagoon waters. The Very High and High assessment classes were determined (by expert opinion) to be the most ecologically relevant for the assessment. It is recognised that these classes are relative and that the areas of coral reefs and seagrass meadows in the lower assessment classes may also be important depending on temporal and spatial variability of the exposure to exceedance in the water quality variables. A more detailed assessment of these patterns in the lower assessment classes was outside of the scope of this project but should be considered in future work, particularly given the potential influence of chronic exposure to pollutants, or the effects of periodic exposure to high concentrations of pollutants.

The final output is a Coral Reef Marine Risk Index and a Seagrass Marine Risk Index.

A number of options for combining the variables were considered which essentially weighted each variable differently, described in Appendix 2. All of these were performed using the *Composite tool* in MCAS-S described above. The selected option involved combining all variables individually with an equal weighting (1/7) and summed. The maximum score for each pixel is therefore 7 (Figure 2.8), and this value is then normalised between 0 and 1 (Figure 2.9). This approach essentially weights the variables differently because there are 3 sediment related variables, 3 nutrient related variables and 1 pesticide related variable which were considered to be appropriate by the assessment team given current evidence of the relative importance of nutrients and sediments compared to pesticides in the GBR. An example of this combination in MCAS-S is shown in Figure 2.7.



Figure 2.8. Example of the results in one pixel (near the arrow) from a Composite in MCAS-S where all individual variables are combined equally (formula shown in the panel to the left). The score for each variable is shown in the Viewer: the upper box shows the combined score for each variable where variables with no result or zero are allocated a value of -9999 (and not counted), the lower box identifies the input scores for each variable in the selected pixel (in this example the COTS Initiation Zone did not receive a score as the pixel is outside the Zone). The scores can then be normalised as shown in Figure 2.9.

Figure 2.9. Example of the results in one pixel (near the arrow) from a normalised Composite in MCAS-S where all individual variables are combined equally (shown in the panel to the left). The classes and distribution of pixels within these classes for the combined layer is shown at the bottom of the left panel. This combined class for each variable is shown in the Viewer: the upper box shows the combined score for the selected pixel, the lower box identifies the scores for each input variable in the selected pixel.

2.3.5 Assessment Method - Part 3: Relative risk of degraded water quality to GBR ecosystems

To inform management priorities that aim to address the risks identified in Part 1 and 2, it is necessary to understand the influence of river discharge in each of the regions, as these discharges carry a majority of the pollutants into the GBR lagoon.

To relate the results of the Marine Risk Index (described in Section 2.3.4) to land based influences, anthropogenic end-of-catchment loads were expressed as the proportion of total GBR load for each Region to generate a **Loads Index** for each region. This recognises that while the total GBR load is important in influencing the marine water quality conditions, it is only the anthropogenic proportion that can be factored into management. The regional proportional contributions were then anchored (to normalise to a standard scale) and averaged to generate a Loads Index for TSS, DIN and PSII herbicides for each NRM region. This assumes that the relative importance of each load is equal which may not be the case, although there is currently insufficient knowledge to weight the importance of the three pollutants relative to each other.

End-of-catchment anthropogenic loads were obtained from the results of the Source Catchments model framework which have been produced as part of the Paddock to Reef Program (Waters et al. in press). First, the Source Catchments modelling framework was used as a synthesis tool that incorporates new information on paddock modelling of TSS, speciated N and P, and PSII herbicides, plus spatially and temporally remote sensed inputs. This resulted in a consistent set of end of catchment pollutant loads for each of the 35 GBR catchments. Anthropogenic load is calculated as the difference between the long term average annual load and the estimated pre-European annual loads. A fixed climate period was used (1986 to 2009) for all model runs to normalise for climate variability and provide a consistent representation of pre-development and anthropogenic generated catchment loads. This therefore represents an 'average' year rather than the extremes such as those recorded in the period 2008 to the current wet season in 2013. In addition, functionality from the previous iteration catchment modelling, SedNet/ANNEX (for example see Cogle et al. 2006), was incorporated into Source Catchments to represent hillslope, gully and streambank erosion and floodplain deposition processes.

It is recognised that assessment of the input of pesticides from each region can be expressed in a number of ways, and while loads allow comparison between regions, it is the toxicity and therefore concentration that is most relevant to the receiving environment. However, pesticide concentration data is currently limited across the GBR. Therefore, in the final conclusions relating to pesticide risk in this assessment, additional evidence is drawn from a combination of load and concentration data from specific locations, assessed in Lewis et al. (2013a).

An index specific to the potential influence of river DIN loads on the initiation of COTS primary outbreaks in the GBR was added (see Furnas et al. 2013a). DIN runoff is considered to be an important factor as approximately 40% of the loss of coral cover in the GBR since 1987 has been attributed to COTS predation (De'ath et al. 2012). A COTS outbreak initiation zone has been defined between Lizard Island (14.5°S) and Cairns (17°S). On total volumetric basis, most (86%) of the estimated freshwater input (direct and indirect) to the Zone comes from Wet Tropics rivers, with the remaining 14% from the Burdekin River (Furnas et al. 2013a). These estimates were used to create a **COTS Influence Index**.

To provide an overall relative ecological risk ranking between the NRM regions, the Marine Risk Indexes for coral reefs and seagrass meadows were summed with the Loads Index, and for coral reefs only, the COTS Influence Index, to generate a Coral Reef Relative Risk Index and a Seagrass Relative Risk Index. These final indexes for coral reefs and seagrass were then summed and normalised to give an overall assessment of the relative risk of degraded water quality to coral reefs and seagrass meadows to generate a **Relative Risk Index** for each region.

Once the relative risks and pollutants are known, the load information for each catchment allows us to track back to management priorities. A detailed assessment of the load contributions at a catchment scale was outside of the scope of this assessment, however, the SCS chapters on sources of pollutants (Kroon et al. 2013) and management practice effectiveness (Thorburn et al. 2013) provide a solid foundation for this analysis to be progressed.

2.4 Results

2.4.1 Habitat and regional NRM areas

The distribution of coral reefs and seagrass used in the risk assessment are shown in Figure 2.10, and the inset table shows the area of coral reef, seagrass and area of GBR lagoon waters in each marine NRM region. The total area of the Great Barrier Reef Marine Park in this spatial dataset is 345,804 km²; this represents a difference of well less than half of 1% the published area assessment from the GBRMPA of 345,400km². The very slight difference between the area estimates used here from ArcGIS and the published area assessments are a result of slightly different estimates of the areas around coastlines and the Marine Park boundary. We consider the difference here to be negligible.

The total area of coral reef in the GBR is estimated around 24,000 km². The Regions with the largest areas of coral reef are Cape York (10,354 km²), Fitzroy (4,855 km²), and Mackay Whitsunday (3,213 km²). Approximately 35,000km² of seagrass has been mapped in the coastal waters around Queensland and Torres Strait since the mid 1980s. Surveys and statistical modelling of seagrass in offshore waters deeper than 15m shows 37,454 square kilometres of the sea floor within the Great Barrier Reef World Heritage Area and Torres Strait has some seagrass present making Queensland's seagrass resources globally significant.

From the mapping data used in this assessment, the Cape York marine NRM region also has the highest area of seagrass with 11,378km². The regions with the second and third-highest seagrass area are the Burnett Mary (6,330 km²) and the Burdekin (6,083 km²), respectively. The area of seagrass in the Mackay Whitsunday region is relatively low compared to other Regions with only approximately 430 km². Deepwater seagrasses are sparse in the Mackay Whitsunday region, particularly south of Mackay, where tidal velocities are high and no major deepwater seagrass meadows exist (Coles et al. 2009). High current stress, low Secchi readings and coarse mobile sediments generally make this an unsuitable habitat for seagrass growth.

It is important to note that the habitats of the Burnett Mary region are under estimated in this assessment, as the GBR Marine Park and World Heritage Area boundary does not include all of the habitat areas that would be affected by the catchments of the Burnett Mary region. In particular, there is a large area of seagrass to the south of the boundary in Hervey Bay which is known to provide important habitat, and foraging grounds, for species that also inhabit the GBR Marine Park. The total area estimate for seagrass in the Burnett Mary region is around 8,000km² (McKenzie et al. 2010b). These limitations are discussed further in Part B Section 4.

2.4.2 Part 1: Relative importance of different pollutants to GBR ecosystems

The following section presents the results of the individual variables considered in this assessment (refer to Section 2.3.3 for a description of the methods). This part of the risk assessment identifies the areas where each water quality variable is considered to pose the greatest relative risk to coral reefs and seagrass between the NRM regions. The output can be used to guide priorities for management of individual pollutants between NRM regions.



Figure 2.10. Locations of coral reefs and seagrass meadows used for the risk assessment. Coral reef outlines used are per the GBRMPA Spatial Data Centre official reefs spatial data layer 2013. Seagrass areas are observed (composite of surveyed data as at June 2010) and modelled deepwater seagrass habitat after Coles et al. (2009). Inset table shows the area of coral reef, seagrass, those habitats combined and region for each marine NRM region.

The area of coral reefs, seagrass and GBR lagoon waters is shown for all assessment classes for all variables in this section. Area calculations are rounded to the nearest whole km² for ease of reporting but does include summed portions of some 1 km² pixels. For each variable there is a table of results, and a map shows the areas within each assessment class for each variable. Within the map, a graph compares the areas of coral reef and seagrass affected by the highest class and a comparison of these areas with the total (whole of GBR) areas of habitat affected among the NRM regions is presented as a pie chart.

Note that in this section the regions are simply referred to as their name, for example 'the Burdekin' rather than 'the Burdekin region' to improve readability.

a) Sediments

Total suspended solids threshold exceedance, Threshold a – 2 mg/L

As shown in Table 2.1, five assessment classes were used for TSS 2 mg/L based on the frequency of exceedance of this concentration (in days) in the period 2002 to 2012, expressed as a percentage of the total number of valid daily observations ranging from Very Low to Very High. However, as the Very Low class receives a score of 0, it is not reported here. The results of the assessment are shown in Table 2.4 and Figure 2.11. The areas within the Very High class are constrained to the coast and concentrated in the Burdekin and Fitzroy. These inshore areas are locations with some of the highest use and visitation rates; this is a result common to all individual variables and is reviewed in the discussion.

Key findings:

- The area of coral reef in the Very High class of TSS exceedance at 2 mg/L is greatest in the Burdekin (9 km²), and second-greatest in the Fitzroy (6 km²) (see inset in Figure 2.11). The greatest area within the High class is in the Fitzroy (103 km²) and Mackay Whitsunday is second (90 km²). The proportion of coral reefs in each region in the Very High class is less than 1%. More than 95% of the coral reefs within in each region are within the lowest classes (Very Low and Low) for exceedance of TSS 2 mg/L.
- The area of seagrass within in the Very High class of TSS exceedance at 2 mg/L is also greatest in the Burdekin (209 km²), which is ~10 times greater than the second-greatest area in the Wet Tropics (22 km²) (see inset Figure 2.11). In the High class Cape York has the greatest area of seagrass (747 km²), with Fitzroy second (300 km²). The proportion of seagrass meadows in each region in the Very High class is less than 5%. More than 59% of the seagrass meadows within in each region are within the combined Very Low and Low class for exceedance of TSS 2 mg/L and a large proportion of these are deepwater meadows (>15m). The proportion of seagrass within all other categories is similar across regions (all less than 10%) with the notable exception of Mackay Whitsunday which has 18% and 16% within the Medium and High classes respectively.
- The pie chart insets in Figure 2.11 show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the Very High class of TSS exceedance at 2 mg/L. The area of coral reef in the Very High class in Burdekin represents 52% of the total GBR coral reef area within the Very High class, and the area of seagrass in the Very High class in the Burdekin represents 78% of the total GBR seagrass area within the Very High class. The lowest areas of coral reefs and seagrass in both the Very High and High classes are in the Burnett Mary (<5 km² for coral and 55km² for seagrass).
- The area of GBR lagoon waters in the Very High class of TSS exceedance at 2 mg/L is greatest in the Burdekin (932 km²), and second in the Fitzroy (502 km²). The greatest area in the High class is in the

Fitzroy (3,390 km²) with Mackay Whitsunday second (2,122 km²). The proportion of GBR lagoon waters in each region in the Very High class is less than 5%.

• When summed, the area of coral reef within the High and Very High classes (frequency of exceedance 20-50% and 50-100%) of TSS exceedance at 2 mg/L represents 0.8% (Cape York) to 2.8% (Mackay Whitsunday) of the total coral reef area in the regions; seagrass ranges from 0.9% (Burnett-Mary) to 17% (Mackay Whitsunday) of the seagrass area in the regions, and GBR lagoon waters from 1% (Burnett-Mary) to 5% (Burdekin) of the total region area. The lowest total areas in the Very High and High classes are in Burnett Mary, however the assessment area is bound to the GBRWHA and therefore does not incorporate coral reefs that have not been fully mapped within the region, or the large seagrass meadows in Hervey Bay the large seagrass meadows in Hervey Bay.

TSS 2 mg/L	Low		Medium		High		Very High	
Frequency of exceedance class	1-10%		10-20%		20-50%		50-100%	
NRM Regions	Area (km²)	% of habitat in region						
Coral Reefs								
Cape York	10,163	98	71	1	84	1	0	0
Wet Tropics	2,355	97	17	1	46	2	0	0
Burdekin	2,930	99	2	<1	9	<1	9	<1
Mackay Whitsunday	2,886	90	191	6	90	3	1	<1
Fitzroy	4,559	94	173	4	103	2	6	<1
Burnett-Mary	277	98	1	<1	3	1	2	1
Seagrass								
Cape York	8,949	79	463	4	747	7	0	0
Wet Tropics	4,450	91	57	1	117	2	22	<1
Burdekin	4,289	71	54	1	165	3	209	3
Mackay Whitsunday	255	59	79	18	70	16	4	1
Fitzroy	5,042	87	159	3	300	5	11	<1
Burnett-Mary	5,926	94	24	<1	35	1	20	<1
GBR lagoon waters								
Cape York	92,137	96	2,016	2	1,809	2	0	0
Wet Tropics	28,647	91	1,238	4	1,446	5	147	<1
Burdekin	43,582	93	734	2	1,497	3	932	2
Mackay Whitsunday	43,948	91	2,274	5	2,122	4	131	<1
Fitzroy	79,095	92	2,440	3	3,390	4	502	1
Burnett-Mary	36,514	98	412	1	233	1	145	<1

Table 2.4. Area of coral reefs, seagrass meadows and GBR lagoon waters within the Low to Very High assessment classes for TSS 2 mg/L and the percent of the NRM region that the area represents. Results for the assessment are based on frequency of exceedance of TSS 2 mg/L using daily remote sensing data 2002-2012 (see methods in Section 2.3.1a(i)).

The results presented here and used in the risk assessment have been compared with spatial patterns in secchi depth shown in De'ath and Fabricius (2008). The visual comparison of our outputs for TSS exceedance at 2 mg/L and those from De'ath and Fabricius (2008) are shown in Appendix 1 along with a similar comparison for Chl 0.45 μ g/L. The spatial patterns are very similar suggesting the remotely sensed data used here are at least comparable to the long term patterns measured in situ.



Figure 2.11. Results for the assessment of frequency of exceedance of TSS 2 mg/L using daily remote sensing data 2002-2012. Results for the assessment are based on frequency of exceedance of TSS 2 mg/L (see methods in Section 2.3.1a(i)). Inset bar chart compares coral reef and seagrass area within the Very High class (50-100% exceedance); pie charts show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the Very High class.

Total suspended solids threshold exceedance, Threshold b - 7 mg/L (turbidity 5NTU)

As shown in Table 2.1, five assessment classes were used for TSS 7 mg/L (5NTU) based on the frequency of exceedance of this concentration (in days) in the period 2002 to 2012, expressed as a percentage of the total number of valid daily observations ranging from Very Low to Very High, but only the results of the Medium to Very High classes are presented as the most relevant here. Note that the assessment classes are different from those for TSS 2 mg/L to reflect the greater severity of the higher concentration; however, there were no pixels where the frequency of exceedance was greater than 50%. The results of the assessment are shown in Table 2.5 and Figure 2.12. The area exposed in the Very High class is constrained to the coast (see map forming Figure 2.12) and concentrated in the Fitzroy region.

Table 2.5. Area of coral reefs, seagrass meadows and GBR lagoon waters within the Medium to Very High assessment classes for TSS 7 mg/L (5 NTU) and the percent of the NRM region that the area represents. Results for the assessment are based on frequency of exceedance of TSS 7 mg/L (5 NTU) using daily remote sensing data 2002-2012 (see methods in Section 2.3.1a(ii)).

TSS 7 mg/L	Medium		ŀ	ligh	Very High	
Frequency of exceedance class	1-	10%	10-20%		20-100%	
NRM Regions	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region
Coral Reefs						
Cape York	1,280	12	3	<1	0	0
Wet Tropics	423	17	0	0	0	0
Burdekin	105	4	<1	<1	<1	<1
Mackay Whitsunday	747	23	5	<1	0	0
Fitzroy	775	16	25	1	11	<1
Burnett Mary	58	20	0	0	0	0
Seagrass						
Cape York	2,437	21	30	<1	0	0
Wet Tropics	624	13	50	1	9	<1
Burdekin	600	10	96	2	35	1
Mackay Whitsunday	219	51	16	4	0	0
Fitzroy	484	8	29	<1	9	<1
Burnett Mary	393	6	6	<1	0	0
GBR lagoon waters						
Cape York	9,661	10	80	<1	0	0
Wet Tropics	5,063	16	204	1	30	<1
Burdekin	7,381	16	451	1	170	<1
Mackay Whitsunday	6,026	12	185	<1	17	<1
Fitzroy	9,329	11	1,094	1	504	1
Burnett Mary	2,062	6	26	<1	0	0

Key findings:

• The area of coral reef in the Very High class of TSS exceedance at 7 mg/L is greatest in the Fitzroy (11 km²); all other regions have less than 1 km² or zero (see inset in Figure 2.8). The greatest area within the High class is in the Fitzroy (25 km²) and Mackay Whitsunday is second (5 km²). The proportion of coral

reefs in each region in the High and Very High class is less than 1%. More than 80% of the coral reefs within each region are within the lowest classes where the frequency of exceedance of TSS 7 mg/L was <1% (not shown here). The lowest areas of coral reefs in all assessment classes are in Burnett Mary.

- The area of seagrass within in the Very High class of TSS exceedance at 7 mg/L is also greatest in the Burdekin (35 km²), with 9 km² within in both the Wet Tropics and Fitzroy (see inset Figure 2.12). In the High class Burdekin has the greatest area of seagrass (96 km²), with Wet Tropics second (50 km²). The proportion of seagrass meadows in each region in the High class is less than 5%, and less than 1% in the Very High class. More than 80% of the seagrass meadows within all regions except Mackay Whitsunday are within the lowest classes where the frequency of exceedance of TSS 7 mg/L was <1% (not shown here). In Mackay Whitsunday 55% of the seagrass is within the Medium to Very High classes.
- The area of GBR lagoon waters in the Very High class of TSS exceedance at 7 mg/L is greatest in the Fitzroy (504 km²), and second in the Burdekin (170 km²). The greatest area in the High class is in the Fitzroy (1,094 km²) with Mackay Burdekin second (451 km²). The proportion of GBR lagoon waters in each region in the Very High class is less than 1%. More than 80% of the GBR lagoon waters within in each region are within the lowest classes where the frequency of exceedance of TSS 7 mg/L was <1% (not shown here).
- The pie chart insets in Figure 2.12 show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the Very High class of TSS exceedance at 7 mg/. The area of coral reef in the Very High class in Fitzroy represents 97% of the total GBR coral reef area within the Very High class, and the area of seagrass in the Very High class in the Burdekin represents 65% of the total GBR seagrass area within the Very High class.
- When summed the area of coral reef within the High and Very High of TSS exceedance at 7 mg/L represents 0% (Wet Tropics) to 0.7% (Fitzroy) of the total coral reef area in the regions; seagrass ranges from 0.1% (Burnett Mary) to 3.7% (Mackay Whitsunday) of the total seagrass area in the regions, and GBR lagoon waters from 0.1% (Burnett Mary) to 1.9% (Fitzroy) of the total region area. The total areas in the High and Very High classes are lowest in Burnett Mary; however the assessment area is bound to the GBRWHA and therefore does not incorporate coral reefs that have not been fully mapped within the region, or the large seagrass meadows in Hervey Bay.

For both concentrations of TSS exceedance, there is a distinct area of exceedance of the Medium class in the coastal areas in the Cape York region north of Cooktown and Princess Charlotte Bay. Further validation of this result is required, however recent studies by Brooks et al. (2013) indicate that suspended sediment loads from the Normanby River are likely influence this area. Similar patterns exist in the coastal areas around Shoalwater Bay in the northern part of the Fitzroy region which are also known to be naturally turbid and uncertainties in the remote sensing results in these areas have not been resolved. While these uncertainties may be important for regionally specific analyses, they are not considered to be significant enough to influence the overall conclusions of this assessment.



Figure 2.12. Results for the assessment of frequency of exceedance of TSS 7 mg/L (5NTU) using daily remote sensing data 2002-2012. Results for the assessment are based on frequency of exceedance of TSS 7 mg/L (5NTU) (see methods in Section 2.3.1a(i)). Inset bar chart compares coral reef and seagrass area within the Very High class (50-100% exceedance); pie charts show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the Very High class.

TSS plume loading (mean 2007-2011)

As shown in Table 2.1, three assessment classes were used for TSS plume loading based on plume frequency information from remote sensing and scaled river load data. To factor in inter-annual variability, we used the mean result of assessments completed annually from 2007 to 2011. The results of the assessment are shown in Table 2.6 and Figure 2.13. Note that there are areas within the GBRWHA that are assessed as being outside of the area of plume loading influence which are not reported in this assessment. The areas within the highest assessment class (High) are inshore (Figure 2.13) and mostly concentrated in the Burdekin region.

Table 2.6. Area of coral reefs, seagrass and GBR lagoon waters within the assessment classes for TSS mean plume loadings between 2007 and 2011, and the percent of the NRM region that the area represents. The assessment classes are relative and derived from a combination of scaled river loads data and flood plume frequency analysis from remote sensing data (see methods in Section 2.3.1b).

TSS plume loadings	L	.0W	Me	edium	High	
NRM Regions	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region
Coral Reefs						
Cape York	8,903	86	0	0	0	0
Wet Tropics	2,267	93	23	1	2	<1
Burdekin	2,746	93	35	1	20	1
Mackay Whitsunday	401	12	229	7	1	0
Fitzroy	694	14	161	3	6	0
Burnett Mary	268	95	3	1	0	0
Seagrass						
Cape York	11,433	100	0	0	0	0
Wet Tropics	4,535	93	223	5	48	1
Burdekin	5,113	84	354	6	610	10
Mackay Whitsunday	<1	<1	407	95	4	1
Fitzroy	5,244	91	490	8	30	1
Burnett Mary	5,750	91	341	5	0	0
GBR lagoon waters						
Cape York	54,536	57	0	0	0	0
Wet Tropics	21,464	68	2,817	9	445	1
Burdekin	21,899	47	7,160	15	4,553	10
Mackay Whitsunday	13,004	27	12,331	25	136	0
Fitzroy	26,488	31	13,506	16	763	1
Burnett Mary	10,172	27	1,802	5	0	0

Key findings:

• The area of coral reef in the High class for TSS plume loading is greatest in the Burdekin (20 km²), and second-greatest in the Fitzroy (6 km²) (see inset in Figure 2.13). The greatest area within the Medium class is in Mackay Whitsunday (229 km²) and Fitzroy is second (161 km²). The proportion of coral reefs in each region in the High class is less than 1%. More than 85% of the coral reef area in the Cape York, Wet Tropics and Burdekin are within the Low class, and approximately 80% of the area of coral reefs in the Mackay Whitsunday and Fitzroy are outside the mapped plume loading area. There is no occurrence of coral reefs in the High class in Cape York and Burnett Mary, but there are limitations to the method

applied to for Cape York described in Section 2.3.1c, and the area of coral reefs in the Burnett Mary is not considered to be accurate.

- The area of seagrass within the High class for TSS plume loading is also greatest in the Burdekin (610 km²), which is considerably larger than the second-greatest area in the Wet Tropics (48 km²) (see inset Figure 2.13). In the Medium class Fitzroy has the greatest area of seagrass (490 km²) with Mackay Whitsunday second (407 km²). The proportion of seagrass meadows in each region within the High class is less than 1%, except for Burdekin which is 10%. Approximately 95% of the seagrass area in Mackay Whitsunday region is in the Medium class, and all other regions are less than 10%, which is associated with the distribution of seagrass in the region which is predominantly inshore (see 2.4.1 for further explanation). For all other regions, more than 80% of the seagrass in the region is within the Low class. There is no occurrence of seagrass meadows in the High class in Cape York and Burnett Mary, but there are limitations to the method applied to for Cape York described in Section 2.3.1c, and large seagrass meadows exist outside of the GBRWHA boundary and were not included in this assessment.
- The pie chart insets in Figure 2.13 show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the High class for TSS plume. The area of coral reef and seagrass in the Burdekin represents 68% and 88% of the total GBR coral reef area and seagrass area respectively of habitats within the High class across the GBR.
- The total extent of influence of all TSS plume loading classes varies between the regions but is greater than 50% of the total region area in all cases except for Burnett Mary which is 32%. Large proportions (>70%) of the Wet Tropics and Burdekin are within the total area of TSS exposure. The area of GBR lagoon waters in the High class is greatest in the Burdekin (4,553 km²), and second in the Fitzroy (763 km²), associated with large river discharges in the assessment period (Figure 2.13). The greatest area in the Medium class is in the Fitzroy (13,506 km²) with Mackay Whitsunday second (12,331 km²). The proportion of GBR lagoon waters in each region in the High class is less than 1%, except for the Burdekin which is 10%. In Cape York, 43% of the area is outside of the estimated plume loading area, although there are obvious uncertainties in the method used for this Region which must be taken into account (see Section 2.3.1c).
- When summed the coral reef area in the Medium and High classes for TSS plume loading represents 0% (Cape York) to 7.1% (Mackay Whitsunday) of the total coral reef area in the regions; seagrass ranges from 0% (Cape York) to 95% (Mackay Whitsunday) of the total seagrass area in the regions, and GBR lagoon waters from 0% (Cape York) to 26% (Mackay Whitsunday) of the total region area. The areas of lowest exposure are within the Cape York and Burnett Mary regions, although these results cannot be concluded with a great degree of certainty due to limited or no validation in those locations.



Figure 2.13. Results for the assessment of TSS plume loading (mean of annual assessments 2007 to 2011). The assessment classes are relative and derived from a combination of scaled river loads data and flood plume frequency analysis from remote sensing data (see methods in Section 2.3.1c). Inset bar chart compares coral reef and seagrass area within the High class; pie charts show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the High class.

b) Nutrients

Chlorophyll threshold exceedance 0.45 μ g/L

Chlorophyll (Chl) concentrations are relevant year round as an indication of nutrient enrichment in marine waters. As shown in Table 2.1, five assessment classes were used for Chl 0.45 μ g/L based on the frequency of exceedance of this concentration (in days) in the period 2002 to 2012, expressed as a percentage of the total number of valid daily observations ranging from Very Low to Very High. However, as the Very Low class receives a score of 0, it is not reported here. The results of the assessment are shown in Table 2.7 and Figure 2.14. The areas within the Very High class are constrained to the coast and concentrated in the Burdekin and Fitzroy, as was the case for TSS 2 mg/L (Section 2.4.2a). These inshore areas are locations with some of the highest use and visitation rates; this is a result common to all individual variables and is reviewed in the discussion.

Table 2.7. Area of coral reefs, seagrass meadows and GBR lagoon waters within the Low to Very High assessment classes for Chl 0.45 μ g/L and the percent of the NRM region that the area represents. Results for the assessment are based on frequency of exceedance of Chl 0.45 μ g/L using daily remote sensing data 2002-2012 (see methods in Section 2.3.1b).

Chl 0.45 μg/L	Low		Medium		High		Very High	
Frequency of exceedance class	1-10%		10-20%		20-50%		50-100%	
NRM Regions	Area (km²)	% of habitat in region						
Coral Reefs								
Cape York	8,364	81	1,324	13	614	6	<1	<1
Wet Tropics	2,063	85	288	12	65	3	<1	<1
Burdekin	2,623	88	256	9	34	1	12	<1
Mackay Whitsunday	707	22	1,599	50	846	26	10	<1
Fitzroy	2,463	51	1,620	33	707	15	46	1
Burnett-Mary	216	76	54	19	8	3	4	1
Seagrass								
Cape York	6,527	57	2,410	21	2,410	21	14	<1
Wet Tropics	4,055	83	554	11	204	4	45	1
Burdekin	5,435	89	79	1	231	4	315	5
Mackay Whitsunday	46	11	114	27	223	52	23	5
Fitzroy	5,042	87	291	5	230	4	196	3
Burnett-Mary	5,942	94	217	3	110	2	47	1
GBR lagoon waters								
Cape York	47,149	49	7,632	8	6,225	6	24	<1
Wet Tropics	17,750	56	2,033	6	3,218	10	617	2
Burdekin	29,989	64	2,072	4	2,439	5	2,202	5
Mackay Whitsunday	26,845	55	10,641	22	7,431	15	632	1
Fitzroy	63,314	74	9,886	12	7,601	9	2,140	3
Burnett-Mary	34,563	93	1,013	3	829	2	419	1

Key findings:

- The area of coral reef in the Very High class of Chl exceedance of 0.45 μg/L is greatest in the Fitzroy (46 km²), and second-greatest in the Burdekin (12 km²) (see inset in Figure 2.14). The greatest area within the High class is in Mackay Whitsunday (846 km²) and Fitzroy is second (707 km²). The proportion of coral reefs in each region in the Very High class is less than 1%. More than 75% of the coral reefs in Cape York, Wet Tropics, Burdekin and Burnett Mary are within the Low class, and 50% of the Fitzroy is within the Low class of frequency of exceedance of 0.45 μg/L.
- The area of seagrass within in the Very High class Chl exceedance of 0.45 μg/L is greatest in the Burdekin (315 km²), and the second-greatest area is the Fitzroy (196 km²) (see inset Figure 2.10). In the High class Cape York has the greatest area of seagrass (2,410 km²) with the Wet Tropics, Burdekin, Mackay Whitsunday and Fitzroy all similar with 200-230 km² of seagrass in the High class of Chl 0.45 μg/L exceedance. The proportion of seagrass meadows in each region in the Very High class is less than 5%. More than 80% of the seagrass meadows within the Wet Tropics, Burdekin, Fitzroy and Burnett Mary are within the Low class. The proportion of seagrass within the Medium and High categories in Cape York is around 20%. In Mackay Whitsunday 52% of the seagrass is within the High class of Chl exceedance.
- The lowest areas of coral reefs and seagrass in the Very High and High classes of Chl exceedance of 0.45 μ g/L are in Burnett Mary (12 km² and 157 km² respectively).
- The pie chart insets in Figure 2.14 show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the Very High class of Chl exceedance of 0.45 μg/L. The area of coral reef in the Very High class in Fitzroy represents 63% of the total GBR coral reef area within the Very High class, and the area of seagrass in the Very High class in the Burdekin represents 49% of the total GBR seagrass area within the Very High class.
- The area of GBR lagoon waters in the Very High class of Chl exceedance of 0.45 μg/L is greatest in the Burdekin (2,202 km²), and second in the Fitzroy (2,140 km²). The greatest area in the High class is in the Fitzroy (7,601 km²) with Mackay Whitsunday second (7,431 km²). The proportion of GBR lagoon waters in each region in the Very High class is less than 5%.
- When summed, the area of coral reef within the High and Very High classes (frequency of exceedance 20-50% and 50-100%) of Chl exceedance of 0.45 µg/L represents 1.54% (Burdekin) to 26.6% (Mackay Whitsunday) of the total coral reef area in the regions; seagrass ranges from 2.5% (Burnett Mary) to 57% (Mackay Whitsunday) of the total seagrass area in the regions, and GBR lagoon waters from 3.3% (Burnett Mary) to 17% (Mackay Whitsunday) of the total region area. The lowest total areas in the Very High and High classes are in Burnett Mary, however the assessment area is bound to the GBRWHA and therefore does not incorporate coral reefs that have not been fully mapped within the region, or the large seagrass meadows in Hervey Bay the large seagrass meadows in Hervey Bay.

The results presented here and used in the risk assessment have been compared with spatial patterns in Chlorophyll shown in De'ath and Fabricius (2008). The visual comparison of our outputs for Chl 0.45 μ g/L and those from De'ath and Fabricius (2008) are shown in Appendix 1 along with a similar comparison for TSS 2 mg/L. The spatial patterns are similar suggesting that the remotely sensed data used here are robust with a few key exceptions. In a similar pattern as TSS 2mg/L exceedance using remote sensing data, there is a distinct area of exceedance of the Medium class in coastal areas in the Cape York region north of Cooktown and Princess Charlotte Bay. Further validation of this result is required and may be associated with high turbidity interfering with the quantification of chlorophyll when TSS concentrations. Further assessment is required to quantify the concentrations of TSS that influence the Chl assessment. Similar patterns exist in the coastal areas around Shoalwater Bay in the northern part of the Fitzroy region which are also known to be naturally turbid, thereby potentially interfering with the Chl signal, and uncertainties in the remote sensing results in these areas have not been resolved. Similarly to TSS, these uncertainties are not considered to be significant enough to influence the overall conclusions of this assessment but may be important for regionally specific analyses in any further assessment of the data.



Figure 2.14. Results for the assessment of frequency of exceedance of Chl 0.45 µg/L using daily remote sensing data 2002-2012. Results for the assessment are based on frequency of exceedance of Chl 0.45 µg/L (see methods in Section 2.3.1b). Inset bar chart compares coral reef and seagrass area within the Very High class (50-100% exceedance); pie charts show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the Very High class.

DIN plume loading (mean 2007-2011)

As shown in Table 2.1, three assessment classes were used for DIN mean plume loading based on plume frequency information from remote sensing and scaled river load data. To factor in inter-annual variability, we used the mean result of assessments completed annually from 2007 to 2011. The results of the assessment are shown in Table 2.8 and Figure 2.15. Note that there are areas within the GBRWHA that are assessed as being outside of the area of plume loading influence which are not reported in this assessment. The areas within the highest assessment class (High) are constrained inshore (Figure 2.15) and mostly concentrated in the inshore southern portion of the Wet Tropics region and in the inshore areas of the Burdekin, Fitzroy and Burnett Mary regions.

Table 2.8. Area of coral reefs, seagrass and GBR lagoon waters within the assessment classes for DIN mean plume loadings between 2007 and 2011, and the percent of the NRM region that the area represents. The assessment classes are relative and derived from a combination of scaled river loads data and flood plume frequency analysis from remote sensing data (see methods in Section 2.3.1c).

DIN plume loadings	L	.OW	Me	dium	High		
NRM Regions	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region	
Coral Reefs							
Cape York	6,351	61	0	0	0	0	
Wet Tropics	2,262	93	74	3	25	1	
Burdekin	2,698	91	49	2	54	2	
Mackay Whitsunday	393	12	230	7	8	<1	
Fitzroy	584	12	140	3	138	3	
Burnett Mary	242	86	26	9	3	1	
Seagrass							
Cape York	10,911	96	0	0	0	0	
Wet Tropics	3,754	77	917	19	191	4	
Burdekin	4,106	67	1,328	22	642	11	
Mackay Whitsunday	<1	<1	360	84	51	12	
Fitzroy	1,926	33	3,320	57	519	9	
Burnett Mary	5,081	80	808	13	204	3	
GBR lagoon waters							
Cape York	38,149	40	0	0	0	0	
Wet Tropics	19,832	63	3,081	10	2,792	9	
Burdekin	19,917	43	6,553	14	7,140	15	
Mackay Whitsunday	9,893	20	14,725	30	898	2	
Fitzroy	14,949	17	15,308	18	10,527	12	
Burnett Mary	8,454	23	2,174	6	1,344	4	

Key findings:

• The area of coral reef in the High class of DIN plume loading is greatest in the Fitzroy (138 km²), and second-greatest in the Burdekin (54 km²) (see inset in Figure 2.15). The greatest area within the Medium class is in Mackay Whitsunday (230 km²) and Fitzroy is second (140 km²). The proportion of coral reefs in each region in the High class is less than 5%. More than 85% of the coral reef area in the Wet Tropics, Burdekin and Burnett Mary are within the Low class, and approximately 80% of the area of coral reefs in

the Mackay Whitsunday and Fitzroy are outside the mapped plume loading area. There is no or limited occurrence of coral reefs in the High class in Cape York and Burnett Mary, but there are limitations to the method applied to for Cape York described in Section 2.3.1c, and the area of coral reefs in the Burnett Mary is not considered to be accurate.

- The area of seagrass within the High class of DIN plume loading is also greatest in the Burdekin (642 km²), and the second-greatest area is in the Fitzroy (519 km²) (see inset Figure 2.15). In the Medium class Fitzroy has the greatest area of seagrass (3,320 km²) with Burdekin second (1,328 km²). The proportion of seagrass meadows in each region within the High class is less than 12% (Mackay Whitsunday). The proportion of the seagrass in each region that is in the Medium class is relatively high for Mackay Whitsunday (84%) and Fitzroy (57%). For all other regions, more than 60% of the seagrass in the region is within the Low class. There is no occurrence of seagrass meadows in the High class in Cape York, but there are limitations to the method applied to for Cape York described in Section 2.3.1c. In addition, the large seagrass meadows in the Burnett Mary region that are outside of the GBRWHA boundary were not included in this assessment.
- The pie chart insets in Figure 2.15 show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the High class of DIN plume loading. The area of coral reef in the High class in Fitzroy represents 61% of the total GBR coral reef area within the High class, and the area of seagrass in the High class in the Burdekin represents 40% of the total GBR seagrass area within the High class.
- The total extent of influence of all DIN plume loading assessment classes varies between the regions but is greater than 30% of the total region area in all cases, with 82% of the Wet Tropics and 72% of the Burdekin within the total area of DIN plume loading. Approximately half of the Fitzroy and Mackay Whitsunday are in the exposure area, around 40% of the Cape York region and 32% of the Burnett Mary region. The area of GBR lagoon waters in the High class is greatest in the Fitzroy (10,527 km²), and second in the Burdekin (7,140 km²), associated with large river discharges in the assessment period (Figure 2.15). The greatest area in the Medium class is also in the Fitzroy (15,308 km²) with Mackay Whitsunday second (14,725 km²). The proportion of GBR lagoon waters in each region in the High class ranges between 9 to 15% in the Burdekin, Wet Tropics and Fitzroy; the area of GBR lagoon waters in the High class in the other regions is less than 5%. In Cape York, 60% of the area is outside of the estimated plume loading area, although there are obvious uncertainties in the method used for this Region which must be taken into account (see Section 2.3.1c).

When summed the area of coral reef in the Medium and High classes of DIN plume loading represents 0% (Cape York) to 10% (Burnett Mary) of the total coral reef area in the regions; and seagrass area ranges from 0% (Cape York) to 96% (Mackay Whitsunday) of the total seagrass area in the regions. The lowest total areas within the plume loading influence area are within the Cape York and Burnett Mary regions, although these results cannot be concluded with a great degree of certainty due to limited or no validation in those locations.


Figure 2.15. Results for the assessment of DIN plume loading (mean of annual assessments 2007 to 2011). The assessment classes are relative and derived from a combination of scaled river loads data and flood plume frequency analysis from remote sensing data (see methods in Section 2.3.1c). Inset bar chart compares coral reef and seagrass area within the High class; pie charts show the area of coral reef and seagrass within each region as a percentage of total area (GBR-wide) of coral reef and seagrass within the High class.

COTS Initiation Zone

The COTS Initiation Zone has been identified as the area of highest risk with respect to initiating COTS primary outbreaks, described in further detail in the methods Section 2.3.1b and Chapter 1 of the Supporting Studies (Furnas et al. 2013a). This area is assessed here as the coral reef area between 14.5°S and 17°S inside the GBR Marine Park boundary. The areas affected by this zone are within the Cape York and Wet Tropics regions (see Figure 2.16). The total area of reefs affected is 3,284 km², of which 59% of the area is in the Cape York region. This accounts for 11% and 34% of the coral reefs in the Cape York and Wet Tropics regions respectively. Evidence for the degree to which each river in these regions (and in the Burdekin region) influences nutrient conditions in the Initiation Zone is included in Section 2.4.4(b).



Figure 2.16. Location of the COTS Outbreak Initiation Zone, defined as a high risk area for COTS primary outbreaks based on current understanding of the outbreak initiation zone between 14.5^oS and 17^oS. Refer to Furnas et al. (2013a) for further explanation.

c) Pesticides

PSII Herbicide modelled concentration, 2009-2011

As shown in Table 2.1, five assessment classes were used for PSII herbicides based on the toxicity of diuron calculated in several studies on coral and seagrass species (see Chapter 4 of the Supporting Studies, Lewis et al. 2013a) ranging from Very Low to Very High. These were then used for assessing the results of an estimate of the relationship between additive PSII herbicide concentrations and CDOM (salinity proxy) in flood plume conditions between January 2010 and March 2011 (in the 2009-2010 and 2010-2011 wet seasons; see Section 2.3.1c and Chapter 4 of the Supporting Studies). Note that no assessment has been undertaken for Cape York due to limited data availability, and was therefore assessed to be No risk; however, the risk in this region is unknown. The results of the assessment are shown in Table 2.9 and Figure 2.17. There are no marine areas in the Very High class, and marine areas within the Medium and High class are constrained inshore and only occur in the Mackay Whitsunday region (see Figure 2.17). The strong delineation between the regions in the mapping is due to regionally specific analysis of the data, which is explained further in Chapter 4 of the Supporting Studies.

Table 2.9. Area of coral reefs, seagrass and GBR lagoon waters within the Very Low to High assessment classes for exposure to PSII herbicides in 2009-2011 and the percent of the NRM region that the area represents. No areas were within the Very High assessment class (>10 μ g/L). Results for the assessment are based on the exposure assessment undertaken by Lewis et al. (2013a) and described in Section 2.3.1c.

	Ver	y Low	Low		Medium		High	
PSII herbicide concentration	<0.1	1 μg/L	0.1 -	0.5 μg/L	0.5 - 2.3 μg/L		2.3 - 10 μg/L	
NRM Regions	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region	Area (km²)	% of habitat in region
Coral Reefs								
Cape York	0	0	0	0	0	0	0	0
Wet Tropics	134	6	30	1	0	0	0	0
Burdekin	<1	<1	0	0	0	0	0	0
Mackay Whitsunday	127	4	113	4	18	1	6	<1
Fitzroy	28	1	6	<1	0	0	0	0
Burnett Mary	5	2	1	<1	0	0	0	0
Seagrass								
Cape York	0	0	0	0	0	0	0	0
Wet Tropics	744	15	247	5	0	0	0	0
Burdekin	10	<1	0	0	0	0	0	0
Mackay Whitsunday	84	19	205	48	82	19	57	13
Fitzroy	14	<1	<1	<1	0	0	0	0
Burnett Mary	40	1	45	1	0	0	0	0
GBR lagoon waters								
Cape York	0	0	0	0	0	0	0	0
Wet Tropics	4,360	14	3,015	10	0	0	0	0
Burdekin	1,124	2	0	0	0	0	0	0
Mackay Whitsunday	3,790	8	4,377	9	1,573	3	509	1
Fitzroy	1,295	2	751	1	0	0	0	0
Burnett Mary	427	1	357	1	0	0	0	0

Key findings:

- There were no areas within the Very High assessment class (>10 μg/L) of PSII herbicide modelled concentration in the GBR lagoon.
- The extent of PSII herbicide exposure for the High and Medium classes of PSII herbicide modelled concentration only occurs in the Mackay Whitsunday region (Figure 2.17 and inset bar graph). Under this scenario of High and Medium class conditions, up to 70% of phototrophic species would be affected with up to 20% of phototrophic species experiencing major impacts. This includes considerable inhibition of photosystem II as well as reductions in growth of coral reef and seagrass species. It should be remembered that these areas only characterise the additive risk of six PSII herbicides which is only a portion of the 34 pesticides that have been recorded in catchments at the end-of system (Chapter 4 of the Supporting Studies).
- The area of coral reef in the High class of PSII herbicide modelled concentration in the Mackay Whitsunday region is 6 km², with 18 km² in the Medium class, both representing less than 1% of the coral reef in the region (see Figure 2.17). The greatest areas within the Low class are in Mackay Whitsunday (113 km²) and Wet Tropics (30 km²), both representing less than 5% of the coral reef area in each region. In all regions that were assessed the greatest areas of coral reefs are in the Very Low class, with 134 km² in the Wet Tropics, 127 km² in Mackay Whitsunday and less than 30 km² in the Fitzroy, Burnett Mary and Burdekin.
- The area of seagrass within in the High class of PSII herbicide modelled concentration is greatest in the Mackay Whitsunday (57 km²), 82 km² in the Medium class (see inset Figure 2.17), representing 19% and 13% of the seagrass in the region respectively. The greatest areas within the Low class are in the Wet Tropics (247 km²) and Mackay Whitsunday (205 km²). This area in Mackay Whitsunday includes almost 50% of the seagrass meadows in the region. In all regions that were assessed, the greatest areas of seagrass are in the Very Low class, with 744 km² in the Wet Tropics, 84 km² in Mackay Whitsunday and less than 40 km² in the Fitzroy, Burnett Mary and Burdekin. This represents 19% of the seagrass in Mackay Whitsunday, and 15% of the seagrass in Wet Tropics.
- The total area of GBR lagoon waters in the Very Low to High classes of PSII herbicide modelled concentration is estimated to be 21,500 km². Approximately 50% of this area is in the Very Low class, 40% in the Low class and less than 10% in the Medium and High classes. In this assessment the Burdekin region is only exposed to PSII herbicide concentrations in the Very Low class, however it is noted that the influence of PSII herbicides is significant in the freshwater and estuarine environments in the Lower Burdekin (see below plus Chapter 4 of the Supporting Studies).

A limitation of this risk assessment was that only the major river within each region was assessed using the adopted method, i.e. Wet Tropics: Tully River; Burdekin: Burdekin River; Mackay Whitsunday: Pioneer River; Fitzroy: Fitzroy River; Burnett-Mary: Burnett River, in comparison to the plume loading assessment of TSS and DIN which accounted for annual flood plume assessments from all GBR catchments from 2007 to 2011. Therefore, the exposure area of PSII herbicides may be an underestimate.

Additional analysis using an ms-PAF method was undertaken for comparison with this assessment in Chapter 4 of the Supporting Studies (Lewis et al. 2013a). This method is a modification of the species sensitivity distribution methods used to calculate water quality guidelines for individual chemicals (e.g. ANZECC and ARMCANZ, 2000; GBRMPA, 2009). The ms-PAF method expands on the single chemical approach by incorporating multiple substances from a mixture into the calculation. It calculates the potentially affected

fraction (PAF) which is the percentage of species (s) that will theoretically be affected at a specified environmental concentration of each component of the mixture. Similar to the herbicide equivalent method, ms-PAF thresholds were allocated to each of the risk classifications.

The ms-PAF method also showed that the risk of herbicides in the Mackay Whitsunday region was also Medium to High and when multiple exposures were considered as well as discharge, the Pioneer River was considered to present the highest risk of PSII inhibiting herbicides in the GBR lagoon. The other NRM regions of the GBR fell within the Low or Very Low risk categories with the Wet Tropics having the next highest risk of PSII inhibitor herbicides to coral reefs and seagrass meadows. Overall, the two methods showed good agreement in their assessment of the risk posed by PSII inhibiting herbicides in the GBR lagoon from individual plumes.

While this assessment has provided a general regional prioritisation of risk for PSII inhibiting herbicides, we note that some of the smaller creeks in the regions were not properly assessed as shown by the analysis of the monitoring data in Table 2.10. For example, on occasion, Barratta Creek in the Burdekin region displayed chronic exposures of PSII herbicides that were rated as Very High (and hence a high risk to the Ramsar wetland and to Bowling Green Bay) and Sandy Creek in the Mackay Whitsunday region regularly had chronic exposures rated as High - Medium in 2009-2010.

The ms-PAF assessment showed that Barratta Creek within the Burdekin region reached a Very High risk (more than 70% phototrophic species with minor impacts and more than 20% of phototrophic species experiencing major impacts to populations) for one of the analysed periods; this result is of concern given that Barratta Creek feeds into high value Ramsar-listed wetlands. Indeed the smaller coastal creeks that contain relatively large areas of intensive agriculture (> 20%) such as Barratta and Sandy Creeks yielded the highest pesticide concentrations in monitoring programs.

It should be noted that this assessment of risk of PSII herbicides did not take into consideration the period of exposure to PSII herbicides within each region, as is considered for other water quality variables. Toxicity of PSII herbicides is time dependent (Vallotton et al. 2008), i.e. the toxicity of PSII herbicides to phototrophs increases with an increasing exposure duration. For this risk assessment, acute exposure was used to assess the potential impacts to seagrass and corals, however, for regions in where the exposure period lasts days or weeks, the risk from PSII herbicides would have been underestimated.



Figure 2.17. Results for the assessment of exposure to PSII herbicides based on an estimate of the relationship between additive PSII herbicide concentrations and CDOM (salinity proxy) in flood plume conditions from January 2010 to March 2011. Results for the assessment are based on the exposure assessment undertaken by Lewis et al. (2013a) and described in Section 2.3.1c. Inset bar chart compares reef and seagrass area exposed for the High class. Note that Cape York is assessed to be No risk but was not included in the remote sensing analysis and is therefore unknown.

Table 2.10. Risk assessment of individual events from the Wet Tropics, Burdekin/Haughton, Mackay Whitsunday, Fitzroy and Burnett NRM regions during the 2009-2010 wet season. Events assessed using satellite imagery are indicated by an asterisk. The Exposure Coverage Ratio accounts for repeated exposure to event and flood event plumes (an approximation of the aerial extent of each flood plume was calculated and compared to that of the appropriate flood plume with the satellite image and expressed as a ratio). Refer to Chapter 4 of the Supporting Studies (Lewis et al. 2013a) for further explanation of the methods used in this assessment.

	Event	Details		Risk Assessment			
Site	Event No	Duration	Discharge (GL)	Exposure Type	Risk Category	Exposure Coverage Ratio	
Wet Tropics							
South Johnstone River	1*	25 d 3 h	89.7	Chronic	Very Low	1.0	
	2	22 d 14 h	137.6	Chronic	Very Low	1.5	
Tully River	1*	19 d 22 h	244.1	Chronic	Low	1.0	
	2	13 d 6 h	215.4	Chronic	Low	0.9	
	3	23 d 22 h	540.5	Chronic	Very Low	2.2	
			Burdekin/	Haughton			
Barratta Creek	1	15 d 22 h	5.6	Chronic	Very High	0.2	
	2	12 d 18 h	92.2	Chronic	Moderate	3.5	
	3	16 d 3 h	55.5	Chronic	Very Low	2.1	
	4	17 d 10 h	30.7	Chronic	Very Low	1.2	
Burdekin River	1	46 d 7 h	4738.8	Chronic	Very Low	1.1	
			Mackay W	/hitsunday			
Pioneer River	1	7 d 19 h	74	Chronic	High	0.4	
	2*	4 d 23 h	196	Chronic	Moderate	1.0	
	3	13 d 17 h	332	Chronic	Very Low	1.7	
Sandy Creek	1	5 d 23 h	44	Chronic	High	0.7	
	2*	4 d 21 h	60	Chronic	Moderate	1.0	
	3	4 d 11 h	30	Chronic	Moderate	0.5	
	4	3 d 13 h	33	Acute	Moderate	0.6	
	5	4 d	30	Chronic	Moderate	0.5	
	6	7 d 3 h	63	Chronic	Low	1.1	
			Fitz	roy			
Fitzroy River	1*	10 d 21 h	1419.3	Chronic	Very Low	1.0	
	2	23 d 9 h	4408.9	Chronic	Low	3.1	
	3	14 d 16 h	3072.7	Chronic	Very Low	2.2	
	4	24 d 15 h	1815.4	Chronic	Very Low	1.3	
	1		Bur	nett			
Burnett River	1*	33 days, 2	856.0	Chronic	Very Low	1.0	
		hour					

d) Comparisons among variables and Regions

The assessment of individual variables presented in Section 2.4.2 above (a – sediment, b - nutrients and c - PSII herbicides) can be used to guide priorities for management of individual pollutants between NRM regions. The different ways of reporting TSS and nutrient variables provide different perspectives of either acute (surface exposure mapping) or chronic (long term exceedance) effects of the variables on coral reefs and seagrass.

To compare results among regions, the areas of coral reefs, seagrass, and GBR lagoon waters in the highest assessment class for each variable (see Table 2.1 and 2.3) has been set as an anchor point (see Methods Section 2.3.2) and given a value of 100. All other area calculations were then expressed as a percentage of the maximum (values between 0 and 100), as shown in Table 2.11, to show relative differences between regions. Summaries of these results are provided below.

Table 2.11. Anchored scores for the area of coral reefs and seagrass for each NRM region affected by the highest assessment classes for the water quality variables included in the risk analysis (final output of Part 1 in Figure 2.3). In the case of PSII herbicides, the two highest assessment classes ware used. The Region that had the largest area affected was given a score of 100; all other Regions are expressed as a percentage based on the area affected in each Region relative to the area in the Region with the maximum area affected. To highlight differences between Regions, cells with the greatest and second-greatest areas affected are shaded dark and light grey respectively. Refer to Table 2.1 for further explanation of the variables.

		Sediments				Pesticides	
NRM regions	TSS exceedance 2mg/L	TSS exceedance 7mg/L	TSS plume loading 07-11	Chl exceedance 0.45 ug/L	DIN plume loading 07-11	COTS Initiation Zone	PSII herbicide modelled concentration
Coral Reefs							
Cape York	0	0	0	0	0	100	0
Wet Tropics	0	0	9	1	18	70	0
Burdekin	100	4	100	26	39	0	0
Mackay Whitsunday	14	0	6	22	6	0	100
Fitzroy	61	100	32	100	100	0	0
Burnett Mary	18	0	0	9	2	0	0
Seagrass						N/A	
Cape York	0	0	0	5	0		0
Wet Tropics	11	27	8	14	30		0
Burdekin	100	100	100	100	100		0
Mackay Whitsunday	2	0	1	7	8		100
Fitzroy	5	27	5	62	81		0
Burnett Mary	9	0	0	15	32		0
GBR lagoon waters						N/A	
Cape York	0	0	0	1	0		0
Wet Tropics	16	6	10	28	27		0
Burdekin	100	34	100	100	68		0
Mackay Whitsunday	14	3	3	29	9		100
Fitzroy	54	100	17	97	100		0
Burnett Mary	16	0	0	19	13		0

Sediments

Assessment of the exceedance of TSS 2 mg/L (annual) results provides an indication of suspended sediment conditions in the GBR against the Water Quality Guidelines year round. The areas within the highest assessment classes are constrained to the coast and mostly concentrated in the Burdekin and Fitzroy regions. This is a similar pattern for the exceedance of the higher categories of the greater concentration of TSS 7 mg/L which is known to have measurable effects on coral and seagrass health, and fish behaviour (refer to Chapter 3 of the Supporting Studies Brodie et al. 2013a) and is therefore of greater significance when considering potential ecosystem impacts.

For TSS exceedance of 2 mg/L, the area of coral reef in the highest assessment class (Very High) is relatively small but is greatest in the Burdekin region, and second-greatest in the Fitzroy region (61%). For the higher concentration of TSS exceedance of 7 mg/L, the area within the Very High class is extremely constrained to the coast, and is by far the greatest in the Fitzroy region.

The area of seagrass within the Very High class of TSS exceedance of 2 mg/L and 7 mg/L is greatest in the Burdekin region. The area of seagrass in these areas is significantly larger than the areas in other regions; in both cases the next greatest area is the Wet Tropics region with less than one quarter of the seagrass area affected compared to the Burdekin.

The plume loading mapping for TSS over the period 2007 to 2011 provides an indication of the extent of the influence of flood plumes on the GBR in terms of frequency and extent of influence of flood plumes with differing characteristics of suspended sediment concentrations. The area of coral reef and seagrass in the highest assessment class (High) is by far the greatest in the Burdekin region.

The area of seagrass affected by all of the sediment variables is greatest in the Burdekin. The area of seagrass within the Wet Tropics is second for all variables but is less than one quarter of the areas affected in the Burdekin in all cases.

The area of GBR lagoon waters within the highest assessment classes for the sediment variables follows the same pattern as for coral reefs. The area of GBR lagoon waters within the Very High assessment class for TSS exceedance of 2 mg/L is greatest in the Burdekin and is double that of the Fitzroy. For TSS exceedance of 7 mg/L (5 NTU) the area affected in the Very High assessment class is greatest in the Fitzroy, and triple of that in the Burdekin. For all three sediment variables all of the other regions have less than 17% of the area affected in the Burdekin and Fitzroy.

Nutrients

Assessment of Chl exceedance of 0.45 μ g/L results provides an indication of nutrient conditions in the GBR against the Water Quality Guidelines year round. The area of coral reefs affected by the highest assessment classes of the nutrient variables (Chl exceedance 0.45 μ g/L Very High, and DIN plume loading 2007-2011 High) is greatest in the Fitzroy and then the Burdekin region, but in both cases the area within the Burdekin is less than 40% of that in the Fitzroy. All other regions have less than one quarter of the coral reef area within the highest assessment classes compared to the Fitzroy.

The area of seagrass within the highest assessment classes of the nutrient variables is greatest in the Burdekin region, and then the Fitzroy, but in both cases the relative difference is not as great as that for coral reefs (62% for Chl exceedance of 0.45 μ g/L and 81% for DIN plume loading). The other regions have less than one third of the seagrass area within the highest assessment classes compared to the Burdekin and in the case of Cape York is less than 5%.

The area GBR lagoon waters within the highest assessment class for Chl excedance of 0.45 μ g/L is greatest and very similar in the Burdekin and Fitzroy; these regions switch for DIN plume loading with Fitzroy having the greatest area of regional waters affected and the Burdekin region being second (68%). The other regions have less than one third of area of GBR lagoon waters within the highest assessment classes compared to the maximum, and in the case of Cape York is less than 1%.

Only coral reefs in the Cape York and Wet Tropics regions are counted within the COTS Initiation Zone. The area of coral reefs is 0% for all other regions. The maximum area of coral reefs within the COTS Initiation Zone is in Cape York, and the area in Wet Tropics is equivalent to 70% of this area.

PSII Herbicides

Based on the methods used here, coral reefs, seagrass, and GBR lagoon waters are only affected by the highest assessment classes of pesticides (Very High and High) in the Mackay Whitsunday region. This is a result unique to pesticides. However, it should be noted that the high risks of PSII herbicides to wetland, estuarine and coastal habitats, which provide important ecosystem services to the GBR including fish nursery habitats, were not included in this assessment. However, further investigation was conducted in the support studies of pesticide risk included as Chapter 4 of the Supporting Studies (Lewis et al. 2013a. It showed that the ecosystems of highest risk from pesticides are those coastal habitats (e.g. wetlands, estuaries, mangroves and seagrass) which provide important ecological services to GBR biota, including nursery habitats, primary productivity and nutrient cycling. The assessment concluded that the risk of pesticides in the GBR can be classified into five separate areas of ecological importance with assessments ranging from Very Low to Very High.

- Within the *freshwater reaches of rivers and freshwater/coastal wetlands*, the risk of pesticides (PSII herbicides and some non-PSII pesticides) is rated in the Very High class (>70% of phototrophic species affected), High (5 40% of phototrophic species affected) and Medium class (5 40% of phototrophic species affected depending on the region and stream) particularly for the coastal stream networks that drain a relatively large area (> 20%) of intensive agriculture such as Barratta and Sandy Creeks.
- In the *estuarine reaches of the rivers*, the risk of pesticides is largely in the High to Low category (1 70% of phototrophic species affected). A similar risk occurs for the *coastal marine environment* including intertidal and subtidal seagrass meadows (major reduction in photosynthesis by between 50% and 90% seagrass and microalgae (High) and up to 10% photosynthesis reduction (Low)).
- Coral reefs and seagrass meadows on the inner shelf whichincludes but not limited to areas
 extending up to 20 km from the coast, generally fall into the Low (reduction in photosynthesis by
 between 10% and 50% in corals, seagrass and microalgae) to Very Low categories (no observed
 effect on corals, seagrass or microalgae; < 40% of phototrophic species affected) depending on the
 region and adjacent catchment(s).
- The risk to *coral reefs on the mid and outer shelf* is considered Very Low to no risk (< 1% phototrophic species affected).

For all variables included in the risk assessment except for PSII herbicides, at least two (for COTS Initiation Zone) and otherwise 4 or 5 other regions have areas of coral reefs, seagrass and GBR lagoon waters in the highest assessment classes of the water quality variables. The common exception is Cape York; no coral reef area is affected by the highest assessment classes of all sediment variables, or Chl 0.45 μ g/L and DIN plume loading. Further, no seagrass or GBR lagoon waters in Cape York are within the highest assessment classes of all sediment variables, and DIN plume loading 2007-2011 (see Table 2.11).

2.4.3 Part 2: Combined risk of degraded water quality to GBR ecosystems

The combined assessment takes into account all assessment classes for each variable to identify the areas of highest relative risk to degraded water quality in the GBR and hence where coral reefs and seagrass are most likely to be under pressure from degraded water quality.

To consider the combined risk of the selected water quality variables to coral reefs and seagrass, the results for the individual variables described above were summed at the 1 km² pixel scale and normalised between 0 and 1 using the Multi-Criteria Analysis Shell for Spatial Decision Support (MCAS-S) (see methods Section 2.3.4). The results were further classified into five even break classes ranging from Very Low to Very High. The results of combining the variables are shown in Figure 2.18 and the area of coral reefs, seagrass meadows and GBR lagoon waters within each of those 5 classes is shown in Table 2.12.

A Marine Risk Index was calculated by summing the areas of coral reefs, seagrass and GBR lagoon waters in the Very High and High classes and anchoring those results to the maximum areas. This shows relative differences between regions (see final column in Table 2.12) by comparing the total area of habitat at risk among regions. However, the proportion of coral reefs and seagrass in each region that is within the Marine Risk Index is also presented as this is most relevant for determining management priorities within a region.



Figure 2.18. Combined assessment (1 km² resolution) of the relative risk of water quality variables (output of Part 2 shown in Figure 2.2). The areas (in km²) of habitat types within each class are shown in Table 2.12. The Marine Risk Index for coral reefs and seagrass is defined from the combined areas of the High and Very High relative risk classes. Inset bar chart shows reef and seagrass areas in the Marine Risk Index by Region; pie charts show affected coral reef and seagrass areas for each Region as a percentage of the total GBR coral reef and seagrass areas in the Marine Risk Index.

Table 2.12. Area of coral reefs, seagrass meadows, and GBR lagoon waters within the 5 relative risk classes. The sum of the area within the High and Very High classes form the Risk Index, which compares all summed areas to the maximum area, which is given a score of 100. The greatest area/highest scores are shaded dark grey, second is shaded light grey.

Habitat NRM Regions	V Low	Low	Medium A	High area (km²)	V High	Total	Sum High & V High	% of habitat in Region (High &	Marine Risk Index
Coral Reefs								v mgm	
Cape York	3,585	6,147	546	17	0	10,295	17	<1	15
Wet Tropics	121	1,809	448	36	<1	2,415	36	1	32
Burdekin	704	2,206	24	14	<1	2,948	15	<1	13
Mackay Whitsunday	2,147	826	133	61	5	3,171	66	2	59
Fitzroy	3,726	964	38	103	8	4,840	111	2	100
Burnett Mary	10	265	1	5	0	282	5	2	4
Seagrass									
Cape York	1,258	8,711	1,329	32	0	11,330	32	<1	7
Wet Tropics	175	4,240	278	102	59	4,855	161	3	34
Burdekin	1,281	4,160	150	333	136	6,060	470	8	100
Mackay Whitsunday	5	21	210	136	25	396	160	37	34
Fitzroy	201	5,050	177	303	15	5,746	319	6	68
Burnett Mary	458	5,561	210	76	7	6,313	83	1	18
GBR lagoon waters									
Cape York	59,433	30,785	3,823	73	0	94,114	73	<1	2
Wet Tropics	13,544	12,558	2,698	1,777	462	31,038	2,239	7	51
Burdekin	22,258	17,138	3,313	2,794	777	46,280	3,571	8	82
Mackay Whitsunday	26,300	12,309	5,964	3,433	326	48,332	3,759	8	86
Fitzroy	48,084	26,371	6,482	3,696	660	85,293	4,355	5	100
Burnett Mary	26,127	9,565	848	658	54	37,252	712	2	16

Key findings:

- The greatest area of coral reef within the Very High and High relative risk classes is in the Fitzroy region (see inset Figure 2.18 and Figure 2.19). These classes are summed for the Marine Risk Index and reflect that less than one third of the coral reef area affected in the Fitzroy is affected in all other regions except Mackay Whitsunday (59%). However, the proportion of coral reefs in each region that are within the High and Very High classes are approximately equal for all regions and is less than 2%.
- The area of coral reef in the Very High class of relative risk is greatest in the Fitzroy (8 km²), and second-greatest in Mackay Whitsunday (5 km²). Less than 1 km² of coral reef area is within the Very High class for the other four regions (Table 2.12). The greatest area within the High class is in Fitzroy (103 km²) and Mackay Whitsunday is second (61 km²), and less than 72 km² in total for all of the other regions combined (Table 2.12 and Figure 2.19). In the Medium class (not in the Marine Risk Index but shown in Table 2.12) the greatest area of coral reef is within Cape York (546 km²) and second in the Wet Tropics (448 km²). However, the relatively large area of coral reefs within the Medium class in Cape York is assessed with low confidence due to issues with the remote sensing of

Chl and TSS exceedance identified in the above, in addition to the method used for allocating DIN and TSS plume loading in the Region; all of these factors require further validation. The largest areas of coral reefs are within in the Very Low or Low classes for all regions.

- The greatest area of seagrass within the Very High and High relative risk classes is in the Burdekin region (see inset Figure 2.18 and Figure 2.19). These classes are summed for the Risk Index and reflect that less than one third of the seagrass area affected in the Burdekin is affected in all other regions except Fitzroy (68%). The proportion of seagrass in each region that is within the Very High and High classes is less than 10% of the total seagrass area within all regions except for Mackay Whitsunday which is 37%.
- The area of seagrass in the Very High relative risk class is greatest in the Burdekin (136 km²), and second-greatest in the Wet Tropics (59 km²), and less than 47 km² of seagrass area is within all of the other four regions combined (Figure 2.18, 2.19 and Table 2.12). The greatest area within the High class is in Burdekin (333 km²), second-greatest in the Fitzroy (303 km²), and less than 346 km² in all of the other regions combined. When the sum of the area in the Very High and High classes are expressed as a percentage of the total seagrass area in each region, the proportion is highest in Mackay Whitsunday at 37%, and second-highest in the Burdekin (8%) and Fitzroy (6%). However, the total area of seagrass affected is far greater in the Burdekin and Fitzroy; the differences in percentages indicate how much greater the total seagrass areas are in the Burdekin and Fitzroy regions (6,060 km² and 5,746 km² respectively; Figure 2.10) compared to 430 km² for the Mackay Whitsunday region. In the Medium class (not in the Marine Risk Index but shown in Table 2.12) the greatest area of seagrass is by far the greatest in Cape York (1,329 km²). However, as noted above further validation of the datasets in the Cape York region is required.
- The greatest area of GBR lagoon waters within the Very High and High relative risk classes is in the
 Fitzroy (Figure 2.18). These classes are summed for the Marine Risk Index and reflect that the area of
 GBR lagoon waters in the Marine Risk Index are reasonably comparable for the Fitzroy (100%),
 Mackay Whitsunday (86%) and Burdekin (82%). The result in the Wet Tropics is approximately half of
 the area affected in the Fitzroy, 16% in Burnett Mary and less than 5% in Cape York.
- The area of GBR lagoon waters in the Very High relative risk class is greatest in the Burdekin (777 km²) and second in the Fitzroy (660 km²). The greatest area within the High class is in the Fitzroy (3,696 km²), Mackay Whitsunday (3,433 km²) and Burdekin (2,794 km²) regions (Table 2.12 and Figure 2.19). When the sum of the area in the Very High and High classes are expressed as a percentage of the total area in the regions, the percentage is less than 10% for all regions. For the Medium class, unlike coral reefs and seagrass where the greatest area is within the Cape York region (not in the Marine Risk Index but shown in Table 2.12), the greatest total area affected is in the Fitzroy (6,482 km²) and Mackay Whitsunday (5,964 km²) regions.

While the areas of coral reef and seagrass within the highest assessment classes for individual variables and the Marine Risk Index are relatively small, they often include highly valued tourism and recreation sites of the GBR. Examples include Fitzroy Island, Hinchinbrook Island, Magnetic Island, many of the islands in the Whitsunday Group and the Keppel Island group. In the case of seagrass meadows, many of the highest risk areas overlap with dugong protection areas (DPAs), which are assigned because of the large populations of dugongs feeding in the associated seagrass meadows.

In summary, the combined assessment of water quality variables can be used to guide overall management priorities for addressing the risks from degraded water quality to coral reefs and seagrass between NRM regions. When all water quality variables are combined into the Marine Risk Index, the risk is greatest for coral reefs in the Fitzroy and Mackay Whitsunday regions, and for seagrass in the Burdekin and Fitzroy regions. The total areas influenced by the Marine Risk Index are greatest in the Fitzroy, Mackay Whitsunday and Burdekin regions, which may be important for habitats other than corals and seagrass. In most cases,

the proportion of the habitat area in each Region that is within the High and Very High relative risk classes (forming the Marine Risk Index) is less than 10%, except in the case of seagrass meadows in the Mackay Whitsunday region where 37% of the area of seagrass in the region is within the Marine Risk Index area.



Figure 2.19. Area of coral reefs, seagrass, and GBR lagoon waters within the Medium, High and Very High relative risk classes from the combined water quality assessment. The y-axis scale differs among the plots.

2.4.4 Part 3: Relative risk of degraded water quality to GBR ecosystems

To inform management priorities to address the risks identified above, it is necessary to understand the influence of river discharge in each of the Regions, as these discharges carry a majority of the anthropogenic pollutant load into the GBR lagoon. The results of this part of the assessment are presented in two components: a) assessment of the pollutant loads and b) combining the marine relative risk (Marine Risk Index) with the pollutant load information (Loads Index).

a) Assessment of end-of-catchment pollutant loads

Understanding the sources of pollutants is important for management prioritisation, and when considering river discharges to the GBR, the anthropogenic pollutant loads are the most relevant. However, in the case of pesticides, it is the toxicity and therefore concentration that is most relevant but concentration data is limited across the GBR. Therefore, in the final conclusions relating to pesticide risk the evidence is drawn from a combination of load and concentration data from specific locations.

To relate the results of the Marine Risk Index to catchment influences, the anthropogenic loads were expressed as the proportion of total GBR load for each Region for TSS, DIN, and PSII herbicides (Table 2.13). This recognises that while the total GBR load is important in influencing the marine water quality conditions, it is only the anthropogenic proportion that can be factored into management. These proportional contributions were then anchored (to normalise to a standard scale) and averaged to generate a Loads Index (Table 2.14). This assumes that the relative importance of each load is equal, which in reality is unlikely to be the case, although there is currently insufficient knowledge to weight the importance of the three pollutants relative to each other.

Once the marine relative risks and relative pollutant loads are known, the pollutant load information allows managers to track back to management priorities within catchments and land uses. To inform these decisions, further analysis of catchment pollutant loads could be undertaken for TSS, DIN, and PSII herbicides and include: comparisons between catchments and among regions of the total and anthropogenic load contributions to GBR total loads, and identification of the top 5 catchments in terms of the anthropogenic load expressed as a proportion of the GBR total load. For example, based on modelling:

- The top 5 contributors to the total GBR TSS anthropogenic load are (from 1-5) [Catchment (Region, catchment load)]: Burdekin River (Burdekin, 2318 kilotonnes), Fitzroy River (Fitzroy, 1362 kilotonnes), Herbert River (Wet Tropics, 352 kilotonnes), Mary River (Burnett-Mary, 301 kilotonnes), and the Don River (Burdekin, 211 kilotonnes).
- The top 5 contributors to the total GBR DIN anthropogenic load are (from 1-5): the Johnstone River (Wet Tropics, 851 tonnes), Fitzroy River (Fitzroy, 475 tonnes), Haughton River (Burdekin, 474 tonnes), Burdekin River (Burdekin, 401 tonnes), and the Tully River (Wet Tropics, 345 tonnes).
- The top 5 contributors to the total GBR PSII herbicide load are (from 1-5): Herbert River (Wet Tropics, 2,660 kg,), Johnstone River (Wet Tropics, 2401 kg), Mulgrave-Russell River (Wet Tropics, 1,776 kg), Tully River (Wet Tropics, 1,740 kg), and the Haughton River (Burdekin, 1,426 kg). However, these figures are currently under review and show distinct differences from monitoring data (see below) which reflects annual variability.
- There are 7 rivers with average anthropogenic contributions of combined TSS, DIN and PSII herbicides loads exceeding 5%, ordered from highest to lowest average contribution here: (1) Burdekin River (17% average), (2) Fitzroy River (12%), (3) Johnstone River (12%), (4) Herbert River (10%), Haughton River (7.1%), Tully River (7.3%) and the Mulgrave-Russell River (5.7%).

More detailed analyses of the load data are outside of the scope of this assessment but are recommended to be undertaken to guide future management priorities.

For PSII herbicides the concentrations are more ecologically relevant than loads. For determining risk to aquatic biota from PS II herbicides, assessing toxic effects from concentration data (Section 2.4.5) is a more ecologically relevant method than an assessment of the PSII load transported to the marine environment. These issues are illustrated with the most recent annual loads based on monitoring data from the 2010-11 monitoring year. By region, Fitzroy was the greatest contributor (66%), followed by Mackay Whitsunday (15%), Burdekin/Haughton (10%), Wet Tropics (6%) and Burnett (3%). The greatest contributor of total PSII herbicide loads was the Fitzroy River (8,532 kg, 66% of total GBR load), followed by the Pioneer River (1,254 kg, 10%), Burdekin River (886 kg, 7%), Sandy Creek (701 kg, 5%), Tully River (455 kg, 4%), Burnett River (446 kg, 3%), Barratta Creek (358 kg, 3%), Herbert River (287 kg, 2%) and Johnstone River (58 kg, <1%). However, the PSII modelled load does provide an indication of the contribution of PSII herbicides from each catchment based on an 'average' year, i.e. a long term average that adjusts for extreme weather conditions.

Table 2.13. Total and total anthropogenic loads for DIN, TSS and pesticides from rivers discharging into the marine NRM regions and, for each region, as percentages of the total GBR load, and total regional and GBR anthropogenic load.

Parameter NRM Regions	Total Load	Anthropogenic Load	Regional Anthropogenic load % of Total GBR Load
TSS (000s tonnes)			
Cape York	495	229	3
Wet Tropics	1,228	782	9
Burdekin	4,104	2,807	32
Mackay Whitsunday	514	363	4
Fitzroy	2,034	1,487	17
Burnett Mary	475	382	4
Total	8,850	6,051	
DIN (tonnes)			
Cape York	492	5	<1
Wet Tropics	4,437	2,023	20
Burdekin	2,352	1,114	11
Mackay Whitsunday	901	627	6
Fitzroy	1,272	503	5
Burnett Mary	563	442	4
Total	10,018	4,714	
PSII Herbicides (kg)			
Cape York		5	<1
Wet Tropics		10,229	61
Burdekin		2,219	13
Mackay Whitsunday		2,045	12
Fitzroy		638	4
Burnett Mary		1,556	9
Total		16.692	

To generate a Loads Index from the information presented in Table 2.13, we averaged the Regional anthropogenic load contributions as a proportion of the total GBR load for TSS, DIN and PSII herbicides. These results were then anchored it to the maximum value to generate a relative assessment of load contributions to the GBR, forming a Loads Index for each Region (Table 2.14). The assessment shows the greatest relative contribution of combined end of catchment loads to the GBR is from the Wet Tropics

region, followed by the Burdekin region. The contributions from the Mackay Whitsunday, Fitzroy and Burnett Mary regions are similar and are around one quarter of the contributions of the Wet Tropics. The load contributions from the Cape York region are minor compared to all other regions.

Table 2.14. Loads Index for TSS, DIN and PSII Herbicides derived from the average of the Regional anthropogenic load contributions to the total GBR load. The Region that had the largest average load was given a score of 100; all other Regions are expressed as a percentage based on the area affected in each Region relative to the area in the Region with the largest average. Source: Derived from Waters et al. (in press).

Regional Anthropogenic Load as a proportion of the Total GBR Load							
Region	TSS	DIN	PSII	Average	Loads Index	Loads Index Rank	
Cape York	2.6	0.05	0.03	0.04	0	6	
Wet Tropics	8.8	20.2	61.3	30.1	100	1	
Burdekin	31.7	11.2	13.3	18.7	62	2	
Mackay Whitsunday	4.1	6.3	12.2	7.5	25	4	
Fitzroy	16.8	5	3.8	8.5	28	3	
Burnett Mary	4.3	4.4	9.3	6.0	20	5	
			Мах	30.1			

An important factor in attributing the Marine Risk Index to the influence of the Regions or individual rivers is that the rivers in each Region may influence a greater area than just that within the marine NRM region boundary which has been directly assigned to each of the 6 NRM regions. The sheer size and episodic nature of river flow in the large catchments of the Burdekin and Fitzroy regions means that marine areas and habitats outside of the associated marine NRM regions may also be affected by these rivers. For example, satellite imagery during periods of high flow has shown that the Burdekin River influences areas as far north as the Wet Tropics marine NRM region, and that the Fitzroy River may influence the Mackay Whitsunday marine NRM region. Without more detailed analysis of river flow and hydrodynamics in the GBR, it is necessary to make the assumption that the rivers in each catchment NRM region only contribute to the risk in the corresponding marine NRM region. This is noted as a limitation to this aspect of the assessment, and suggestions on how this might be addressed for future assessments are included in Section 4.

However, cross regional influence of a selection of GBR rivers has been considered in the assessment of the influence of river discharge on the COTS Initiation Zone in Chapter 1 of the Supporting Studies (Furnas et al. 2013a). This is considered to be an important factor in the context that over 40% of the loss of coral cover in the GBR since 1987 is attributed to COTS (De'ath et al. 2012) and river discharges are known to play an important role in driving primary outbreaks (Furnas et al. 2013a). Furnas et al. (2013a) defined the volumetric contribution of river discharge and DIN loading from 8 rivers to the COTS Initiation Zone between Lizard Island in Cairns (Chapter 1 of the Supporting Studies, Table 1.3c).

For this assessment, we used the total volumetric contribution of river discharge to the COTS Initiation Zone (Table 2.15). An assessment of modelled DIN load data was also investigated but the uncertainties associated with limited validation with monitoring results were considered to be too limiting at this time. These included: the Normanby River in the Cape York region; the Daintree, Barron, Russell-Mulgrave, Johnstone, Tully and Herbert Rivers in the Wet Tropics region; and the Burdekin River in the Burdekin region. The greatest contribution was identified from the Daintree River, and all other contributions were normalised against the Daintree to show relative contributions (Table 2.15). The proportion that each Region contributes was then calculated by summing all values and presenting each region as a percentage of the total. The majority of the estimated influence (86%) is from the Wet Tropics rivers and a small proportion (14%) from the Burdekin River (Table 2.15 below). This information was incorporated into an additional

factor, the 'COTS Influence Index' by anchoring these results which are then 100% for the Wet Tropics region, 16% for the Burdekin region and zero contribution from all other Regions. While this assessment relies on many assumptions associated with DIN loads in river discharges and differences in inter annual variability, it does provide a way of recognising the relative influence of the regional river discharges to the ecologically important initiation area of COTS primary outbreaks.

Table 2.15. Relative volumetric contribution of individual rivers to the COTS outbreak initiation zone between Cairns and Lizard Island. The relative contributions of different rivers were normalised against the largest riverine contribution to the region, the Daintree. The COTS Load Influence is based on the proportional contribution of the Regions to the COTS Initiation Zone, and then anchored to the maximum value and expressed as a relative percentage (0-100).

River	Volumetric Contribution to the COTS Initiation Zone	NRM region	COTS Load Influence by NRM region	COTS Load Index (Anchored to max COTS Load Influence) by NRM region
Normanby	0	Cape York	0	0
Daintree	100	Wet Tropics	86	100
Barron	52			
Russell-				
Mulgrave	59			
Johnstone	29			
Tully	57			
Herbert	7			
Burdekin	49	Burdekin	14	16
Total	353			
		Mackay		
		Whitsunday	0	0
		Fitzroy	0	0
		Burnett Mary	0	0

b) Combined assessment: Relative Risk Index

Using the information obtained through the above analyses for the marine water quality variables and end of catchment pollutant loads, it is possible to make an assessment of the management priorities for minimising the risk of water quality impacts in the GBR. This section presents an option for a quantitative combined assessment to inform water quality management priorities across the NRM regions in the GBR. In this assessment the limitation identified above related to the assumption that river influence from an NRM region is constrained to that same marine NRM region is recognised.

As described in the methods (Section 2.3.5), to provide an overall ranking of relative risk between the NRM regions the Loads Index, COTS Influence Index (for coral reefs) and the Marine Risk Index for coral reefs and seagrass meadows were combined to generate a Coral Reef Relative Risk Index and Seagrass Relative Risk Index (Table 2.16). For coral reefs, the two loads related indexes, i.e. the COTS Influence Index and the Loads Index, were combined by summing the scores and then anchoring the result. For seagrass only the Loads Index and the Marine Risk Index are used. The final indexes for coral reefs and seagrass were then summed and anchored to provide an overall assessment of the relative risk of water quality to coral reefs and seagrass meadows – the Relative Risk Index (Table 2.17); summarised in Figure 2.20.

Table 2.16. Results of the overall risk assessment from summing the Loads, COTS Influence (for coral reefs only) and Marine Risk Index for coral reefs and seagrass. The Region that had the maximum value was given a score of 100; all other Regions are expressed as a percentage based on the value in each Region relative to the area in the Region with the maximum value.

Coral Reefs	Coral Reef	Loads &	Sum of	Coral Reef	Rank
	Iviarine Risk		Indexes	Relative	
	Index	Index		Risk Index	
NRM region				(Anchored)	
Cape York	15	0	15	12	5
Wet Tropics	32	100	132	100	1
Burdekin	13	39	52	40	4
Mackay Whitsunday	59	13	72	54	3
Fitzroy	100	14	114	86	2
Burnett Mary	4	10	14	11	6
		Max	132		
Seagrass	Seagrass	Loads	Sum of	Seagrass	Rank
Seagrass	Seagrass Marine Risk	Loads Index	Sum of Indexes	Seagrass Relative	Rank
Seagrass	Seagrass Marine Risk Index	Loads Index	Sum of Indexes	Seagrass Relative Risk Index	Rank
Seagrass NRM region	Seagrass Marine Risk Index	Loads Index	Sum of Indexes	Seagrass Relative Risk Index (Anchored)	Rank
Seagrass NRM region Cape York	Seagrass Marine Risk Index 7	Loads Index	Sum of Indexes 7	Seagrass Relative Risk Index (Anchored) 4	Rank 6
Seagrass NRM region Cape York Wet Tropics	Seagrass Marine Risk Index 7 34	Loads Index 0 100	Sum of Indexes 7 134	Seagrass Relative Risk Index (Anchored) 4 83	Rank 6 2
Seagrass NRM region Cape York Wet Tropics Burdekin	Seagrass Marine Risk Index 7 34 100	Loads Index 0 100 62	Sum of Indexes 7 134 162	Seagrass Relative Risk Index (Anchored) 4 83 100	Rank 6 2 1
Seagrass NRM region Cape York Wet Tropics Burdekin Mackay Whitsunday	Seagrass Marine Risk Index 7 34 100 34	Loads Index 0 100 62 25	Sum of Indexes 7 134 162 59	Seagrass Relative Risk Index (Anchored) 4 83 100 37	Rank 6 2 1 4
Seagrass NRM region Cape York Wet Tropics Burdekin Mackay Whitsunday Fitzroy	Seagrass Marine Risk Index 7 34 100 34 68	Loads Index 0 100 62 25 28	Sum of Indexes 7 134 162 59 96	Seagrass Relative Risk Index (Anchored) 4 83 100 37 59	Rank 6 2 1 4 3
Seagrass NRM region Cape York Wet Tropics Burdekin Mackay Whitsunday Fitzroy Burnett Mary	Seagrass Marine Risk Index 7 34 100 34 68 18	Loads Index 0 100 62 25 28 20	Sum of Indexes 7 134 162 59 96 38	Seagrass Relative Risk Index (Anchored) 4 83 100 37 59 23	Rank 6 2 1 4 3 5

Table 2.17. Results of the overall risk assessment using a sum of the anchored Indexes for coral reefs and seagrass (final output from Part 3 in Figure 2.3). The Region that had the largest sum of indexes was given a score of 100; all other Regions are expressed as a percentage based on sum of indexes in each Region relative to the sum in the Region with the maximum sum of indexes.

Coral Reef and Seagrass	Coral Reef Relative Risk Index	Seagrass Relative Risk Index	Sum of Indexes	Relative Risk Index (Anchored)	Rank
Cape York	12	4	16	9	6
Wet Tropics	100	83	183	100	1
Burdekin	40	100	140	76	3
Mackay Whitsunday	54	37	91	50	4
Fitzroy	86	59	145	80	2
Burnett Mary	11	23	34	19	5
· ·		Max	183		•



Figure 2.20. Map summarising the results of the Loads Index, Marine Risk Indexes (coral reefs and seagrass) and the Relative Risk Index. The assessment for coral reefs also includes the COTS Influence Index associated with the extent of the influence of regional river discharge to the COTS Initiation Zone; the scores are 100 for the Wet Tropics, and 16 for the Burdekin region. The method for deriving and combining the indexes is provided in Section 2.4.4.

These results show the greatest risk to each habitat in terms of the potential water quality impact from all of the assessment variables in the GBR and end -of -catchment anthropogenic loads of DIN, TSS and PSII herbicide. These are:

- **Coral reefs:** Wet Tropics region, followed by the Fitzroy region. The rank of the remaining regions is the Mackay Whitsunday, Burdekin, Cape York and Burnett Mary region.
- **Seagrass meadows:** Burdekin region, followed by the Wet Tropics. The rank of the remaining regions is the Fitzroy, Mackay Whitsunday, Burnett Mary and Cape York region.
- Coral reefs and seagrass meadows combined: Wet Tropics, followed by the Fitzroy and Burdekin (both ~80% of the relative risk compared to the Wet Tropics). The relative risk to the Mackay Whitsunday region is half the greatest risk, followed by the Burnett Mary and Cape York regions. It must be reiterated that these are relative assessments and therefore indicates that the Burnett Mary and Cape York are low relative to the other regions, but may be exposed to a range of risks that still warrant management to maintain the ecosystem values in these regions.

A number of differences are also evident in cross shelf locations of reefs and seagrass, leading to the following conclusions:

- **Mid shelf reefs south of Cooktown:** DIN due to the link to the initiation of COTS outbreaks, and to a lesser extent, the effects of increased nutrients on bleaching susceptibility.
- Inner shelf reefs south of Cooktown: Primarily TSS, and secondly, DIN and PSII herbicides.
- **Seagrass:** Primarily TSS, and secondly PSII herbicides. The effects of nutrients are not fully understood but there is evidence of changes in seagrass cover linked to excess nutrients.

It is noted that the value of using this type of final assessment where all values are combined in the end can reduce the intrinsic value of each of the multiple datasets and stages of assessment used in this study and without sufficient explanation, may leave the final results subject to misinterpretation. However, it does provide an overall assessment of the relative risk of all water quality in the marine environment in the context of the end of catchment anthropogenic loads which may be useful for managers in prioritising catchments based investments.

2.4.5 Conclusions

The results of the three main parts of the study show some clear patterns, and in many cases distinct differences, between the relative risk of the key pollutants to coral reefs and seagrass meadows among NRM regions. In summary:

- The area of coral reefs at greatest risk from all of the sediment and nutrient variables (except for the COTS Initiation Zone) was highest in the Burdekin and Fitzroy regions. The COTS Initiation Zone straddles the boundary between the Cape York and Wet Tropics regions, and approximately 60% of reefs within the Zone are located in the Cape York marine NRM region.
- The area of seagrass at greatest risk from all of the sediment and nutrient variables was highest in the Burdekin region and second highest in the Wet Tropics region, however, the areas are less than one quarter of the areas affected in the Burdekin region in all cases. The area of seagrass within the highest assessment classes of the nutrient variables is greatest in the Burdekin region, and then the Fitzroy.
- Of the NRM regions examined in the assessment, the Mackay Whitsunday region presents the highest ecological risk of pesticides with the PSII herbicide risk of 'High' and 'Medium' extending off

the mouths of the Pioneer and O'Connell Rivers and Sandy Creek. This is followed by the Burdekin (due to the Barratta Creek and Haughton Rivers but not the Burdekin River itself), Wet Tropics, Fitzroy and Burnett Mary regions. It should be noted that the risk to 'pesticides' here is represented by PSII herbicides as these are the dominant pesticides detected in catchments, however a total of 34 pesticides (herbicides, insecticides and fungicides) have been detected.

- A qualitative assessment of the potential risk of PSII herbicides to other GBR ecosystems showed that a variety of coastal habitats (e.g. wetlands, estuaries, mangroves and seagrass) which provide important ecological services (including nursery habitats, primary productivity and nutrient cycling) to GBR biota are at risk. The assessment classified the risk of pesticides in the GBR into five groups ranging from Very Low to Very High. Within the freshwater reaches of rivers and freshwater/coastal wetlands, the risk of pesticides (PSII herbicides and some non-PSII pesticides) was assessed to be Very High to Medium, particularly for the coastal stream networks that drain a relatively large area (> 20%) of intensive agriculture such as Barratta and Sandy Creeks. In the estuarine reaches of the rivers, the risk of pesticides was mostly within High to Low risk. A similar risk occurs for the coastal marine environment including intertidal and subtidal seagrass meadows. Coral reefs and seagrass meadows on the inner shelf (includes but not limited to areas extending up to 20 km from the coast) generally fell into the lowest risk classes depending on the region and adjacent catchment(s). The risk to coral reefs on the mid and outer shelf was considered Very Low to no risk.
- From a combined assessment of all of the water quality variables the risk was found to be greatest for coral reefs in the Fitzroy and Mackay Whitsunday regions, and for seagrass in the Burdekin and Fitzroy regions. In most cases, the proportion of the habitat area in each region within the High and Very High classes is less than 10%, except in the case of seagrass meadows in the Mackay Whitsunday region where 37% of the area of seagrass in the region is within the Marine Risk Index area. It should be noted that this assessment does not account for the potential synergistic or antagonistic effects that these multiple stressors when acting together may have on ecosystems.
- In many cases, these differences in relative risk to coral reefs and seagrasses between regions can be related to the characteristics of pollutant load inputs to the GBR lagoon. The largest sources of anthropogenic TSS loads is from rivers in the Burdekin region, the greatest contribution to the anthropogenic DIN load is from rivers in the Wet Tropics region, and the greatest contribution to the PSII herbicide load is from rivers in the Wet Tropics region. There are 7 rivers where the average of the summed anthropogenic contributions of TSS, DIN and PSII herbicides loads exceed 5% of the GBR total, ordered from highest to lowest average contribution here: (1) Burdekin River (17% average), (2) Fitzroy River (12%), (3) Johnstone River (12%), (4) Herbert River (10%), Haughton River (7%), Tully River (7%) and the Russell Mulgrave River (6%).
- The combination of the assessment of marine water quality relative risk and anthropogenic end-ofcatchment pollutant loads allows us to draw conclusions about the overall risk of pollutants to the GBR. The ranking of the relative risk of degraded water quality between the NRM regions in the GBR is: Wet Tropics > Fitzroy > Burdekin > Mackay Whitsunday > Burnett Mary > Cape York. Priority areas for management of degraded water quality in the GBR are: Wet Tropics for nitrogen management; Mackay Whitsunday and lower Burdekin for PSII herbicide management; and Burdekin and Fitzroy for suspended sediment management.

While the areas of coral reef and seagrass affected in highest risk categories for individual variables and all of the variables combined are relatively small, they are often located in some of the highly valued tourism and recreation sites of the GBR. In addition, the habitats of the Burnett Mary region are considered to be undervalued in this assessment, as the GBR Marine Park and World Heritage Area boundary does not include all of the habitat areas that would be affected by the catchments of the Burnett Mary region. In particular, there is a large area of seagrass to the south of the boundary in Hervey Bay which is known to provide

important habitat for species that also inhabit the GBR Marine Park that should be incorporated in any future assessments of the influence of water quality on regional habitats.

The final conclusions of the relative risk from water quality to GBR ecosystems are presented in Part B of this report drawing on the results of this assessment and the Supporting Studies (compiled as a separate report).

Part B: Overall conclusions of the relative risk between sediments, nutrients and pesticides and between land uses, industries and catchments in the GBR

3 Relative risk from degraded water quality to GBR ecosystems

This section summarises the outcomes of the risk assessment using additional evidence from the supporting studies to draw conclusions about the relative risk of water quality to GBR ecosystems.

3.1 Additional information to support the assessment in Part A

Several limitations to the quantitative assessment (Part A of this report) are identified in Section 4; however, a number of these can be overcome by incorporation of new knowledge in a qualitative way to make conclusions about the relative risk of degraded water quality to the GBR. The Supporting Studies have informed the selection of variables and methods of analysis used in the risk assessment described in Section 2. The Supporting Studies also strengthen our understanding of the consequences of pollutant impacts on coral reefs and seagrasses. In particular: Chapter 1 emphasises the importance of nutrients in the initiation of COTS and therefore loss of coral cover; Chapter 2 confirms the relative importance between nitrogen and phosphorus in driving productivity in the GBR lagoon; Chapter 3 reviews the effects of sediments and sedimentation on coral reef communities; Chapter 4 provides new information on the relative risk of PSII herbicides to coastal and marine ecosystems, identifying the highest risk areas to be in freshwater and coastal wetlands, estuarine areas and coastal seagrass and inshore reef communities; Chapter 5 develops relationships between flood plume frequency and end of catchment pollutant loadings to define plume water types related to water quality characteristics; and Chapter 6 provides a case study that shows a correlation between plume water types and seagrass cover with a supporting review on the impacts of water quality on seagrass in Chapter 7. A small amount of resources were also provided to analyse water quality and phytoplankton populations in flood conditions, provided as Chapter 8.

Supplementary evidence that is important the conclusions of our assessment are also included below.

Seagrass meadows and dugong and green turtle populations

The links between the condition and extent of seagrass meadows in the GBR has important implications for populations of dugongs and green turtles in the GBR, and consideration of this relationship in this assessment is therefore critical. Loss of seagrass habitat as a result of severe weather events and degraded water quality has led to increased mortality of dugong and turtles in recent years. Cyclone Yasi in the central GBR and flooding in the southern GBR in 2011 resulted in devastating loss of seagrass (Coles et al. 2011; McKenzie and Unsworth, 2011; McKenzie et al. 2012) along the GBR coast from Hervey Bay to Cairns. This loss came on top of declining seagrass health in the central GBR (McKenzie et al. 2010a). The dugong population in the GBR is totally reliant on seagrass communities. Evidence shows that the southern dugong population was significantly reduced by commercial harvesting between 1847 and 1969 so the population is at best only about 25% what might be expected (Marsh et al. 2005). Given this population (reduced though it is) is a potential surrogate for the quantity of seagrass needed to maintain it, the loss of seagrass from the chronic impacts of water quality followed by the acute impact of the extreme cyclonic event and flooding in 2011 saw the southern dugong population reduce from an estimate of 2500 animals in 2005 to 600 in November 2011 (Sobtzick et al. 2012). It is considered that much of the change in population estimates between 2005 and 2011 can be explained by animals moving to locations outside the survey area to search for seagrasses (Sobtzick et al. 2012).

The number of recorded dugong and turtle strandings has also increased significantly in recent years. In 2011 there were over 180 recorded deaths of dugong on the Queensland coast (Meager and Limpus, 2012) believed to be due mainly to starvation associated with the loss of seagrass (Bell and Ariel, 2011). As shown in Figure 3.1, turtle deaths (mostly green turtles) on the Queensland coast nearly doubled between 2010 and 2011 (Meager and Limpus, 2012) have been attributed to a range of complications resulting from a lack of

food. An assessment of the effects of the 2010-2011 flood events in the GBR show that the dugong deaths were potentially increasing both because of the chronic loss of seagrass (2009 and 2010) and increased dramatically in 2011 (GBRMPA, 2011).

This evidence, coupled with current knowledge of the impacts of degraded water quality on seagrass meadows in the GBR (see above and Chapter 7 of the Supporting Studies), strengthens the importance of the implications of increased suspended sediment discharge from land based sources to the GBR.

GBR catchment land use

Major land uses in the Great Barrier Reef region are shown in Figure 3.2. Grazing is the dominant agricultural land use. Sugarcane and horticultural crops dominate in the high rainfall and coastal irrigation areas. Large sections of summer and winter rain-fed grain crops and irrigated cotton are prevalent in the inland areas of the Fitzroy region.

To relate the results of the Relative Risk Index to catchment management priorities, reference can be made to current knowledge of the pollutant contributions from different land uses in each Region. The latest information on pollutant sources in the GBR catchments is synthesised in SCS Chapter 4 *Sources of sediment, nutrients, pesticides and other pollutants in the Great Barrier Reef catchment* (Kroon et al. 2013). In summary:

- Grazing lands contribute 75% of the total TSS load. Land use, soil properties, the extensive area involved, river discharge flows and geomorphology all contribute to the loads. Provenance tracing shows that most sediment comes from a combination of gully and streambank erosion and subsoil erosion from hillslope rilling, rather than broad-scale hillslope sheetwash erosion.
- Eighty percent of DIN originates from the Wet Tropics, Burdekin and Mackay Whitsunday regions, primarily from fertilised land use and in particular, land used for sugarcane cultivation.
- Results from Source Catchments modelling suggest that over 90% of the modelled PSII herbicide load is from land used for sugarcane cultivation, with minor contributions from cropping and grazing lands, in particular from the Fitzroy catchment (Waters et al. in press).

(a) Dugong



(b) Turtle



Figure 3.1. Monthly cumulative strandings by year for the Queensland East Coast, up to 31 January, 2013 for (a) dugong and (b) turtle (only cases confirmed in the field by a trained person, and later verified by an expert are included). Source: Meager and Limpus, 2012.



Figure 3.2. Land use map of the Great Barrier Reef catchment. Source: Waterhouse et al. (2012).

3.2 Overall assessment of the relative risk of degraded water quality to GBR ecosystems

Using the information from Part A and the Supporting Studies we have developed an overall summary of the relative importance of different pollutants to GBR ecosystems, and the relative risk of degraded water quality to GBR ecosystems among NRM regions. This information is presented in Table 3.1, and illustrated in Figure 3.3. In this table we have taken the results of the individual assessments of each variable in the ecological risk assessment to highlight where the variables dominate among the Regions. The primary rank for each variable is listed, with an indication of whether it dominates in terms of coral reef or seagrass area, or both. The Marine Risk Index for coral reefs and seagrass is then shown for the results of the combined analysis of water quality variables. The regional anthropogenic loads as a proportion of the total GBR load for TSS, DIN and PSII herbicides are shown to identify the primary sources of anthropogenic loads delivered to the GBR, in addition to the Loads Index which combines this information. Additional information includes facts related to the influence of river discharges to the COTS Initiation Zone and additional pesticide information. The Relative Risk Index which represents the overall results of the risk assessment in Part A is shown as the critical underpinning for the overall conclusions. This is then informed by the management issues and associated land uses which were derived from published evidence and expert judgement of the assessment team (informed by the preceding columns). The overall ranking of relative risk was developed by the assessment on the basis of the overall content of the table. Importantly, while the Burnett Mary ranked relatively low in the Relative Risk Index, this result is considered to be highly uncertain due to the fact that most reefs and seagrass meadows in this region (but outside of the GBRWHA) were not included formally in the analysis.

The information has also been coupled with the summary of pollutant sources from land uses in each region (above) to generate the management priorities identified Table 3.2. However, cost effective solutions for all of these management issues are not necessarily currently available for all of these priorities. These issues are discussed further in SCS *Chapter 5* regarding the effectiveness of management practices and outstanding information gaps in this field (Thorburn et al. 2013).

Even though the nutrient related variables of Chlorophyll threshold exceedance and DIN plume loading were ranked highest in the Fitzroy region, there is insufficient knowledge of the sources of dissolved inorganic nitrogen in the Region to make recommendations about management priorities for these. Further knowledge of the role of particulate nitrogen, which is largely derived from grazing lands, and the processing of this into dissolved inorganic nitrogen is important for making future management recommendations in the large grazing catchments of the Fitzroy region. Current research is showing differences in sediment and nutrient runoff from native brigalow scrub, newly planted legume based ley pastures and cropping lands which will be useful to guide management priorities in the future. Compared to the native brigalow scrub landscape, cropping exported more TSS and DIN, while grazing exported less total nitrogen and DIN, but more TSS (Thornton and Elledge, 2012). The legume pastures do appear to pose a risk to water quality as they contribute higher nutrient loads than grass only pasture systems, established grass-leucaena pastures, and the native brigalow scrub landscape representative of the environment in its pre-European condition (Elledge and Thornton, 2012).

In the Burnett Mary region all of the variables ranked relatively low, however, the assessment does not include large areas south of the GBRWHA boundary that contain coral reefs and large areas of seagrass (refer to Section 4). The end of catchment TSS total and anthropogenic annual average loads in the Mary River are the fourth highest of all GBR catchments. A majority of this is thought to be derived from grazing lands, although mixed cropping is also a potential contributor (Waters et al. in press). The TSS loads from the Burnett catchment and particularly the role of dams in sediment trapping requires further investigation. It is possible that a similar process of trapping of coarser sediments (>20 μ M) that occurs in the Burdekin catchment (Lewis et al. 2013b) may also occur in the Burnett catchment, resulting in the bulk of the material discharged to the marine environment being the finer fraction. It is the finer sediment fraction that is more harmful to coral reefs (Weber et al. 2006, 2012) and seagrass (Collier et al. 2012). For these reasons sediment management in the Burnett Mary region is considered to be highest priority until a more thorough analysis is undertaken that includes all of the potential ecosystems at risk.

Table 3.1. Summary of the outcomes of the overall assessment of the relative risk of water quality in the GBR. Note that the Burnett Mary Region is shaded in grey to represent the fact that most reefs and seagrass meadows in this region were not included formally in the analysis and thus the validity of the result has high uncertainty.

Region	Dominant variables in marine assessment Variables where max area is in Region	Marine	Risk Index	Anthr as a the T	Regiona opogen proport otal GB (%)	al ic Load ion of R Load	Loads Index	Additional Factors	Relative Risk Index	Management Issues	Associated land uses	Overall Ranking of Relative Risk
	CR = Coral Reef SG = Seagrass	Coral Reef	Seagrass	TSS	DIN	PSII Herb						
Cape York	COTS Initiation Zone (CR)	12	4	3	<1	<1	0	Influence from catchment runoff is predominantly from Wet Tropics Rivers	9	The data in this uncertain due t of marine datas	Region are highly o limited validation ets.	LOW
Wet Tropics		100	83	9	20	61	100	86% volumetric contribution to COTS Initiation Zone	100	Nutrients Pesticides	Sugarcane Bananas	VERY HIGH
Burdekin	TSS 2mg/L (SG, CR) TSS 7mg/L (SG) TSS Plume loading (CR, SG) Chl 0.45μg/L (SG) DIN Plume loading (SG)	40	100	32	11	13	62	14% volumetric contribution to COTS Initiation Zone High risk from PSII herbicides to Ramsar listed freshwater wetlands in the lower Burdekin catchments	76	Sediments Nutrients Pesticides	Grazing Sugarcane (coastal)	HIGH
Mackay Whitsunday	Pesticide exposure (CR, SG)	54	37	4	6	12	25	High risk from PSII herbicides in Sandy Creek	50	Pesticides Nutrients	Sugarcane	MODERATE
Fitzroy	TSS 7mg/L (CR) Chl 0.45μg/L (CR) DIN Plume loading (CR)	86	59	17	5	4	28	Monitored loads of PSII herbicides were high in 2011 (not reflected in modelled baseline)	80	Sediments Pesticides Nutrients ¹	Grazing Cropping	HIGH
Burnett Mary	All variables rank relatively low	11	23	4	4	9	20	The Mary River has the fourth highest total and anthropogenic TSS load of all GBR catchments	19	Sediments	Grazing	UNCERTAIN
										All variables rar however, there in this result giv on the full exter corals and seag outside the GBF this assessment	k relatively low, is high uncertainty ren the lack of data nt and condition of rass (which are RWHA) available for	

¹There is insufficient knowledge of the sources of DIN in the Fitzroy region to make recommendations about management priorities for these. Further knowledge of the role of particulate nitrogen, which is largely derived from grazing lands, and the processing of this into DIN is important for making future management recommendations in the large grazing catchments of the Fitzroy region.



Figure 3.3. Illustration of the overall outcomes of the assessment of the relative risk of degraded water quality to Great Barrier Reef coral reefs and seagrass. The map shows the dominant land uses and priority pollutants and results of the overall relative risk ranking in each NRM region.

Relative		Management Priori	ities			
Priority	Region	Pollutant Management	Key land uses	Comments		
1	Wet Tropics	Fertiliser nitrogen reduction	Sugarcane, Bananas			
	Burdekin	Erosion management in Burdekin	Grazing	•		
	Fitzroy	Erosion management in Fitzroy	Grazing, Cropping			
2 Burdekin		Pesticide reduction in (lower) Burdekin and Haughton	Sugarcane	Note that these actions should not be		
	Mackay Whitsunday	Pesticide reduction in all catchments	Sugarcane	prioritised at the exclusion of other practices that are		
	Burdekin	Fertiliser nitrogen reduction in (lower) Burdekin and Haughton	Sugarcane	already in place to manage losses of		
3	Mackay Whitsunday	Fertiliser nitrogen reduction	Sugarcane	Regions		
	Burnett Mary	Erosion management in all catchments	Grazing			
	Wet Tropics	Pesticide reduction in all catchments	Sugarcane			
	Fitzroy	Pesticide reduction in all catchments	Grazing, Cropping			
4	Burnett Mary	Further information is required to in assessment, including data on the full condition of corals and seagrass (which a GBRWHA) in the Region.	nform the l extent and are outside the	Habitat mapping, ecological value assessment and monitoring of ecosystem condition is required		
5	Cape York	Further information is required to und influences.	erstand local	As a relatively low impacted area, management efforts should aim to maintain the current values of the region		

Table 3.2. Summary of management priorities for reducing the relative risk of degraded water quality to the GBR.

3.3 Overall conclusions

Drawing on the information in this report, the overarching conclusion drawn from this project is that **the** greatest water quality risks to the GBR are from nitrogen discharge, associated with crown of thorns starfish outbreaks and their destructive effects on coral reefs, and fine sediment discharge which drives light reduction for seagrass ecosystems and inshore coral reefs. Pesticide inputs pose a risk to freshwater and some inshore and coastal habitats.

The key points of supporting evidence for this statement are included below. These statements have been included in the *Reef Plan SCS Chapter 3 Relative risks to the GBR from degraded water quality* (Brodie et al. 2013b).

- Overall, increased concentrations of nitrogen from catchments between the Daintree and Burdekin Rivers pose the greatest risk to coral reefs. Runoff from these rivers during extreme and early wet seasons is associated with outbreak cycles of the coral-eating COTS on the northern GBR shelf (15 to 17°S) that subsequently generate secondary outbreaks throughout the central and southern GBR. GBR-wide loss of coral cover due to COTS is estimated to be 1.4% per year over the last 25 years, and a new outbreak is underway. It is estimated that COTS have affected >1000 of the ~3000 reefs within the GBR over the last 60 years.
- Of equal importance is the risk to seagrass meadows from suspended sediments discharged from rivers in excess of natural erosion rates, especially the fine fractions (clays). Whether carried in flood plumes, or resuspended by wave action, suspended particulate matter create a turbid water column that reduces the light required by seagrass and corals. High turbidity affects ~200 inshore reefs and most seagrass areas. Seagrass loss severely impacts green turtle and dugong populations. On a regional basis the Burdekin and Fitzroy regions present the greatest risk to the GBR in terms of sediment loads.
- Loss of seagrass habitat as a result of cyclones, floods and degraded water quality appears to be associated with higher mortality of dugong and turtles.
- At smaller scales, particularly in coastal seagrass habitats and freshwater and estuarine wetlands, pesticides can pose a high risk. Concentrations of a range of pesticides exceed water quality guidelines in many fresh and estuarine waterbodies downstream of cropping lands. Based on a risk assessment of the six commonly used PSII herbicides, the Mackay Whitsunday and Burdekin region are considered to be at highest risk, followed by the Wet Tropics, Fitzroy and Burnett Mary regions. However, the risk of only a fraction of pesticides has been assessed, with only 6 out of the 34 pesticides currently detected included in the assessment, and therefore the effect of pesticides is most likely to have been underestimated.
- The ranking of the relative risk of degraded water quality between the NRM regions in the GBR is: Wet Tropics > Fitzroy > Burdekin > Mackay Whitsunday > Burnett Mary > Cape York. Priority areas for management of degraded water quality in the GBR are: Wet Tropics for nitrogen management; Mackay Whitsunday and lower Burdekin for PSII herbicide management; and Burdekin and Fitzroy for suspended sediment management.
- From a combined assessment of water quality variables in the GBR (using the total area of habitat affected in the highest relative risk areas) and end-of-catchment anthropogenic loads of nutrients, sediments and PSII herbicides, the regional ranking of water quality risk to coral reefs is: Wet Tropics > Fitzroy > Mackay Whitsunday > Burdekin> Cape York > Burnett Mary. The regional ranking of water quality risk to seagrass is: Burdekin > Wet Tropics > Fitzroy > Mackay Whitsunday > Burdekin > Wet Tropics > Fitzroy > Mackay Whitsunday > Burnett Mary > Cape York. Importantly in the Mackay Whitsunday Region, 40% of the seagrass area is in the highest relative risk class compared to less than 10% for all other regions. However the highly valuable

seagrass meadows in Hervey Bay, and the importance to associated dugong and turtle populations in the Burnett Mary Region, were not included in the ranking analysis.

- Runoff-associated risk varies with distance from rivers. Coastal reefs near river mouths will be most adapted to poor marine water quality especially highly turbid conditions.
- Enhanced sedimentation interferes with the normal functioning of benthic animal and plant communities. Effects include reducing free surfaces available for larval recruitment, reductions of early life stage survival, and slowed growth and reproduction by increasing energy expenditure on cleaning.
- Corals in areas of high nutrient concentrations are more susceptible to temperature-induced bleaching due to a lower temperature threshold (less than 2°C change) for bleaching.
- Macroalgal cover on reefs is inversely correlated with indices of water quality, increasing 5-fold with decreasing water clarity and 1.4-fold with increasing chlorophyll (nutrient availability). Enhanced macro-algal cover can affect coral cover and/or larval coral recruitment.
- Dissolved inorganic and particulate forms of nutrients discharged into the GBR are both important in driving ecological effects but increased nitrogen inputs are more important than phosphorus inputs. Dissolved inorganic forms of nitrogen and phosphorus are considered to be of greatest concern compared to dissolved organic and particulate forms of nutrients, as they are immediately and completely bioavailable for algal growth. Particulate forms mostly become bioavailable over longer time frames, and dissolved organic forms typically have limited and delayed bioavailability.
- Little is known about the types and concentrations of contaminants bound to sediment discharged by rivers into the GBR and the risk that these pose to GBR ecosystems.
- Other pollutants (e.g. microplastics, endocrine disrupting substances, oil and PAHs, pharmaceuticals and heavy metals) may pose risks to specific coastal sites or ecosystems, but our limited current knowledge of the extent and degree of their influence precludes a quantitative statement of risk to wider areas.

The risk assessment described in this report provides the best available assessment of the relative risk of degraded water quality to the GBR and can be used as the first step in prioritising management focused on regional 'hot spots' for pollutant sources, contributing industries and resulting impacts in the marine environment. It is recognised that there are several limitations to the assessment that are important to identify and priorities for future work; these are outlined in Section 4.

4 Limitations to the risk assessment and future improvements

The risk assessment described in this report provides the best available assessment of the relative risk of water quality pollutants to the GBR and the information outlined above can be used as the first step in prioritising management based on regional 'hot spots' for pollutant sources, contributing industries and resulting impacts in the marine environment. However, there are several limitations to the assessment that are important to identify, and are summarised below.

Limitations to the input datasets in terms data collection, temporal and spatial resolution, influence the certainty of the outcomes. Several examples can be presented here:

TSS and chlorophyll exceedance is based on daily observations over a 10 year monitoring period (with only 1 or 2 valid observations every 5 days), while TSS and DIN plume loading is based on a mean of 2007 to 2011 (which were in fact relatively wet years in the long term record), and PSII herbicide exposure is based on single flood events. For these reasons the final conclusions of the assessment are supported by additional evidence of known water quality conditions, spatial and temporal patterns and ecological impacts. Additional variables that were considered but not included due to the current lack of temporal and spatial data, and / or knowledge of ecological impacts include chronic exposure to PSII herbicides and non-PSII herbicides, particulate nutrients and phosphorus exposure, and micropollutants presence and distribution in the GBR.

The modelled estimates of anthropogenic end-of-catchment loads are long term averages and do not capture the influence of large floods. Empirical datasets included in the assessment (eg. TSS and DIN surface exposure) do factor in these events. In comparing the modelled results against empirical data, the relative contributions of individual NRM regions are in general agreement with monitoring data except during extreme wet seasons.

No quantitative assessment of uncertainty in the estimates of risk. For all variables, any relative differences in uncertainty and hence our confidence in the data can only be assessed highly subjectively. If such qualitative assessments of uncertainty in our methodologies and data were undertaken, uncertainty would be assessed as varying as much within as among NRM regions. As we compare results for NRM regions in the final combined relative risk assessment, the various methodologies used to generate the data are considered to have roughly the same uncertainty and with the limited time and resources, no specific estimates were considered.

The assessed risk posed by pesticides is most probably an underestimate. Only a few of the pesticides detected in the GBR lagoon are considered. The risk posed by multiple pesticides, in combination with other contaminants found in flood plumes (e.g. elevated TSS and nutrients) and other environmental stressors (temperature) have not been assessed. Cumulative impacts from the multiple plumes that occur each year are also not accounted for. Toxicity of PSII herbicides is time dependent (Vallotton et al. 2008), i.e. the toxicity to phototrophs increases with exposure duration. For this risk assessment, only acute exposure was used to assess the potential impacts to seagrass and corals.

Risk has not been assessed equally across all regions due to limitations with habitat mapping and assessment boundaries. For example, there is less data on the extent and condition of corals and seagrasses of the Burnett Mary region which are outside the GBRWHA.

The wider Burnett Mary region contains a large area of seagrass to the south of the GBRWHA boundary in Hervey Bay which provides important habitat and foraging grounds for mobile species that also inhabit the GBRWHA. The total estimate of seagrass in the Burnett Mary marine NRM regions is 8,000km² (McKenzie et al. 2010b). The seagrass area in Hervey Bay supports the highest density dugong habitat south of the Torres Strait (Grech et al. 2011; see Figure 4.1). These seagrass meadows have been severely impacted by several high discharge events from the rivers in the Burnett Mary region in 1992 (Preen et al. 1994), again in 1999 (Campbell and McKenzie, 2004) and in 2011 (McKenzie and Unsworth, 2011; Coles et al. 2011). The loss of seagrass after these floods had dramatic effects on dugong mortality and migration (Preen et al. 1994).

Mon Repos near Bundaberg supports the largest concentration of nesting marine turtles on the eastern Australian mainland. This is the most significant loggerhead turtle nesting population in the South Pacific Ocean region. Successful breeding here is critical for the survival of this endangered species. The Eastern Australia loggerhead turtle population is recognised as a single genetic stock, with most females nesting at one of five rookeries (Mon Repos, Wreck Rock, Wreck Island, Erskine Island and Tryon Island). Among these, Mon Repos currently supports the biggest nesting population, with ~300 females nesting per year (Limpus et al. 1994; Limpus and Limpus 2003a,b). The Mon Repos nesting site was badly damaged in the Burnett River floods of 2011.

There are also important areas of coastal coral reef in the area south of the GBRWHA boundary (Zann, 2012) that has been impacted by river discharge events from Burnett Mary rivers. Finally, pesticides discharged from the Mary River have been found in estuarine and marine sections of Hervey Bay at concentrations potentially able to reduce photosynthesis in seagrass (McMahon et al. 2005). Given these considerations the conclusions of relative risk between regions is likely to be an underestimate for the Burnett Mary. These areas should be incorporated in any future assessments of the influence of water quality on regional habitats.

Coral communities in Hervey Bay and Great Sandy Strait were recently mapped by Zann (2012) shown in Figure 4.2. In addition, more detailed surveys of coral reefs in the coastal areas around Bundaberg were recently been completed by Central Queensland University (Alquezar et al. 2011) with support from the Burnett Mary Regional Group. The Woongarra Coast, located approximately 10km east of Bundaberg, is the most southerly set of coastal fringing reefs on the East Australian mainland and boasts some of the best coastal reef diving in Queensland. The reef section spans from Burnett Heads in the north to Elliott Heads in the south (~25km). The surveys identified a total of 34 substrate classes, in a range of biodiversity and coral condition. Coral communities at eight sites in Hervey Bay were also surveyed for the Wildlife Preservation Society of Queensland, Fraser Coast Branch in 2010 (De Vantier, 2010). The sites were located adjacent to the mainland coast from Gatakers Bay to Point Vernon, and off Woody and Little Woody Islands in the Great Sandy Strait. A total of 46 species of reef-building (hermatypic) corals were recorded, with the richest individual site at Woody Island (31 species). These areas should be incorporated in any future assessments of the influence of water quality on regional habitats. A similar argument could be made for the coral reef and seagrass habitats of the northern GBR in the Torres Strait.


Figure 4.1. Dugong relative population density along the Queensland coast. Source: Grech et al. (2011).

The areas of greatest density are in the Torres Strait in the north, the far northern coast of the Great Barrier Reef, and the southern areas of Hervey Bay which are the highest density south of the Torres Strait. Derived from spatially-explicit population models of dugong distribution and relative density in northeast Australia. The spatially-explicit models were interpolated from a 20-year time series of systematic aerial surveys of dugong at the scale of 2 km * 2 km planning units. Planning units were initially classified as low, medium, high and very high dugong density on the basis of the relative density of dugongs estimated from the models and a frequency analysis. The model of dugong distribution and relative density in the southern Great Barrier Reef region is from Grech and Marsh (2007).



Figure 4.2. Seagrass and coral reef mapping in Hervey Bay and Great Sandy Straits. Source: Zann (2012). Note that the GBRWHA is just northward of the northern boundary of this map.

Estimates of regional river influences on the GBR are assumed to be constrained within adjacent marine NRM boundaries due to currently limited capacity to quantify the transport and dispersion of individual river plumes in the GBR lagoon. It is fully understood that river plumes, particularly from the large Burdekin and Fitzroy River catchments can cover large areas within the GBR lagoon. Observations (e.g. Wolanski and van Senden, 1983; hydrodynamic modelling (e.g. King et al. 2002) and satellite imagery (e.g. Devlin et al. 2012) during periods of high flow has shown that the Burdekin River influences the shelf into the Wet Tropics marine NRM region, and that the Fitzroy River plume may reach the Mackay Whitsunday marine NRM region.

The risk classes for individual water quality variables are not equivalent in terms of ecological impact, and are therefore not directly comparable without recognition of these differences. Further studies should adequately address this limitation to provide a better representation of the severity of potential ecological impacts between assessment classes for each water quality variable. The approach to classification used is also a potential weakness of MCA, which is an interval scale approach, while risk consequence is inherently oriented to a need for quantification of magnitudes. In addition, the assessment does not account for the potential synergistic or antagonistic effects that these multiple stressors when acting together may have on ecosystems.

Only a limited sensitivity analysis that tested weighting of variables has been conducted. More scenarios that scale or 'weight' individual factors or pollutants as being more or less important and the effect of only selecting the highest assessment classes in the final analysis should be tested. For example, a more detailed assessment of the patterns in the lower assessment classes should be considered in future work, particularly given the potential influence of chronic exposure to pollutants, or the effects of periodic exposure to high concentrations of pollutants.

Further validation of remote sensing-based results is required for locations where high turbidity that confounds existing algorithms may naturally occur. These areas include the Cape York region north of Cooktown, coastal areas around Shoalwater Bay which are naturally turbid. Uncertainties in products derived from remote sensing of these areas have not been resolved. In addition, the number of valid observations for the remote sensing assessment varies between seasons and locations and over the year equates to an average of less than 2 valid observations every 5 days.

The scope of the assessment is limited in terms of the coverage of social and economic issues. It should be recognised and highlighted that the results presented in this study only represent the biophysical perspective of management priorities required to reduce pollutant impacts on the GBR. However, further consideration of the relative priorities between the Regions and industries requires incorporation of the current adoption of management practices, the feasibility of adopting the most effective practices in terms of water quality benefits, the relative cost effectiveness of these practices, existing management programs in place, and the range of management strategies available to address these issues. The Reef Plan 3 Management Prioritisation project will address these aspects to some degree over the coming months, although these aspects will always present a challenge to managers due to the complexity of the issues and varying degrees of knowledge of these aspects between the Regions and industries.

These limitations have been translated into priority information needs for future risk assessments of water quality in the GBR:

 Scoping of the availability of, and acquisition of, more consistent temporal and spatial data for all water quality variables (including those not included in the most recent assessment such as phosphorus and particulate nutrients) and their ecological impacts to enable improved classification in terms of ecological risk and application of a formal risk assessment framework (which includes assessments of likelihood and consequence).

- 2. Better understanding of the responses of key GBR ecosystem components to cumulative impacts of repeated exposure to poor water quality, and the cumulative impacts of multiple water quality pressures.
- 3. Definition of zones of river influence in the GBR for each catchment using hydrodynamic and pollution distribution models so that water quality risk from individual and combined pollutants can be attributed back to individual rivers. With this information it would be possible to estimate source-sink relationships for every pollutant, every river, and every part of the GBR lagoon. It is the ultimate intent of eReefs to deliver this type of information.
- 4. Collation of all existing information on the distribution and condition of coral reef and seagrass habitats in the Burnett Mary and Cape York regions, with support for resources to undertake further assessment if gaps are identified.
- 5. Validation of the remote sensing data for turbidity and chlorophyll, particularly in areas which are known to be naturally highly turbid or where existing validation data is limited such as in Cape York and Burnett Mary regions.
- 6. Better understanding of the prevalence and associated effects of other pollutants (e.g. microplastics, endocrine disrupting substances, oil and PAHs, pharmaceuticals and heavy metals) on GBR ecosystems.
- 7. Extending the habitat assessments beyond coral reefs and seagrass to include coastal ecosystems such as freshwater and coastal wetlands, mangroves and estuarine environments, and non-reef bioregions.

If a number of these limitations were overcome, it would be more feasible to undertake a risk assessment that enabled analysis of the absolute risk of degraded water quality to GBR ecosystems. The Ecological Risk Assessment framework provided in Hayes et al. (2012) provides a solid foundation for this improved approach. In the proposed methodology, Tiers 1-4 were designed to deal with increasingly higher quality data regarding exposure to, and effects of, pollutants. Importantly, the lower tiers would be able to handle any data so long as they were time stamped and location stamped and would not be reliant on data collected over long periods of time with high-frequency. In contrast, Tier 5 would require high quality time series data recorded at locations across the GBR lagoon, and a hind-casting technique to attribute risk back to individual rivers, in order to accurately characterise the relative risk of different rivers. The higher data requirements for the Tier 5 level of the risk assessment was the rationale for proposing a tiered risk assessment approach.

As described in Hayes et al. (2012) the proposed methodology also allows for two sets of loss functions to measure the consequences of exposure to pollutants. The first is a simple indicator function that identifies the probability of exceeding GBR water quality guidelines for each of the pollutants. The water quality guidelines provide environmental standards for water quality stressors that define specific ecological conditions that are protective for a variety of ecological end-points including, but not limited to, specific endpoints such as corals or seagrass. The second set of loss functions involves exposure-effect curves that describe the specific response of the risk assessment endpoint(s) to exposure to one or more pollutants, either individually or in concert. Application of the ERA methodology with the specific exposure effect curves provides for a potentially more accurate assessment but this is contingent on the availability of the more complex loss function for specific endpoints. The tiers are increasingly complex and are contingent upon increasing amounts of data and analysis. The extent to which pollutants will be progressed through each tier depends therefore on the availability of exposure and effects information.

Despite the limitations of the current assessment, the risk analysis described in this Section is considered to be the most quantitative and rigorous, as well as the most collaborative, ever undertaken for assessing the risk of degraded water quality to GBR ecosystems. This is associated with the availability of high resolution spatial data and more in depth analysis of the combination of water quality variables. As a logical next step, current knowledge of the effectiveness of management practices (in terms of water quality benefits and socio economic outcomes) across the main agricultural industries in the GBR catchments can be used to inform investment options to address these water quality priorities. The information presented in Reef Plan SCS on management practice effectiveness (Thorburn et al. 2013) provides the basis for this type of assessment.

5 References

- AS/NZS, 2004. Risk management. Joint Australian/New Zealand Standard prepared by Joint Technical Committee OB-007, Risk Management. AS/NZS 4360:2004.
- Abal, E.G., Dennison, W.C., 1996. Seagrass depth range and water quality in southern Moreton Bay, Queensland, Australia. Marine and Freshwater Research 47, 763-771.
- Alongi, D.M., McKinnon, A.D., 2005. The cycling and fate of terrestrially-derived sediments and nutrients in the coastal zone of the Great Barrier Reef shelf. Marine Pollution Bulletin 51, 239-252.
- Alongi, D.M., Pfitzner, J., Trott, L.A., 2006. Deposition of carbon and nitrogen in carbonate mud of the lagoon of Arlington and Sudbury Reefs, Great Barrier Reef. Coral Reefs 25, 123-143.
- Alongi, D.M., L.A. Trott, Pfitzner, J., 2007a. Biogeochemistry of the inter-reef sediments on the northern and central Great Barrier Reef. Coral Reefs 27, 407-420.
- Alongi, D.M., Trott, L.A., Pfitzner, J. 2007b. Deposition, mineralization and storage of carbon and nitrogen in sediments of the far northern and northern Great Barrier Reef. Continental Shelf Research 27, 2595-2622.
- Alquezar, R., Scannell, J., Boyd, W., 2011. Coastal fringing reefs of the Burnett Mary Region. A report to the Burnett Mary Regional Group. Centre for Environmental Management, CQUniversity Australia, Gladstone Queensland.
- Álvarez-Romero, J.G., Devlin, M., Teixeira da Silva, E., Petus, C., Ban, N.C., Pressey, R.L., Kool, J., Roberts, J.J., Cerdeira-Estrada, S., Wenger, A.S., Brodie, J., 2013. A novel approach to model exposure of coastalmarine ecosystems to riverine flood plumes based on remote sensing techniques. Journal of Environmental Management 119, 194-207.
- Andersen, L. E., 2004. Imposex: a biological effect of TBT contamination in Port Curtis, Queensland. Australian Journal of Ecotoxicology 10, 105-113.
- Anthony, K., Ridd, P.V., Orpin, A.R., Lough, J., 2004. Temporal variation of light availability in coastal benthic habitats: Effects of clouds, turbidity, and tides. Limnology and Oceanography 49, 2201–2211.
- ANZECC and ARMCANZ, 2000. Australian and New Zealand Guidelines for Fresh and Marine Water Quality. Australian and New Zealand Environmental and Conservation Council, Agriculture and Resource Management Council of Australia and New Zealand: Canberra.
- Babcock R, Davies P (1991) Effects of sedimentation on settlement of Acropora–Millepora. Coral Reefs 9(4), 205–208.
- Bainbridge, Z., Wolanski, E., Alvarez-Romero, J., Lewis, S., Brodie, J., 2012. Fine sediment and nutrient dynamics related to particle size and floc formation in a Burdekin River flood plume, Australia. Mar Pollut Bull 65, 236–248
- Bainbridge, Z., Lewis, S., Kuhnert, P., Henderson, B., Smithers, S., Brodie, J. In Review. Sources, transport and delivery of suspended sediment in a large, dry tropical river basin. Water Resources Research.
- Bartley, B., Bainbridge, Z., Lewis, S., Kroon, F., Brodie, J., Wilkinson, S., Silburn, M. In review. Linking sediment impacts on coral reefs to catchment sources processes and management. Science of the Total Environment.
- Bannister, R.J., Battershill, C.N. De Nys, R., 2012. Suspended sediment grain size and mineralogy across the continental shelf of the Great Barrier Reef: Impacts on the physiology of a coral reef sponge. Continental Shelf Research 32, 86-95.
- Bell, I., Ariel, E., 2011. Dietary shift in green turtles. Seagrass-Watch News 44: 2-5.
- Bengtson Nash, S.M., McMahon, K., Eaglesham, G., Muller, J.F., 2005. Application of a novel phytotoxicity assay for the detection of herbicides in Hervey Bay and the Great Sandy Straits. Marine Pollution Bulletin 51, 351-360.

- Berry, K.L.E., O'Brien, D., Burns, K.A., Brodie, J., 2013. Unrecognised Pollutant Risks to the Great Barrier Reef. Centre for Tropical Water and Aquatic Ecosystem Research (TropWATER) Publication, James Cook University, Townsville, 38 pp.
- Birkeland, C., 1982. Terrestrial runoff as a cause of outbreaks of Acanthaster planci (Echinodermata: Asteroidea). Marine Biology 69, 175-185.
- Birrell, C. L., Mccook, L.J., Willis., B.L., 2005. Effects of algal turfs and sediment on coral settlement. Marine Pollution Bulletin 51, 408-414.
- Bostock, H.C., Brooke, B.P., Ryan, D.A., Hancock, G., Pietsch, T., Packett, R., Harle, K., 2007. Holocene and modern sediment storage in the subtropical macrotidal Fitzroy River estuary, Southeast Queensland, Australia. Sedimentary Geology 201(3-4), 321-340.
- Brando, V.E., Dekker, A.G. Schroeder, T., Park, Y.J. Clementson, L.A., Steven, A., Blondeau-Patissier, D., 2008. Satellite retrieval of chlorophyll CDOM and NAP in optically complex waters using a semi-analytical inversion based on specific inherent optical properties. A case study for Great Barrier Reef coastal waters. Ocean Optics XIX, Barga, Italy.
- Brando, V.E., Schroeder, T., Dekker, A.G., 2010. Reef Rescue Marine Monitoring Program: Using remote sensing for GBR-wide water quality. Final Report for 2008/2009 Activities. Canberra, ACT, CSIRO Land & Water.
- Brando, V.E., Steven, A., Schroeder, T., Dekker, A.G., Park, Y.J., Daniel, P., Ford P., 2010b. Remote-sensing of GBR Waters to assist performance monitoring of Water Quality Improvement Plans in Far North Queensland. Final report for Department of the Environment and Water Heritage and the Arts. Canberra, ACT, CSIRO Land & Water.
- Brando, V.E., Dekker, A.G., Park, Y.J., Schroeder, T., 2012. An adaptive semianalytical inversion of ocean colour radiometry in optically complex waters. Applied Optics 51(15): 2808-2833.
- Brando, V.E., Blondeau-Patissier, D., Schroeder, T., Dekker, A.G., Clementson, L., 2013. Reef Rescue Marine Monitoring Program: Assessment of terrestrial run-off entering the Reef and inshore marine water quality monitoring using earth observation data. Final Report for 2011/12 Activities. CSIRO, Canberra.
- Brodie, J., 1992. Enhancement of larval and juvenile survival and recruitment in Acanthaster planci from the effects of terrestrial runoff: a review. Australian Journal of Marine and Freshwater Research 43, 539–554.
- Brodie, J., Fabricius, K.E., De'ath, G., Okaji, K., 2005. Are increased nutrient inputs responsible for more outbreaks of crown-of-thorns starfish? An appraisal of the evidence. Marine Pollution Bulletin 51, 266–278.
- Brodie, J., Binney, J., Fabricius, K., Gordon, I., Hoegh-Guldberg, O., Hunter, H., O'Reagain, P., Pearson, R., Quirk, M., Thorburn, P., Waterhouse, J., Webster, I., Wilkinson, S., 2008. Synthesis of evidence to support the Scientific Consensus Statement on Water Quality in the Great Barrier Reef. The State of Queensland (Department of the Premier and Cabinet) Brisbane.

<http://www.reefplan.qld.gov.au/publications/scientific_consensus_statement.shtm>.

- Brodie, J., Waterhouse, J. 2009. Assessment of relative risk of the impacts of broad-scale agriculture on the Great Barrier Reef and priorities for investment under the Reef Protection Package, Stage 1 Report April 2009. ACTFR Report 09/17.
- Brodie, J., Mitchell, A., Waterhouse, J., 2009. Regional assessment of the relative risk of the impacts of broad-scale agriculture on the Great Barrier Reef and priorities for investment under the Reef Protection Package, Stage 2 Report, July 2009. ACTFR Report 09/30.
- Brodie, J., Wolanski, E., Lewis, S., Bainbridge, Z., 2012a. A review of the residence times of land-sourced contaminants in the Great Barrier Reef lagoon and the implications for management of these contaminants on the Great Barrier Reef catchment. Marine Pollution Bulletin 65, 267-279. doi:10.1016/j.marpolbul.2011.12.011.

- Brodie, J., Kroon, F., Schaffelke, B., Wolanski, E., Lewis, S., Devlin, M., Bainbridge, Z., Waterhouse, J., Davis,
 A., 2012b. Terrestrial pollutant runoff to the Great Barrier Reef: current issues, priorities and
 management responses. Marine Pollution Bulletin 65, 81–100.
- Brodie, J., Fabricius, K., Lewis, S., Bainbridge, Z., Bartley, R., 2013a. Chapter 3: Review of increased suspended sediment delivery to the GBR and the effects of subsequent sedimentation and light reduction on coral reefs. In: Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef: Supporting Studies. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/30, Townsville, Australia.
- Brodie, J., Waterhouse, J., Schaffelke, B., Furnas, M., Maynard, J., Collier, C., Lewis, S., Warne, M., Fabricius, K., Devlin, M., McKenzie, L., Yorkston, H., Randall, L., Bennett, J., Brando, V., 2013b. 2013 Scientific Consensus Statement. Chapter 3: Relative risks to the Great Barrier Reef from degraded water quality The State of Queensland. Published by the Reef Water Quality Protection Plan Secretariat, July 2013. http://www.reefplan.qld.gov.au/about/scientific-consensus-statement/water-quality-risks.aspx
- Brooks, A., Spencer, J., Olley, J., Pietsch, T., Borombovits, D., Curwen, G., Shellberg, J., Howley, C., Gleeson, A., Simon, A., Bankhead, A., Klimetz, D., Eslami-Endargoli, L., Bourgeault, A., 2013. An Empirically-based Sediment Budget for the Normanby Basin: Sediment Sources, Sinks, and Drivers on the Cape York Savannah. Griffith University, 506pp.
- Bruno, J., Petes, L.E., Harvell, D., Hettinger, A., 2003. Nutrient enrichment can increase the severity of coral diseases. Ecology Letters 6: 1056-1061.
- Burke, L., Reytar, K., Spalding, M., Perry, A., 2011. Reefs at risk revisited. World Resources Institute, Washington, DC, p. 114. Available at www.wri.org.
- Burns, K.A., Brinkman, D., 2011. Organic biomarkers to describe the major carbon inputs and cycling of organic matter in the central Great Barrier Reef region. Estuarine Coastal and Shelf Science 93, 132-141.
- Burgman, M.A., 2005. Risks and decisions for conservation and environmental management. Cambridge University Press, Cambridge. 314 p. ISBN 0521835348.
- Campbell, S.J., Mckenzie, L.J., 2004. Flood related loss and recovery of intertidal seagrass meadows in southern Queensland, Australia. Estuarine, Coastal and Shelf Science 60, 477-490.
- Campbell, S.J., Roder, C.A., Mckenzie, L.J., Lee Long, W.J., 2002. Seagrass resources in the Whitsunday region 1999 and 2000, p. 50. DPI Information Series QI02043. DPI.
- Cantin, N.E., van Oppen, M.J.H., Willis, B.L., Mieog, J.C., Negri, A.P., 2009. Juvenile corals can acquire more carbon from high-performance algal symbionts. Coral Reefs 28, 405-414.
- Carilli, J.E., Norris, R.D., Black, B., Walsh, S.M., McField, M., 2010. Century-scale records of coral growth rates indicate that local stressors reduce coral thermal tolerance threshold. Global Change Biology 16, 1247-1257.
- Carilli, J.E., Prouty, N.G., Hughen, K.A., Norris, R.D., 2009. Century-scale records of land-based activities recorded in Mesoamerican coral cores. Marine Pollution Bulletin 58, 1835-1842.
- Cheal, A.J., MacNeil, M.A., Cripps, E., Emslie, M.J., Jonker, M., Schaffelke, S., Sweatman, H., 2010. Coralmacroalgal phase shifts or reef resilience: links with diversity and functional roles of herbivorous fishes on the Great Barrier Reef. Coral Reefs 29, 1005-1015.
- Chen, Z., Muller-Karger, F.E., Hu, C., 2007. Remote sensing of water clarity in Tampa Bay. Remote Sensing of Environment 109, 249-259.
- Chesworth, J.C., Donkin, M.E., Brown, M.T., 2004. The interactive effects of the antifouling herbicides Irgarol 1051 and Diuron on the seagrass *Zostera marina* (L.). Aquatic Toxicology 66, 293-305.
- Codi S, Humphrey C, Klumpp D and Delean S. 2004. Barramundi as an indicator species for environmental monitoring in north Queensland, Australia: laboratory versus field studies. Environmental Toxicology and Chemistry23, 2737-2744.
- Cogle, A.L., Carroll, C., Sherman, B.S., 2006. The use of SedNet and ANNEX to guide GBR catchment sediment and nutrient target setting. QNRM06138. Department of Natural Resources Mines and Water, Brisbane.

- Coles, R., McKenzie, L., De'ath, G., Roelofs, A., Lee Long, W., 2009. Spatial distribution of deepwater seagrass in the inter-reef lagoon of the Great Barrier Reef World Heritage Area. Marine Ecology Progress Series 392, 57–68.
- Coles, R.G., Grech, A., McKenzie, L., Rasheed, M. 2011. Evaluating risk to seagrasses in the tropical Indo-Pacific region. CERF 2011, 21st Biennial Conference of the Coastal and Estuarine Research Federation Conference Abstracts. Societies, Estuaries & Coasts: Adapting to Change. 6-10 November 2011, Daytona Beach, Florida, USA. p.43.
- Collier, C., 2013. Chapter 7: Review of the risks to seagrasses of the Great Barrier Reef caused by declining water quality. In: Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef: Supporting Studies. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/30, Townsville, Australia.
- Collier, C., Waycott, M., 2009. Drivers of change to seagrass distributions and communities on the Great Barrier Reef: Literature review and gaps analysis. Report to the Marine and Tropical Sciences Research Facility. Reef and Rainforest Research Centre Limited.
- Collier, C.J., Uthicke, S., Waycott, M., 2011. Thermal tolerance of two seagrass species at contrasting light levels: implications for future distribution in the Great Barrier Reef. Limnology and Oceanography 56, 2200-2210.
- Collier, C.J., Waycott, M., Giraldo Ospina, A., 2012. Responses of four Indo-West Pacific seagrass species to shading. Marine Pollution Bulletin 65, 342–354.
- Conley, D.J., Paerl, H.W., Howarth, R.W., Boesch, D.F., Seitzinger, S.P., Havens, K.E., Lancelot, C., Likens, G.E., 2009. Controlling eutrophication: nitrogen and phosphorus. Science 323, 1014–1015.
- Cooper, T.F., Ridd, P.V., Ulstrup, K.E., Humphrey, C., Slivkoff, M., Fabricius, K.E. 2008. Temporal dynamics in coral bioindicators for water quality on coastal coral reefs of the Great Barrier Reef. Marine and Freshwater Research 59, 703-716
- Costa Jr., O.S., Leao, Z.M., Nimmo, M., Attrill, M.J., 2000. Nutrification impacts on coral reefs from northern Bahia, Brazil. Hydrobiologia 440: 370-415.
- Cotsell, P., Gale, K., Hajkowicz, S., Lesslie, R., Marshall, N., Randall, L., 2009. Use of a multiple criteria analysis (MCA) process to inform Reef Rescue regional allocations. In: Proceedings of the 2009 Marine and Tropical Sciences Research Facility Annual Conference 28–30 April 2009 Rydges Southbank Hotel, Townsville. Compiled by Shannon Hogan and Suzanne Long Reef and Rainforest Research Centre Limited.<http://www.rrrc.org.au/publications/downloads/Theme-5-RRRC-2009-Annual-Conference-Proceedings.pdf>.
- Cunning, R., Baker, A.C., 2012. Excess algal symbionts increase the susceptibility of reef corals to bleaching. Nature Climate Change doi:10.1038/nclimate1711.
- Dachs, J., Méjanelle, L., 2010. Organic Pollutants in Coastal Waters, Sediments, and Biota: A Relevant Driver for Ecosystems During the Anthropocene? The H.T. Odum Synthesis Essay, Estuaries and Coasts 33, 1–14. DOI 10.1007/s12237-009-9255-8
- Davis, A.M., Lewis, S.E., Bainbridge, Z.T., Glendenning, L., Turner, R., Brodie, J.E., 2012. Dynamics of herbicide transport and partitioning under event flow conditions in the lower Burdekin region, Australia. Marine Pollution Bulletin 65, 182-193.
- Davis, A.M., Thorburn, P.J., Lewis, S.E., Bainbridge, Z.T., Attard, S.J., Milla, R., Brodie, J.E., in press. Environmental impacts of irrigated sugarcane production: Herbicide run-off dynamics from farms and associated drainage systems. Agriculture, Ecosystems and the Environment.
- De'ath, G., Fabricius, K.E., 2008. Water quality of the Great Barrier Reef: distributions, effects on reef biota and trigger values for the protection of ecosystem health. Final Report to the Great Barrier Reef Marine Park Authority. Australian Institute of Marine Science, Townsville. (104 pp.)
- De'ath, G., Fabricius, K., 2010. Water quality as a regional driver of coral biodiversity and macroalgae on the Great Barrier Reef. Ecological Applications 20, 840-850.

- De'ath, G., Fabricius, K.E., Sweatman, H., Puotinen, M., 2012. The 27–year decline of coral cover on the Great Barrier Reef and its causes. Proceedings of the National Academy of Sciences of the United States of America 109 (44), 17734-17735.
- Devlin, M., Brodie, J., Waterhouse, J., Mitchell, A., Audas, D., Haynes, D. 2003. Exposure of Great Barrier Reef inner-shelf reefs to river-borne contaminants. In: Proceedings of the 2nd National Conference on Aquatic Environments: Sustaining our aquatic environments- Implementing solutions. Queensland Department of Natural Resources and Mines, Brisbane.
- Devlin, M., Brodie, J., 2005. Terrestrial discharge into the Great Barrier Reef Lagoon: nutrient behaviour in coastal waters. Marine Pollution Bulletin 51, 9–22.
- Devlin, M., Schaffelke, B., 2009. Spatial extent of riverine flood plumes and exposure of marine ecosystems in the Tully coastal region, Great Barrier Reef. Marine and Freshwater Research 60, 1109-1122.
- Devlin, M.J., McKinna, L.I.W., Alvarez-Romero, J.G., Abott, B., Harkness, P., Brodie, J., 2012. Mapping the pollutants in surface river plume waters in the Great Barrier Reef, Australia. Marine Pollution Bulletin 65, 224–235.
- Devlin, M., Petus, C., da Silva, E., Alvarez-Romero, J.G., Zeh, D., Waterhouse, J., Brodie, J., 2013a. Chapter 5: Mapping of exposure to flood plumes, water types and exposure to pollutants (DIN, TSS) in the Great Barrier Reef: toward the production of operational risk maps for the World's most iconic marine ecosystem. In: Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef: Supporting Studies. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/30, Townsville, Australia.
- Devlin, M., Petus, C., Collier, C., Zeh, D., McKenzie, L., 2013b. Chapter 6: Assessment of seagrass and water quality influences in the GBR: A case study linking annual measurements of seagrass change to satellite water clarity data (Cleveland Bay, Queensland). In: Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef: Supporting Studies. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/30, Townsville, Australia.
- De Vantier, L., 2010. Reef-building corals of Hervey Bay, South-East Queensland. Baseline Survey Report to the Wildlife Preservation Society of Queensland, Fraser Coast Branch, June 2010.
- Dight, I.J., Bode, L., James, M.K., 1990a. Modelling the larval dispersal of *Acanthaster planci* . 1. Large scale hydrodynamics, Cairns Section, Great Barrier Reef Marine Park. Coral Reefs 9, 115-123.
- Dight, I.J., James, M.K., Bode, L./, 1990b. Modelling the larval dispersal of *Acanthaster planci*. 2. Patterns of reef connectivity. Coral Reefs 9, 125-134.
- Douglas, G.B., Ford, P.W., Palmer, M.R., Noble, R.M., Packett, R.J., Krull, E.S., 2008. Fitzroy River Basin, Queensland, Australia. IV. Identification of flood sediment sources in the Fitzroy River. Environmental Chemistry 5, 243-257, 10.1071/en07091.
- Dubinsky, Z., Stambler, N., Ben-Zion, M., McClosky, L., Muscatine, L., Falkowski, P., 1990. The effect of external nutrient resources on the optical properties and photosynthetic efficiency of *Stylophora pistillata*. Proceedings of the Royal Society London, Biological Sciences 239, 231-246.
- Dugdale, R.C., Goering, J.J., 1967. Uptake of new and regenerated forms of nitrogen in primary productivity. Limnology and Oceanography 12, 196-206.
- Elledge, A., Thornton, C., 2012. The Brigalow Catchment Study: Nitrogen runoff generation rates from pasture legumes and changes since land development. In: Proceedings of the 34th Hydrology and Water Resources Symposium. Sydney. (Ed. S. Westra). Pages 1000-1007. Engineers Australia.
- Endean, R., Stablum, W., 1975. Population explosions of *Acanthaster planci* and associated destruction of the hard-coral cover of reefs of the Great Barrier Reef, Australia. Environment Conservation 2, 247-256.
- Entsch, B., Sim, R.G., Hatcher. B.G., 1983. Indications from photosynthetic components that iron is a limiting nutrient in primary producers on coral reefs. Marine Biology 73, 17-30.

- Erftemeijer, P.L.A., Robin Lewis Iii, R.R., 2006. Environmental impacts of dredging on seagrasses: A review. Marine Pollution Bulletin 52, 1553-1572.
- Fabricius, K.E. 2005. Effects of terrestrial runoff on the ecology of corals and coral reefs: review and synthesis. Marine Pollution Bulletin 50: 125-146.

Fabricius, K.E., 2011. Factors determining the resilience of coral reefs to eutrophication: a review and conceptual model. In: Dubinsky, Z., Stambler, N. (Eds.), Coral Reefs: An Ecosystem in Transition. Springer Press, pp. 493–506.

- Fabricius, K.E., Wolanski, E., 2000. Rapid Smothering of Coral Reef Organisms by Muddy Marine Snow. Estuarine and Coastal Shelf Science 50, 115-120.
- Fabricius, K., De'ath, G., 2001. Environmental factors associated with the spatial distribution of crustose coralline algae on the Great Barrier Reef. Coral Reefs 19: 303-309.
- Fabricius, K.E., Wild, C., Wolanski, E., Abele, D., 2003. Effects of transparent exopolymer particles and muddy terrigenous sediments on the survival of hard coral recruits. Estuarine and Coastal Shelf Science 57, 613–621.
- Fabricius, K.E., Okaji, K., De'ath, G., 2010. Three lines of evidence to link outbreaks of the crown-of-thorns seastar *Acanthaster planci* to the release of larval food limitation. Coral Reefs 29, 593-605.
- Fabricius, K.E., Cooper, T.F., Humphrey, C., Uthicke, S., De'ath, G., Davidson, J., LeGrand, H., Thompson, A., Schaffelke, B., 2012. A bioindicator system for water quality on inshore coral reefs of the Great Barrier Reef. Marine Pollution Bulletin 65, 320-332.
- Fabricius, K., De'ath, G., Humphrey, C., Zagorskis, I., Schaffelke, B., 2013. Intra-annual variation in turbidity in response to terrestrial runoff at near-shore coral reefs of the Great Barrier Reef. Estuarine and Coastal Shelf Science 116, 57-65.
- Flores, F., Collier, C.J., Mercurio, P., Negri, A.P., in review. Phototoxicity of four photosystem II herbicides to tropical seagrasses. PLoS ONE.
- Fu, F.-X., Bell, P.R.F., 2003. Growth, N2 fixation and photosynthesis in a cyanobacterium, Trichodesmium sp. under Fe stress. Biotechnology Letters 25, 645-649.
- Furnas, M., Brinkman, R., Fabricius, K., Tonin, H., Schaffelke, B., 2013a. Chapter 1: Linkages between river runoff, phytoplankton blooms and primary outbreaks of crown-of-thorns starfish in the Northern GBR. In: Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef: Supporting Studies. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/30, Townsville, Australia.
- Furnas, M., O'Brien, D. and Warne, M., 2013b. Chapter 2: The Redfield Ratio and potential nutrient limitation of phytoplankton in the Great Barrier Reef. In: Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef: Supporting Studies. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/30, Townsville, Australia.
- Furnas, M., Mitchell, A., Skuza, M., Brodie, J., 2005. The other 90 %: phytoplankton responses to enhanced nutrient availability in the Great Barrier Reef lagoon. Marine Pollution Bulletin 51, 253-265.
- Furnas, M., Alongi, D., McKinnon, D., Trott, L., Skuza, M., 2011. Regional-scale nitrogen and phosphorus budgets for the northern (14ºS) and central (17ºS) Great Barrier Reef shelf ecosystem. Continental Shelf Research 31, 1967-1990.
- Gao, Y., Fang, J., Zhang, J., Ren, L., Mao, Y., Li, B., Zhang, M., Liu, D., Du, M., 2011. The impact of the herbicide atrazine on the growth and photosynthesis of seagrass *Zostera marina* (L.), seedlings. Marine Pollution Bulletin 62, 1628-1631.
- Great Barrier Reef Marine Park Authority, 2009. Water Quality Guidelines for the Great Barrier Reef Marine Park. Great Barrier Reef Marine Park Authority, Townsville, 99p.

- GBRMPA, 2011. http://www.gbrmpa.gov.au/__data/assets/pdf_file/0016/14308/GBRMPA-ExtremeWeatherAndtheGBR-2010-11.pdf
- Grech, A., Marsh, H.D., 2007. Prioritising areas for dugong conservation in a marine protected area using a spatially explicit population model. Applied GIS 3(2), 1–14.
- Grech, A., Sheppard, J., Marsh, H., 2011. Informing species conservation at multiple scales using data collected for marine mammal stock assessments. PLoS ONE 6 (3), e17993. doi:10.1371/journal.pone.0017993.
- Greiner, R., Herr, A., Brodie, J., Haynes, D., 2005. A multi-criteria approach to Great Barrier Reef catchment (Queensland, Australia) diffuse-source pollution problem. Marine Pollution Bulletin 51, 128-137.
- Harrington, L., Fabricius, K., Eaglesham, G., Negri, A., 2005. Synergistic effects of diuron and sedimentation on photosynthetic yields and survival of crustose coralline algae. Marine Pollution Bulletin 51, 415-427.
- Hart, B.T., Pollino, C.A., 2008. Increased Use of Bayesian Network Models Will Improve Ecological Risk Assessments, Human and Ecological Risk Assessment 14, 851-853.
- Hart, B.T., Burgman, M., Grace, M., Pollino, C., Thomas, C., Webb, J.A., Allison, G.A., Chapman, M.,Duivenvoorden, L., Feehan, P., Lund, L., Carey, J., McCrea, A., 2005. Ecological Risk ManagementFramework for the Irrigation Industry. Land and Water Australia, Canberra (Technical Report).
- Hauri, C., Fabricius, K.E., Schaffelke, B., Humphrey, C., 2010. Chemical and physical environmental conditions underneath mat- and canopy-forming macroalgae, and their effects on understorey corals. PLoS ONE 5: e12685. doi:12610.11371/journal.pone.0012685.
- Hayes, K.R., Oliver, D., Kookana, R.S., Kroon, F., 2012. Development of ecological risk assessment methodology for terrestrial pollutants to the Great Barrier Reef ecosystems. CSIRO, Australia.
- Haynes, D., 1997. Marine park debris in the Far Northern Section of the Great Barrier Reef Marine Park. Reef Research, Vol. 7, No. 3-4.
- Haynes, D., Ralph, P., Prange, J., Dennison, W., 2000. The impact of the herbicide diuron on photosynthesis in three species of tropical seagrass. Marine Pollution Bulletin 41, 288–293.
- Haynes, D., Christie, C., Marshall, P., Dobbs, K., 2002. Antifoulant concentrations at the site of the Bunga Teratai Satu grounding, Great Barrier Reef, Australia. Marine Pollution Bulletin 44, 956-976.
- Howarth, R., Chan, F., Conley, D.J., Garnier, J., Doney, S.C., Marino, R., Billen, G., 2011. Coupled biogeochemical cycles: eutrophication and hypoxia in temperate estuaries and coastal marine ecosystems. Frontiers in Ecology and Environment 9, 18–26.
- Hughes, T.P., Rodrigues, M.J., Bellwood, D.R., Ceccerelli, D., Hoegh-Guldberg, O., McCook, L., Moltchaniwskyj, N., Pratchett, M.S., Steneck, R.S. Willis, B.L., 2007. Regime-shifts, herbivory and the resilience of coral reefs to climate change. Current Biology 17, 360–365.
- Humphrey, C., Codi King, S. and Klumpp, D., 2007. The use of biomarkers in barramundi (Lates calcarifer) to monitor contaminants in estuaries of Tropical North Queensland. Report to the Marine and Tropical Sciences Research Facility. Reef and Rainforest Research Centre Limited, Cairns (32pp.)
- Humphrey, C., Weber, M., Lott, C., Cooper, T., Fabricius K., 2008. Effects of suspended sediments, dissolved inorganic nutrients and salinity on fertilisation and embryo development in the coral *Acropora millepora* (Ehrenberg, 1834). Coral Reefs 27, 837-850.
- Johnson, J.E., Maynard, J.A., Devlin, M.J., Wilkinson, S., Anthony, K.R.N., Yorkston, H., Heron, S.F., Puotinen, M.L., van Hooidonk, R., 2013. 2013 Scientific Consensus Statement. Chapter 2: Resilience of Great Barrier Reef marine ecosystems and drivers of change. The State of Queensland. Published by the Reef Water Quality Protection Plan Secretariat, July 2013. http://www.reefplan.qld.gov.au/about/scientificconsensus-statement/resilience-of-marine-ecosystems.aspx
- Jones, R.J., Kerswell, A.P., 2003. Phytotoxicity of photosystem II (PS II) herbicides to coral. Marine Ecology Progress Series 261, 149–159.

- Jones, R.J., Mueller, J.F., Haynes, D., Schreiber, U., 2003. Effects of herbicides diuron and atrazine on corals of the Great Barrier Reef, Australia. Marine Ecology Progress Series 251, 153–167.
- Jones, M.A., Stauber, J., Apte, S., Simpson, S., Vicente-Beckett, V., Johnson, R. and Duivenvoorden, L. 2005. A risk assessment approach to contaminants in Port Curtis, Queensland, Australia. Marine Pollution Bulletin 51, 448-458.
- Kenchington, R.A., 1977. Growth and recruitment of *Acanthaster planci* (L.) on the Great Barrier Reef. Biological Conservation 11, 103-118.
- Kendrick, G.A., 1991. Recruitment of coralline crusts and filamentous turf algae in the Galapagos archipelago: Effect of simulated scour, erosion and accretion. Journal of Experimental Marine Biology and Ecology 147, 47-63.
- Kennedy, K. Schroeder, T. Shaw, M. Haynes, D. Lewis, S. Bentley, C. Paxman, C. Carter, S. Brando, V. Bartkow, M. Hearn, L. Mueller, J.F., 2012a. Long term monitoring of photosystem II herbicides Correlation with remotely sensed freshwater extent to monitor changes in the quality of water entering the Great Barrier Reef, Australia. Marine Pollution Bulletin 65, 292-305.
- Kennedy, K., Devlin, M., Bentley, C., Lee-Chue, K., Paxman, C., Carter, S., Lewis, S.E., Brodie, J., Guy, E., Vardy, S., Martin, K.C., Jones, A., Packett, R., Mueller, J.F., 2012b. The influence of a season of extreme wet weather events on exposure of the World Heritage Area Great Barrier Reef to pesticides. Marine Pollution Bulletin 64, 1495–1507.
- King, B.J., Zapata, M., McAllister, F., Done, T., 2002. Modelling the distribution of river plumes in the central and northern Great Barrier Reef shelf, Technical Report No. 44, CRC Reef Research Centre, Townsville.
- King, J., Alexander, F., Brodie, J.E., in press. Regulation of pesticides in Australia: The Great Barrier Reef as a case study for evaluating effectiveness. Agriculture Ecosystems and Environment.
- Kroon, F., Turner, R., Smith, R., Warne, M., Hunter, H., Bartley, R., Wilkinson, S., Lewis, S., Waters, D., Carroll, C. 2013. 2013 Scientific Consensus Statement. Chapter 4: Sources of sediment, nutrients, pesticides and other pollutants in the Great Barrier Reef catchment. The State of Queensland. Published by the Reef Water Quality Protection Plan Secretariat, July 2013. http://www.reefplan.qld.gov.au/about/scientific-consensus-statement/sources-of-pollutants.aspxLarcombe, P., Ridd, P.V., Prytz, A., Wilson, B., 1995. Factors controlling suspended sediment on inner-shelf coral reefs, Townsville, Australia. Coral Reefs 14, 163–171.
- Lesslie, R.G., Hill, M.J., Hill, P., Cresswell, H.P., Dawson, S. 2008. The Application of a Simple Spatial Multi-Criteria Analysis Shell to Natural Resource Management Decision Making. In Landscape Analysis and Visualisation: Spatial Models for Natural Resource Management and Planning, (Eds. Pettit, C, Cartwright, W, Bishop, I, Lowell, K, Pullar, D & Duncan, D), Springer, Berlin, pp 73–96.
- Lewis, S.E., Brodie, J.E., Bainbridge, Z.T., Rohde, K.W., Davis, A.M., Masters, B.L., Maughan, M., Devlin, M.J., Mueller, J.F., Schaffelke, B., 2009. Herbicides: A new threat to the Great Barrier Reef. Environmental Pollution 157, 2470–2484.
- Lewis, S.E., Schaffelke, B., Shaw, M., Bainbridge, Z.T., Rohde, K.W., Kennedy, K.E., Davis, A.M., Masters, B.L., Devlin, M.J., Mueller, J.F., Brodie, J.E., 2012. Assessing the risks of PS-II herbicide exposure to the Great Barrier Reef. Marine Pollution Bulletin 65, 280-291.
- Lewis, S., Smith, R., O'Brien, D., Warne, M.St.J., Negri, A., Petus, C., da Silva, E., Zeh, D. Turner, R.D.R., Davis, A., Mueller, J., Brodie, J., 2013a. Chapter 4: Assessing the risk of additive pesticide exposure in Great Barrier Reef ecosystems. In: Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef: Supporting Studies. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/30, Townsville, Australia.
- Lewis, S.E., Bainbridge, Z.T., Kuhnert, P.M., Sherman, B.S., Henderson, B., Dougall, C., Cooper, M., Brodie, J.E., 2013b. Calculating sediment trapping efficiencies for reservoirs in tropical settings: a case study from the Burdekin Falls Dam, NE Australia. Water Resources Research 49, 1-13, doi: 10.1002/wrcr.20117.

- Limpus, C.J., Limpus, D.J., 2003a. Loggerhead turtles in the equatorial and southern Pacific Ocean: a species in decline. In 'Loggerhead Sea Turtles'. (Eds A. B. Bolten and B. E. Witherington.) pp. 199–209. (Smithsonian Institution Press: Washington, DC.)
- Limpus, C.J., Limpus, D.J., 2003b. The biology of the loggerhead turtle, Caretta caretta, in southwest Pacific Ocean foraging areas. In 'Biology and Conservation of Loggerhead Turtles'. (Eds B. Witherington and A. Bolten.) pp. 93–113. (Smithsonian Institution Press: Washington, DC.)
- Limpus, C.J., Couper, P.J., Read, M.A., 1994. The loggerhead turtle, Caretta caretta, in Queensland: population structure in a warm temperate feeding area. Memoirs of the Queensland Museum 37, 195– 204.
- Littler, M.M., Littler, D. S., 2007. Assessment of coral reefs using herbivory/nutrient assays and indicator groups of benthic primary producers: a critical synthesis, proposed protocols, and critique of management strategies. Aquatic Conservation-Marine and Freshwater Ecosystems 17, 195-215.
- Loya, Y., Lubinevsky, H., Rosenfeld, M., Kramarsky-Winter, E., 2004. Nutrient enrichment caused by in situ fish farms at Eilat, Red Sea is detrimental to coral reproduction. Marine Pollution Bulletin 49(4), 344-353.
- Lucas, J.S., 1982. Quantitative studies of feeding and nutrition during larval development of the coral reef asteroid *Acanthaster planci* (L.). Journal of Experimental Marine Biology and Ecology 65, 173-194.
- Luick, J.L., Mason, L., Hardy, T., Furnas, M.J., 2007. Circulation in the Great Barrier Reef Lagoon using numerical tracers and in situ data. Continental Shelf Research 27, 757-778.
- Magnusson, M., Heimann, K., Negri, A.P., 2008. Comparative effects of herbicides on photosynthesis and growth of tropical estuarine microalgae. Marine Pollution Bulletin 56, 1545–1552.
- Magnusson, M., Heimann, K., Quayle, P., Negri, A.P., 2010. Additive toxicity of herbicide mixtures and comparative sensitivity of tropical benthic microalgae. Marine Pollution Bulletin 60, 1978–1987.
- Magnusson, M., Heimann, K., Ridd, M., Negri, A.P., 2012. Chronic herbicide exposures affect the sensitivity and community structure of tropical benthic microalgae. Marine Pollution Bulletin 65, 363–372.
- Mallela, J., Roberts, C., Harrod, C., Goldspink, C.R., 2007. Distributional patterns and community structure of Caribbean coral reef fishes within a river-impacted bay. Journal of Fish Biology 70, 523-537.
- Maritorena, S., Siegel, D.A., Peterson, A., 2002. Optimization of a Semi-Analytical Ocean Color Model for Global Scale Applications. Applied Optics 41(15), 2705-2714.
- Marsh, H., De'ath, G., Gribble, N., Lane, B., 2005. Historical marine population estimates: triggers or targets for conservation? The dugong case study. Ecological Applications 15, 481-492. doi:10.1890/04-0673.
- Marshall, P., Christie, C., Dobbs, K., Green, A, Haynes, D., Brodie, J., Michalek-Wagner K., Smithe, A., Storrie, J., Turak, E. 2002. Grounded ship leaves TBT-based antifoulant on the Great Barrier Reef: An overview of the Environmental response. Spill Science & Technology Bulletin, Special Report 7, 215-221.
- Marubini, F., Davies, P.S., 1996. Nitrate increases zooxanthellae population density and reduces skeletogenesis in corals. Marine Biology 127, 319-328.
- Maughan, M., Brodie, J., 2009. Reef exposure to river-borne contaminants: a spatial model. Marine and Freshwater Research 60, 1132–1140.http://dx.doi.org/10.1071/MF08328
- Mckenzie, L.J., Unsworth, R.K.F., 2011. Surviving the flood: How long can seagrass "hold its breath"? SeagrassWatch Magazine 43, 2-4.
- McKenzie, L.J., Unsworth, R.K.F., Waycott, M., 2010a. Reef Rescue Marine Monitoring Program: Intertidal Seagrass, Annual Report for the Sampling Period 1st September 2009 – 31st May 2010. Fisheries Queensland, Cairns p. 136. http://www.gbrmpa.gov.au/__data/assets/pdf_file/0009/7677/RRMMP_Seagrass_annual_report_2009_ 10.pdf.
- McKenzie, L.J., Yoshida, R., Grech, A., Coles, R., 2010b. Queensland seagrasses. Status 2010 Torres Strait and East Coast. Fisheries Queensland (DEEDI), Cairns. 6pp.

- McKenzie, L., Collier, C.J., Waycott ,M., 2012a. Reef Rescue Marine Monitoring Program: Inshore seagrass, annual report for the sampling period 1st September 2010-31st May 2011, p. 230pp. Fisheries Queensland, Cairns.
- McKenzie, L., Collier, C., Waycott, M., Unsworth, R., Yoshida, R., Smith, N., 2012b. Monitoring inshore seagrasses of the GBR and responses to water quality. Proceedings of the 12th International Coral Reef Symposium, Cairns, Australia, 9-13 July 2012. Mini-Symposium 15b – Seagrasses and seagrass ecosystems. Available at: http://www.icrs2012.com/proceedings/manuscripts/ICRS2012_15B_4.pdf
- McMahon, K., Bengtson Nash, S., Eaglesham, G., Müller, J.F., Duke, N.C., Winderlich, S., 2005. Herbicide contamination and the potential impact to seagrass meadows in Hervey Bay, Queensland, Australia. Marine Pollution Bulletin 51, 325–334.
- McQuatters-Gollop, A., Gilbert, A.J., Mee, L.D., Vermaat, J.E., Artioli, Y., Humborg, C., Wulff, F., 2009. How well do ecosystem indicators communicate the effects of anthropogenic eutrophication? Estuarine, Coastal and Shelf Science 82, 583-596.
- Meager, J.J., Limpus, C.J., 2012. Marine wildlife stranding and mortality database annual report 2011. I Dugong. Conservation Technical and Data Report 2011 (1), 1-30.
- Meakins, N.C., Bubb, J.M., Lester, J.N., 1995. The mobility, partitioning and degradation of atrazine and simazine in the salt marsh environment. Marine Pollution Bulletin 30, 812-819.
- Mellors, J.E., 2003. Sediment and nutrient dynamics in coastal intertidal seagrass of north eastern tropical Australia. PhD Thesis. James Cook University, Townsville.
- Moran, P.J., De'ath, G., 1992. Estimates of the abundance of the crown-of-thorns starfish Acanthaster planci in outbreaking and non-outbreaking populations on reefs within the Great Barrier Reef. Marine Biology 113, 509-515.
- Muller, R., Schreiber, U., Escher, B.I., Quayle, P., Bengtson Nash, S.M., Mueller, J.F., 2008. Rapid exposure assessment of PSII herbicides in surface water using a novel chlorophyll a fluorescence imaging assay. Science of the Total Environment 401, 51-59.
- Negri, A.P., Webster, N.S., Hill, R.T., Heyward, A.J., 2001. Metamorphosis of broadcast spawning corals in response to bacteria isolated from crustose algae. Marine Ecology Progress Series 223, 121-131.
- Negri, A., Vollhardt, C., Humphrey, C., Heyward, A., Jones, R., Eaglesham, G., Fabricius, F., 2005. Effects of the herbicide diuron on the early life history stages of coral. Marine Pollution Bulletin 51, 370–383.
- Negri, A.P., Flores, F., Röthig, T., Uthicke, S., 2011. Herbicides increase the vulnerability of corals to rising sea surface temperature. Limnolology and Oceanography 56, 471–485.
- Negri, A.P., Marshall, P., 2009. TBT contamination of remote marine environments: Ship groundings and icebreakers as sources of organotins in the Great Barrier Reef and Antarctica. Journal of Environmental Management 90, S31-S40.
- Okaji, K., 1996. Feeding ecology in the early life stages of the crown-of-thorns starfish, *Acanthaster planci* (L.). James Cook University.
- Orpin, A.R., Ridd, P.V., Thomas, S., Anthony, K.R.N., Marshall, P., Oliver, J., 2004. Natural turbidity variability and weather forecasts in risk management of anthropogenic sediment discharge near sensitive environments. Marine Pollution Bulletin 49(7-8), 602-612.
- Paerl, H.W., 2009. Controlling eutrophication along the freshwater–marine continuum: dual nutrient (N and P) reductions are essential. Estuaries Coasts 32, 593–601.
- Pandolfi, J.M., Bradbury, R.H., Sala, E., Hughes, T.P., Bjorndal, K.A., Cooke, R.G., McArdle, D., McClenachan, L., Newman, M.J.H., Paredes, G., Warner, R.R., Jackson, J.B.C., 2003. Global trajectories of the long-term decline of coral reef ecosystems. Science 301, 955–958.
- Passow, U., 2002. Transparent exopolymer particles (TEP) in aquatic environments. Progress in Oceanography 55, 287-333.

- PPDB, 2011. Pesticide Properties DataBase, University of Hertfordshire, UK, viewed 22 May 2013, http://sitem.herts.ac.uk/aeru/footprint/en/index.htm
- Pratchett, M.S., 2005. Dynamics of an outbreak population of Acanthaster planci at Lizard Island, northern Great Barrier Reef (1995–1999). Coral Reefs 24, 453–462.
- Preen, A., Marsh, H., 1995. Response of dugongs to large-scale loss of seagrass from Hervey Bay, Queensland, Australia. Wildlife Research 22, 507-519.
- Preen, A.R., Long, W.J., Coles, R.G., 1995. Flood and cyclone related loss, and partial recovery, of more than 1000 km2 of seagrass in Hervey Bay, Queensland, Australia. Aquatic Botany 52, 3-17.
- Ptacnik, R., Andersen, T., Tamminen, T., 2010. Performance of the Redfield Ratio and a family of nutrient limitation indicators as thresholds for phytoplankton N vs P limitation. Ecosystems 13, 1201-1214.
- Queensland Department of the Premier and Cabinet, 2009. Reef Water Quality Protection Plan 2009: For the Great Barrier Reef World Heritage Area and adjacent catchments, Queensland Department of Premier and Cabinet, Brisbane. http://www.reefplan.qld.gov.au/library/pdf/paddock-to-reef.pdf
- QDNRM. State of Queensland (2012), Department of Natural Resources and Mines 'Watershed', http://www.derm.qld.gov.au/watershed/index.html (accessed on the 1/11/2012).
- Redfield, A.C., 1934. On the proportion of organic derivatives in sea water and their relation to the composition of plankton. Pp. 176-192, In: James Johnstone Memorial Volume, R.J. Daniel (Ed.), University of Liverpool.
- Reichelt RE, Bradbury RH, Moran PJ (1990) Distribution of Acanthaster planci outbreaks on the Great Barrier Reef between 1966 and 1989. Coral reefs 9, 97-103
- Reinert, K.H., Giddings, J.M., Judd, L., 2002. Effects analysis of time-varying or repeated exposures in aquatic ecological risk assessment of agrochemicals. Environmental Toxicology and Chemistry, 21, 1977–1992.
- Schaffelke, B. 1999. Particulate organic matter as an alternative nutrient source for tropical Sargassum species (Fucales, Phaeophyceae). J. Phycol. 35: 1150-1157.
- Schaffelke, B., Anthony, K., Blake, J., Brodie, J., Collier, C., Devlin, M., Fabricius, K., Martin, K., McKenzie, L., Negri, A., Ronan, M., Thompson, A. and Warne, M. 2013. 2013 Scientific Consensus Statement. Chapter 1: Marine and coastal ecosystem impacts The State of Queensland. Published by the Reef Water Quality Protection Plan Secretariat, July 2013. http://www.reefplan.qld.gov.au/about/scientific-consensusstatement/ecosystem-impacts.aspx
- Schoellhamer, D.H., 1996. Factors affecting suspended solids concentrations in South San Francisco Bay, California. Journal of Geophysical Research 101 (C5), 12087-12095.
- Schoellhamer, D.H., 2002. Variability of suspended-sediment concentration at tidal to annual time scales in San Francisco Bay, USA. Continental Shelf Research 22, 1857-1866.
- Schoellhamer, D.H., Mumley, T.E., Leatherbarrow, J.E., 2007. Suspended sediment and sediment-associated contaminants in San Francisco Bay. Environmental Research 105, 119-131.
- Schindler, D.W. (1977) The evolution of phosphorus limitation in lakes. Science 195: 260-262.
- Schindler, D.W., Hecky, R.E., Findlay, D.L., Stainton, M.P., Parker, B.R., Paterson, M.J., Beaty, K.G., Lyng, M., Kasian, S.E.M., 2008. Eutrophication of lakes cannot be controlled by reducing nitrogen input: Results of a 37-year whole-ecosystem experiment. Proc. Natl. Acad. Sci. U S A 105, 11254–11258.
- Schroeder, T., Brando, V.E., Cherukuru, N.R.C., Clementson, L.A., Blondeau-Patissier, D., Dekker, A.G., Schaale, M., Fischer, J., 2008. Remote sensing of apparent and inherent optical properties of Tasmanian coastal waters: application to MODIS data. Ocean Optics XIX, Barga, Italy.
- Schroeder T., Devlin, M.J., Brando, V.E., Dekker, A.G., Brodie, J.E., Clementson, L.A., McKinna, L., 2012. Interannual variability of wet season freshwater plume extent into the Great Barrier Reef lagoon based on satellite coastal ocean colour observations. Marine Pollution Bulletin 65, 210–223.
- Schwarzenbach, R.P., Escher, B.I., Fenner, K., Hofstetter, T.B., Johnson, C.A., von Gunten, U., Wehrli, B., 2006. Science 313 (5790), 1072e1077. doi:10.1126/ science. 1127291

- Shaw, M., Furnas, M.J., Fabricius, K., Haynes, D., Carter, S., Eaglesham, G., Mueller, J.F., 2010. Monitoring pesticides in the Great Barrier Reef. Marine Pollution Bulletin 60, 113-122.
- Smith, S.V., 1984. Phosphorus versus nitrogen limitation in the marine environment. Limnology and Oceanography 29, 1149-1160.
- Smith, J., Douglas, G.B., Radke, L.C., Palmer, M., Brooke, B.P., 2008. Fitzroy River Basin, Queensland, Australia. III. Identification of sediment sources in the coastal zone. Environmental Chemistry 5, 231–242.
- Smith, R., Middlebrook, R., Turner, R., Huggins, R., Vardy, S., Warne, M., 2012. Largescale pesticide monitoring across Great Barrier catchments – paddock to Reef Integrated Monitoring, Modelling and Reporting Program. Marine Pollution Bulletin 65, 117–127.
- Sobtzick, S., Hagihara, R., Grech, A., Marsh, H., 2012. Aerial survey of the urban coast of Queensland to evaluate the response of the dugong population to the widespread effects of the extreme weather events of the summer of 2010-11. Final Report to the Australian Marine Mammal Centre and the National Environment Research Program June 1 2012. James Cook University, Queensland.
- Storlazzi, C.D., Field, M.E., Bothner, M.H., Presto, M.K., Draut, A.E., 2009. Sedimentation processes in a coral reef embayment: Hanalei Bay, Kauai. Marine Geology 264, 140-151.
- Sweatman, H., 2008. No-take reserves protect coral reefs from predatory starfish. Current Biology 18, R598-R599.
- Thomas, K.V., McHugh, M., Waldock, M., 2002. Antifouling paint booster biocides in UK coastal waters: inputs, occurrence and environmental fate. Science of the Total Environment 293, 117-127.
- Thompson, A., Costello, P., Davidson, J., Logan, M., Schaffelke, B., Uthicke, S., Takahashi, M., 2011. Reef Rescue Marine Monitoring Program. Report of AIMS Activities Inshore coral reef monitoring 2011. Australian Institute of Marine Science, Townsville.
- Thorburn, P.J., Biggs, J.S., Attard, S.J., Kemei, J., 2011. Environmental impacts of irrigated sugarcane production: nitrogen lost through runoff and leaching. Agriculture Ecosystems and Environment 144, 1–12. doi:10.1016/j.agee.2011.08.003.
- Thorburn, P., Rolfe, J., Wilkinson, S., Silburn, M., Blake, J., Gongora, M., Windle, J., VanderGragt, M.,
 Wegschield, C., Ronan, M., Carroll, C., 2013. 2013 Scientific Consensus Statement. Chapter 5: The water
 quality and economic benefits of agricultural management practices. The State of Queensland. Published
 by the Reef Water Quality Protection Plan Secretariat, July 2013.
 http://www.reefplan.qld.gov.au/about/scientific-consensus-statement/agricultural-managementpractices.aspx
- Thornton, C., Elledge, A., 2012. The Brigalow Catchment Study: Increases in runoff associated with land development can still be detected in flood events at a small catchment scale. In: Proceedings of the 34th Hydrology and Water Resources Symposium. Sydney. (Ed. S. Westra). Pages 1565-1570. Engineers Australia.
- Traas, T.P., Van de Meent, D., Posthuma, L., Hamers, T., Kater, B.J., De Zwart, D., Aldenberg, T., 2002. The potentially affected fraction as a measure of ecological risk. In: Posthuma L, Suter II GW, Traas TP, eds. Species sensitivity distributions in ecotoxicology. Boca Raton, FL, USA: CRC Press.
- Turner, R., Huggins, R., Wallace, R., Smith, R., Vardy, S., Warne, M.St.J., 2012. Sediment, Nutrient and Pesticide Loads: Great Barrier Reef Catchment Loads Monitoring 2009-2010, Department of Science, Information Technology, Innovation and the Arts, Brisbane.
- Turner, R., Huggins, R., Wallace, R., Smith, R., Vardy, S., Warne, M.St.J., 2013. Loads of sediment, nutrients and pesticides discharged from high priority Queensland rivers in 2010-2011. Great Barrier Reef Catchment Loads Monitoring Program, Department of Science, Information Technology, Innovation and the Arts, Brisbane. <u>http://reefplan.qld.gov.au/measuring-success/paddock-to-reef/catchment-loadsmonitoring.aspx</u>

- Udy, J.W., Dennison, W.C., 1997. Growth and physiological responses of three seagrass species to elevated sediment nutrients in Moreton Bay, Australia. Journal of Experimental Marine Biology and Ecology 217, 253-277.
- Udy, J.W., Dennison, W.C., Long, W.J.L., Mckenzie, L.J., 1999. Responses of seagrass to nutrients in the Great Barrier Reef, Australia. Marine Ecology Progress Series 185, 257-271.
- US EPA (U.S. Environmental Protection Agency) 1998. Guidelines for Ecological Risk Assessment. EPA/630/R-95/002F. Risk Assessment Forum, Washington, DC, USA.
- Vallotton, N., Eggen, R.I.L., Escher, B.I., Krayenbühl, J., Chèvre, N., 2008. Effect of pulse herbicidal exposure on *Scenedesmus vacuolatus*: A comparison of two photosystem II inhibitors. Environmental Toxicology and Chemistry 27, 1399–1407. doi: 10.1897/07-197.
- van Dam, J.W., Negri, A.P., Mueller, J.F., Uthicke, S., 2012. Symbiont-specific responses in foraminifera to the herbicide diuron. Marine Pollution Bulletin 65, 373–383.
- Wagner, D. E., Kramer, P., Van Woesik, R., 2010. Species composition, habitat, and water quality influence coral bleaching in southern Florida. Marine Ecology Progress Series 408, 65-78.
- Waterhouse, J., Brodie, J., Lewis, S., Mitchell., A., 2012. Quantifying the sources of pollutants to the Great Barrier Reef. Mar. Pollut. Bull. 65, 394–406.
- Waterhouse, J. (compiler), 2013. Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef: Supporting Studies. A report to the Department of the Environment and Heritage Protection, Queensland Government with contributions from Furnas, M., Brodie, J., Lewis, S., Devlin, M., Collier, C., Schaffelke, B., Fabricius, K., Brinkman, R., Tonin, H., O'Brien, D., Smith, R., Warne, M.St.J., Bainbridge, Z., Bartley, R., Negri, A., Turner, R.D.R., Davis, A., Bentley, C., Mueller, J., Petus, C., da Silva, E., Zeh, D., Alvarez-Romero, J.G., McKenzie, L., Waterhouse, J., Ajani, P. TropWATER Report 13/30, Townsville, Australia.
- Waterhouse, J., Maynard, J., Brodie, J., Randall, L., Zeh, D., Devlin, M., Lewis, S., Furnas, M., Schaffelke, B., Fabricius, K., Collier, C., Brando, V., McKenzie, L., Warne, M.St.J., Smith, R., Negri, A., Henry, N., Petus, C., da Silva, E., Waters, D., Yorkston, H., Tracey, D., 2013. Section 2: Assessment of the risk of pollutants to ecosystems of the Great Barrier Reef including differential risk between sediments, nutrients and pesticides, and among NRM regions. In: Brodie et al., Assessment of the relative risk of water quality to ecosystems of the Great Barrier Reef. A report to the Department of the Environment and Heritage Protection, Queensland Government, Brisbane. TropWATER Report 13/28, Townsville, Australia.
- Waters, D.K., Carroll, C., Ellis, R., Hateley, L., McCloskey, J., Packett, R., Dougall, C., Fentie, B., in press. Modelling reductions of pollutant loads due to improved management practices in the Great Barrier Reef Catchments – Whole of GBR, Volume 1. Department of Natural Resources and Mines. Technical Report (ISBN: 978-1-7423-0999).
- Weber, M., De Beer, D. Lott, C. Polerecky, L. Kohls, K. Abed, R. Ferdelman, T. Fabricius, K., 2012. Mechanisms of damage to corals exposed to sedimentation. Proceedings of the National Academy of Sciences of the United States of America 109(24), E1558–E1567.
- Weber, M., Lott, C., Fabricius, K., 2006. Different levels of sedimentation stress in a scleractinian coral exposed to terrestrial and marine sediments with contrasting physical, geochemical and organic properties. Journal of Experimental Marine Biology and Ecology 336, 18-32.
- Webster, N.S., Smith, L.D., Heyward, A.J., Watts, J.E.M., Webb, R.I., Blackall, L.L., Negri, A.P., 2004. Metamorphosis of a scleractinian coral in response to microbial biofilms. Applied and Environmental Microbiology 70, 1213-1221.
- Wenger, A.S., Johansen, J.L., Jones, G.P., 2011. Suspended sediment impairs habitat choice and chemosensory discrimination in two coral reef fishes. Coral Reefs 30: 879-887.
- Wenger, A.S., Johansen, J.L., Jones, G.P., 2012. Increasing suspended sediment reduces foraging, growth and condition of a planktivorous damselfish. Journal of Experimental Marine Biology and Ecology 428, 43-48.

- Wenger, A.S., McCormick, M., McLeod, I.M., Jones, G.P., 2013. Suspended sediment alters predator–prey interactions between two coral reef fishes. Coral Reefs 32, 369-374. http://dx.doi.org/10.1007/s00338-012-0991-z.
- Wenger, A.S., McCormick, M.I., 2013. Determining trigger values of suspended sediment for behavioral changes in a coral reef fish. Marine Pollution Bulletin 70, 73-80. http://dx.doi.org/10.1016/j.marpolbul.2013.02.014
- West, J.M., Salm., R.V., 2003. Resistance and resilience to coral bleaching: implications for coral reef conservation and management. Conservation Biology 17, 956-967.
- Wiedenmann, J., D'Angelo, C., Smith, E.G., Hunt, A.N., Legiret, F.-E., Postle, A.D., Achterberg, E.P., 2013.
 Nutrient enrichment can increase the susceptibility of reef corals to bleaching. Nature Climate Change 3(5), 160-164.
 doi:http://www.nature.com/nclimate/journal/vaop/ncurrent/abs/nclimate1661.html#supplementary
 - doi:http://www.nature.com/nclimate/journal/vaop/ncurrent/abs/nclimate1661.html#supplementaryinformation
- Wilcox, C., Hardesty, B.D., Sharples, R., Griffin, D.A., Lawson, T.J., Gunn, R., 2013. Ghostnet impacts on globally threatened turtles, a spatial risk analysis for northern Australia. Conservation Letters, 1-8.
- Wismer, S., Hoey, A.S. and Bellwood, D.R., 2009 Cross-shelf benthic community structure on the Great Barrier Reef: relationships between macroalgal cover and herbivore biomass. Marine Ecology Progress Series 376, 45–54.
- Wolanski, E., Fabricius, K., Spagnol, S., Brinkman, R., 2005. Fine sediment budget on an inner-shelf coralfringed island, Great Barrier Reef of Australia. Estuarine and Coastal Shelf Science 65, 153-158.
- Wolanski, E., Richmond, R. Mccook, L., Sweatman, H., 2003. Mud, marine snow and coral reefs. American Scientist 91, 44-51.
- Wolanski, E., van Senden, D., 1983. Mixing of Burdekin River flood waters in the Great Barrier Reef. Australian Journal of Marine and Freshwater Research 34, 49-63.
- Wooldridge, S.A., 2009. Water quality and coral bleaching thresholds: Formalising the linkage for the inshore reefs of the Great Barrier Reef, Australia. Marine Pollution Bulletin 58, 745–751.
- Wooldridge, S.A., Done, T.J., 2009. Improved water quality can ameliorate effects of climate change on corals. Ecological Applications 19, 1492–1499.
- Zann, M., 2012. The use of remote sensing and field validation for mapping coral communities of Hervey Bay and the Great Sandy Strait and implications for coastal planning policy. Unpublished M.Phil.thesis, the University of Queensland, Brisbane.



Appendix 1. Further information regarding remote sensing assessments

Figure Ap1a. Map showing the number of valid observations in the assessment period 2002-2012 where remote sensing images have been processed to estimate concentrations of Total suspended solids and Chlorophyll.

Figure Ap1b.Results of (a) TSS 2 mg/L assessment in this project (2002 to 2012) (see section 2.4.2) and (b) the secchi depth assessment presented in De'ath and Fabricius (2008). Visual comparison of the two assessment outputs suggests roughly comparable latitudinal and shelf-based spatial patterns. For (b) Locations that are presently at less than the water quality guideline trigger value of a minimum annual mean of 10 m Secchi depth are shown in green. Orange zones show areas that exceed the guideline trigger values, having Secchi depths of 5 – 10 m. Red zones show areas of greatest concern with Secchi depth <5 m. The level of opacity for the colouring in the right-most panel indicates the level of confidence in the estimates; faded areas have higher uncertainty (mostly offshore).



Figure Ap1c. a) Results of (a) the Chlorophyll 0.45 ug/L assessment in this project (2002 to 2012) and (b) the analysis of exceedance of Chlorophyll presented in De'ath and Fabricius (2008). Visual comparison of the two assessment outputs suggests roughly comparable latitudinal and shelf-based spatial patterns. For (b), locations that are presently at less than the water quality guideline trigger value of a maximum annual mean of 0.45 μ g/L chlorophyll are shown in green. Orange zones show areas that exceed the guideline trigger values, having chlorophyll values of 0.45 – 0.8 μ g/L. Red zones show areas of greatest concern with >0.8 μ g/L chlorophyll. The level of opacity for the colouring in the right-most panel indicates the level of confidence in the estimates; faded areas have higher uncertainty (mostly offshore).



Appendix 2. Sensitivity analysis of options for combining water quality variables for determining relative risk

Three options for combining the water quality variables in the marine relative risk assessment were considered. These options essentially weighted each variable differently, and provide a very simple sensitivity analysis. All of these were performed using the *Composite tool* in MCAS-S described in Section 2.3.4 and Figures 2.7, 2.8 and 2.9 of this report. The overall weightings for each variable for each of the options are shown in Table A2.1.

- 1. Option 1 (selected option): Combining all variables individually. All variables were treated with equal weighting (1/7) and summed. The maximum score for each pixel is therefore 7, and this value is then normalised between 0 and 1. This approach essentially weights the variables differently because there are 3 sediment related variables, 3 nutrient related variables and 1 pesticide related variable which were considered to be appropriate by the assessment team given current evidence of the relative importance of nutrients and sediments compared to pesticides in the GBR.
- 2. Option 2: Grouping the sediment and nutrient variables, then combining with pesticide exposure. The results for TSS exceedance 2 mg/L and TSS 7 mg/L, and TSS plume loading 2007-2011 were averaged to produce a 'sediment' layer, and then the results for Chl exceedance 0.45 µg/L, DIN plume loading 2007-2011 and the COTS Initiation Zone were average to produce a 'nutrients' layer. The pesticide exposure layer formed the 'pesticide' layer. These layers were all normalised in MCAS-S, and then combined using the *Composite tool*. This essentially weights the sediments, nutrients and pesticides groupings equally but changes the weightings for individual variables as shown in Table A2.1. Experts considered that the weighting of COTS (1/9) was too low compared to pesticides (1/3), so Option 3 was tested.
- 3. Option 3: Separating the COTS Initiation Zone to be a separate group from Option 2. The incorporation of the COTS Initiation Zone recognises the importance of the destructive impacts of COTS on coral reefs in the GBR, and the significance of the water quality conditions in the region between Lizard Island and Cairns in initiating outbreaks. Removal of this layer reduced the overall risk in the Wet Tropics region. This approach kept the COTS Initiation Zone as a separate group with a weighting of ¼, comparable to other groupings. It was decided that an additional factor related to river influence on COTS outbreaks should be factored in a later stage of the assessment rather than weighting the COTS Initiation Zone equally to the groupings of sediments, nutrients and pesticides.

Each method was tested and mapped in MCAS-S and the results were compared by the assessment team. Figure A2.1 presents a comparison of the results for the Marine Risk Index for Option 1 and Option 2 showing only the High and Very High relative risk classes. The results of the area calculations for each relative risk class are shown in Table A2.2. The greatest differences between the options were evident in the lowest assessment classes and in the results for Mackay Whitsunday – attributable to the higher weighting given to PSII herbicide modelled concentration. This also resulted in a higher ranking for Mackay Whitsunday for coral reefs and seagrass.

The results for the Relative Risk Index for Option 2 are shown in Tables A2.3 and A2.4. When combined to produce the Relative Risk Index, the overall ranking among the Regions was elevated for Mackay Whitsunday which was ranked approximately equal to Burdekin and Fitzroy region (all ~70% of the risk assigned to the highest region, the Wet Tropics) in comparison to fourth for Option 1.

We selected Option 1 as the preferred approach given current evidence of the relative importance of nutrients and sediments compared to pesticides in the GBR. The effect of the inherent weighting of 3 for sediments, 3 for nutrients, and 1 for pesticides applied in Option 1 was most evident for the area of seagrass in the Very High and High assessment classes in the Mackay Whitsunday and Wet Tropics regions. In these regions the total area of the Very High and High assessment classes are greater in Option 2, and these areas contain mapped areas of seagrass meadows but limited coral reefs. For the Mackay Whitsunday region this means that an even greater proportion of seagrass meadows would have been in the area used to calculate the Marine Risk Index. Table A2.1. Summary of options considered for combining and weighting the seven variables used in the risk assessment. The colours represent groupings of variables, where the variables within the group were averaged prior to combining with other groups.

Weighting Options	TSS threshold exceedance (2 mg/L)	TSS threshold exceedance (7 mg/L)	TSS Plume Loading (2007- 2011)	Chl threshold exceedance (0.45µg/L)	DIN Plume Loading (2007- 2011)	COTS Initiation Zone	PSII Herbicide modelled concentration (2009-2011)
Option 1 - Selected Option: Equal weighting of all individual variables by summing	1/7	1/7	1/7	1/7	1/7	1/7	1/7
Option 2: Grouping of variables into sediments, nutrients, pesticides by averaging and then summing the grouped layers	1/9	1/9	1/9	1/9	1/9	1/9	1/3
Grouped weighting		1/3			1/3		1/3
Option 3: Separating out the effect of COTS by adding the COTS layer after summing all other variables	1/12	1/12	1/12	1/8	1/8	1/4	1/4
Grouped weighting	1/4			1/4 1/4			1/4

Table A2.2. Results of the area of coral reef and seagrass for Option 1 (individual variables) and Option 2 (grouped variables) described above.

Option 1: Individual variables

Area (km²)									
	V Low	Low	Medium	High	V High	Total	Sum High & V High	% habitat in region (High & V High)	Marine Risk Index
Coral Reefs									
Cape York	3,585	6,147	546	17	0	10,295	17	0%	15
Wet Tropics	121	1,809	448	36	0	2,415	36	1%	32
Burdekin	704	2,206	24	14	0	2,948	15	0%	13
Mackay Whitsunday	2,147	826	133	61	5	3,171	66	2%	59
Fitzroy	3,726	964	38	103	8	4,840	111	2%	100
Burnett Mary	10	265	1	5	0	282	5	2%	4
						Max	111		
Seagrass									
Cape York	1,258	8,711	1,329	32	0	11,330	32	0%	7
Wet Tropics	175	4,240	278	102	59	4,855	161	3%	34
Burdekin	1,281	4,160	150	333	136	6,060	470	8%	100
Mackay Whitsunday	5	21	210	136	25	396	160	37%	34
Fitzroy	201	5,050	177	303	15	5,746	319	6%	68
Burnett Mary	458	5,561	210	76	7	6,313	83	1%	18
						Max	470		

Option 2: Grouped variables

NRM region	V Low	Low	Medium	High	V High	Total	Sum High & V High	% of habitat in region (High & V High)	Marine Risk Index
Coral Reefs									
Cape York	6,642	3,100	573	0		10,316	0	0%	1
Wet Tropics	870	1,058	448	39	1	2,416	40	2%	47
Burdekin	2,560	353	23	13	0	2,950	14	0%	16
Mackay Whitsunday	2,226	730	129	72	13	3,169	85	3%	100
Fitzroy	3,959	736	59	79	6	4,839	85	2%	100
Burnett-Mary	196	80	2	3	1	282	4	1%	4
						Max	85		
Seagrass									
Cape York	5,679	4,575	1,106	3		11,363	3	0%	1
Wet Tropics	2,468	1,874	312	119	83	4,856	202	4%	52
Burdekin	4,063	1,380	237	350	39	6,069	389	6%	100
Mackay Whitsunday	0	11	139	152	104	407	256	60%	66
Fitzroy	1,820	3,433	270	237	1	5,760	238	4%	61
Burnett Mary	5,259	758	210	80	10	6,317	90	1%	23
						Max	389		

Table A2.3. Results of the overall risk assessment from summing the Loads, COTS Influence (for coral reefs only) and Marine Risk Index for coral reefs and seagrass for Option 2 (grouped variables). The Region that had the maximum value was given a score of 100; all other Regions are expressed as a percentage based on the value in each Region relative to the area in the Region with the maximum value.

Coral Reefs	Coral Reef Marine Risk	Loads & COTS	Sum of Indexes	Coral Reef Relative	Rank Option 2	Comparison Rank Option
	Index	Index		Risk Index		1
NRM region				(Anchored)		
Cape York	1	0	1	0	5	5
Wet Tropics	47	100	147	100	1	1
Burdekin	16	39	55	37	4	4
Mackay Whitsunday	100	13	113	77	3	3
Fitzroy	100	14	114	78	2	2
Burnett Mary	4	10	14	10	6	6
		Max	147			
Seagrass	Seagrass	Loads	Sum of	Seagrass	Rank	Comparison
Seagrass	Seagrass Marine Risk	Loads Index	Sum of Indexes	Seagrass Relative	Rank Option 2	Comparison Rank Option
Seagrass	Seagrass Marine Risk Index	Loads Index	Sum of Indexes	Seagrass Relative Risk Index	Rank Option 2	Comparison Rank Option 1
Seagrass NRM region	Seagrass Marine Risk Index	Loads Index	Sum of Indexes	Seagrass Relative Risk Index (Anchored)	Rank Option 2	Comparison Rank Option 1
Seagrass NRM region Cape York	Seagrass Marine Risk Index 1	Loads Index 0	Sum of Indexes	Seagrass Relative Risk Index (Anchored) 1	Rank Option 2	Comparison Rank Option 1 6
Seagrass NRM region Cape York Wet Tropics	Seagrass Marine Risk Index 1 52	Loads Index 0 100	Sum of Indexes	Seagrass Relative Risk Index (Anchored) 1 94	Rank Option 2 6 2	Comparison Rank Option 1 6 2
Seagrass NRM region Cape York Wet Tropics Burdekin	Seagrass Marine Risk Index 1 52 100	Loads Index 0 100 62	Sum of Indexes	Seagrass Relative Risk Index (Anchored) 1 94 100	Rank Option 2 6 2 1	Comparison Rank Option 1 6 2 1
Seagrass NRM region Cape York Wet Tropics Burdekin Mackay Whitsunday	Seagrass Marine Risk Index 1 52 100 66	Loads Index 0 100 62 25	Sum of Indexes 1 152 162 91	Seagrass Relative Risk Index (Anchored) 1 94 100 56	Rank Option 2 6 2 1 4	Comparison Rank Option 1 6 2 1 4
Seagrass NRM region Cape York Wet Tropics Burdekin Mackay Whitsunday Fitzroy	Seagrass Marine Risk Index 1 52 100 66 61	Loads Index 0 100 62 25 28	Sum of Indexes 1 152 162 91 89	Seagrass Relative Risk Index (Anchored) 1 94 100 56 55	Rank Option 2 6 2 1 4 3	Comparison Rank Option 1 6 2 1 4 3
Seagrass NRM region Cape York Wet Tropics Burdekin Mackay Whitsunday Fitzroy Burnett Mary	Seagrass Marine Risk Index 1 52 100 66 61 23	Loads Index 0 100 62 25 28 20	Sum of Indexes 1 152 162 91 89 43	Seagrass Relative Risk Index (Anchored) 1 94 100 56 55 27	Rank Option 2 6 2 1 4 3 3 5	Comparison Rank Option 1 6 2 1 4 4 3 5

Table A2.4. Results of the overall risk assessment using a sum of the anchored Indexes for coral reefs and seagrass for Option 2 (grouped variables). The Region that had the largest sum of indexes was given a score of 100; all other Regions are expressed as a percentage based on sum of indexes in each Region relative to the sum in the Region with the maximum sum of indexes.

Coral Reef and Seagrass	Coral Reef Relative Risk Index	Seagrass Relative Risk	Sum of Indexes	Relative Risk Index (Anchored)	Rank Option 2	Comparison Rank Option 1
		Index			c.	
Cape York	0	1	1	0	6	6
Wet Tropics	100	94	194	100	1	1
Burdekin	37	100	137	71	3	3
Mackay Whitsunday	77	56	133	69	2	4
Fitzroy	78	55	133	69	2	2
Burnett Mary	10	27	37	19	5	5
		Max	194			



Figure A2.1. Comparison of the Marine Risk Index results for Option 1 (individual variables combined) and Option 2 (grouped variables for sediments, nutrients and then pesticides combined).